University of Northern Iowa

UNI ScholarWorks

Dissertations and Theses @ UNI

Student Work

2005

Teacher perceptions of cross-cultural adaptability and instructional practices in international locations

Debra S. Lee University of Northern Iowa

Let us know how access to this document benefits you

Copyright ©2005 Debra S. Lee

Follow this and additional works at: https://scholarworks.uni.edu/etd

Part of the Bilingual, Multilingual, and Multicultural Education Commons, and the Curriculum and Instruction Commons

Recommended Citation

Lee, Debra S., "Teacher perceptions of cross-cultural adaptability and instructional practices in international locations" (2005). *Dissertations and Theses @ UNI*. 893. https://scholarworks.uni.edu/etd/893

This Open Access Dissertation is brought to you for free and open access by the Student Work at UNI ScholarWorks. It has been accepted for inclusion in Dissertations and Theses @ UNI by an authorized administrator of UNI ScholarWorks. For more information, please contact scholarworks@uni.edu.

Offensive Materials Statement: Materials located in UNI ScholarWorks come from a broad range of sources and time periods. Some of these materials may contain offensive stereotypes, ideas, visuals, or language.

TEACHER PERCEPTIONS OF CROSS-CULTURAL ADAPTABILITY

AND

INSTRUCTIONAL PRACTICES IN INTERNATIONAL LOCATIONS

A Dissertation

Submitted

In Partial Fulfillment

of the Requirements for the Degree

Doctor of Education

Approved:

Dr. Greg Stefanich, Chair

Dr. Bruce Rogers, Committee Member

Dr. Daver Kuster Committee Member

Dr. Røger Kueter, Committee Member

Dr. Janey Montgomery, Committee Member

Dr. Tim O'Connor, Committee Member

Debra S. Lee

University of Northern Iowa

December 2005

UMI Number: 3195940

Copyright 2005 by Lee, Debra S.

All rights reserved.

INFORMATION TO USERS

The quality of this reproduction is dependent upon the quality of the copy submitted. Broken or indistinct print, colored or poor quality illustrations and photographs, print bleed-through, substandard margins, and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if unauthorized copyright material had to be removed, a note will indicate the deletion.



UMI Microform 3195940

Copyright 2006 by ProQuest Information and Learning Company.

All rights reserved. This microform edition is protected against unauthorized copying under Title 17, United States Code.

ProQuest Information and Learning Company 300 North Zeeb Road P.O. Box 1346 Ann Arbor, MI 48106-1346 Copyright by
DEBRA S. LEE
2005
All Rights Reserved

TEACHER PERCEPTIONS OF CROSS-CULTURAL ADAPTABILITY AND

INSTRUCTIONAL PRACTICES IN INTERNATIONAL LOCATIONS

An Abstract of a Dissertation

Submitted

In Partial Fulfillment

of the Requirements for the Degree

Doctor of Education

Approved:

Dr. Greg Stefanich

Faculty Advisor

Dr. Susan J. Koch

Dean of the Graduate College

Debra S. Lee

University of Northern Iowa

December 2005

ABSTRACT

The purpose of the study was to investigate differences in cross-cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices of international teachers.

It is well documented that in American schools today, classrooms are becoming more diverse in regard to socio-economic status, ethnicity, race, intellectual ability, language, and religious heritage. School children today represent an incredibly diverse population. Conversely, the teacher population continues to be primarily female, White and from a monocultural background. This creates a mismatch between student and teacher backgrounds, perspectives, and cultural understanding, which can significantly impact student achievement.

Literature provides evidence of programs and experiences that have positive impact on the cross-cultural learnings and attitudes of preservice educators. However, there is limited research addressing cross-cultural effectiveness of classroom teachers and their instructional practices in regard to diverse classrooms.

Both quantitative and qualitative research methodologies were utilized to investigate respondents' cross-cultural effectiveness and teaching practices. All participants completed demographic questions and the Cross-Cultural Adaptability Inventory (CCAI). The international teachers also completed narrative, open-ended survey questions and in-depth interview questions. All data was collected via electronic, online communication links and analyzed.

Results of the study indicated that while the international teachers were more cross-culturally adaptive than the Iowa teachers on all four dimensions of the CCAI and on the overall score, only on the emotional resilience dimension was the difference statistically significant. Findings also indicated the international teachers were highly engaged in school, community, and cultural activities, connected personalized learning to student and family cultures, set high expectations for students, valued and respected students and their cultures, believed they could make a difference, used appropriate multicultural materials and instructional strategies, and exhibited great passion and commitment to their teaching. These findings directly link to the multicultural education research and further define the qualities of an effective multicultural teacher as perceived within the reality of international educational practice.

ACKNOWLEDGEMENTS

Those who hope in the Lord will renew their strength.

They will soar on wings like eagles;

They will run and not grow weary,

They will walk and not be faint.

Isaiah 40:31

Amidst the challenges of each day comes hope that lives in the Lord - the hope, faith, and knowledge that whatever journey we face, He will provide the strength to endure, a path to follow, and the energy to complete. My dissertation journey was not a journey that I took alone. Each step was supported by the unwavering support of my husband, Randy, my two sons, Matt and Andy, my parents, parents-in-law, extended family, and friends. They believed in me. I must add a very special thank you to Randy, my husband and best friend. He cooked the meals, cleaned the house, ran the errands, always made sure we had ice cream, and carried the burden of many duties we usually share to support my commitment to this project. Most importantly, he encouraged me in the most difficult times and loved me through the journey. I am very thankful for the international teachers who took time to participate in this research study and to my church prayer warriors who are special to my heart.

I also wish to thank my dissertation committee for their support and especially Greg Stefanich, my dissertation chair and friend. His steadfast support, knowledge, and guidance have been never-ending.

I am profoundly grateful to my family, my friends, and my work colleagues, who were the Lord's hands and heart as they helped me renew my strength, run and not grow weary, and walk and not grow faint. I am so very blessed. Thank you.

TABLE OF CONTENTS

	PAGE
LIST OF TABLES	viii
CHAPTER 1. INTRODUCTION TO THE STUDY	1
Overview	1
Purpose of the Study	7
Research Questions	8
Research Hypotheses	9
Definitions	10
Significance of the Study	15
Limitations and Delimitations of the Study	16
Limitations	16
Delimitations	17
Theoretical Perspective	18
Methods and Procedures Overview	19
Organization of the Study	20
CHAPTER 2. REVIEW OF LITERATURE	22
Introduction	22
Historical Foundations of Multicultural and International Education	22
Multicultural Education Foundations	22
International Education Foundations	43
Summary	48

PAGE

Diverse Student Populations Compared to the Teacher Population	50
The Link Between Teacher Characteristics and Student Achievement.	56
Effects of Cross-Cultural Field Experiences on Preservice and Classroom Teachers	63
Effective Educational Practices within a Multicultural Education Framework	75
Summary	87
CHAPTER 3. METHODS AND PROCEDURES	90
Introduction	90
Design of the Study	90
Demographic Questions	91
Inventory	91
Narrative, Open-Ended Questions	92
Discussion Questions	92
Measurement Instruments and Materials	93
Demographic Data	93
Cross-Cultural Adaptability Inventory (CCAI)	94
Narrative, Open-Ended Questions	97
Discussion Questions	92
Population of Interest	98
Data Collection and Analysis	99
Data Collection	00

	PAGE
Data Analysis	102
CHAPTER 4. RESULTS OF THE STUDY	104
Introduction	104
Demographic Characteristics of the Participants	104
Statistical Analyses of the CCAI Results	117
Narrative, Open-Ended Question Results	121
Question 1	123
Question 2	126
Question 3	129
Question 4	132
Question 5	135
Question 6	137
Question 7	138
Question 8	139
Discussion Question Results	140
Question 1	143
Question 2	150
Question 3	153
Question 4	156
Summary	158
CHAPTER 5. SUMMARY, CONCLUSIONS AND REFLECTIONS,	160

	PAGE
Introduction	160
Summary of the Study	160
Conclusions and Reflections	163
Research Question 1	163
Research Question 2	164
Research Question 3	165
Research Question 4	166
Research Question 5	166
Research Question 6	168
General, overall attributes of multicultural education	168
Multicultural teachers	169
Multicultural schools	173
Multicultural instructional materials	175
Multicultural instructional strategies	176
Social justice perspective	180
Research Question 7	181
Implications for Future Research	183
REFERENCES	188
APPENDIX A: LOCATIONS OF PARTICIPATING TEACHERS	198
APPENDIX B: DEMOGRAPHIC QUESTIONS	200
APPENDIX C: CROSS CULTURAL ADAPTABILITY INVENTORY	204

		PAGE
APPENDIX D:	NARRATIVE, OPEN-ENDED QUESTIONS	209
APPENDIX E:	DISCUSSION QUESTIONS	211
APPENDIX F:	E-MAIL INVITATION TO PARTICIPATE – INTERNATIONAL TEACHERS	213
APPENDIX G:	E-MAIL INFORMED CONSENT – INTERNATIONAL TEACHERS	215
APPENDIX H:	PARTICIPANT INSTRUCTIONS – INTERNATIONAL TEACHERS	219
APPENDIX I:	IOWA PRINCIPAL/SUPERINTENDENT PHONE SCRIPT	222
APPENDIX J:	E-MAIL IOWA TEACHER IDENTIFICATION – PRINCIPALS/SUPERINTENDENTS	224
APPENDIX K:	EMAIL INVITATION TO PARTICIPATE – IOWA TEACHERS	
APPENDIX L:	EMAIL INFORMED CONSENT – IOWA TEACHERS	231
APPENDIX M:	PARTICIPANT INSTRUCTIONS – IOWA TEACHERS	235
APPENDIX N:	CROSS-CULTURAL ADAPTABILITY INVENTORY DATA	237
APPENDIX O:	NARRATIVE, OPEN-ENDED QUESTION RESPONSES	249
ADDENINIY D	DISCUSSION OFFERDINGES	250

LIST OF TABLES

ΓABLE	PAGE
National Public School Comparison of Current and Projected Pre-K – 12 Enrollment by Racial/Ethnic Group	51
2 Iowa Public School Comparison of Pre-K – 12 Enrollment by Racial/Ethnic Group	53
3 School Locations of International Teachers	105
4 School Locations of Iowa Teachers	106
5 Age in Years of Teacher Participants	107
6 Level of Teaching (Matched Criteria)	108
7 Content Subject Areas of the Teachers	109
8 Total Years of Teaching Experience (Matched Criteria)	110
9 Total Years Taught in an International Location	111
10 Level of Education	111
11 Cultural Environment of Student Teaching Experience	112
12 Cultural Environment of School When Growing Up	113
13 Student Enrollment of Current School	114
14 CCAI – Descriptive Statistics	119
15 CCAI – Independent Samples t-Test Between International/Iowa Gr	oups120
16 Themes Revealed in the Narrative, Open-Ended Questions	122
17 Themes Revealed in the Discussion Questions	144
18 Teacher Perceptions of the Most Powerful Instructional Strategy	151

CHAPTER 1

INTRODUCTION TO THE STUDY

Overview

Every day in countless educational classrooms across America, instructional decisions are being made that affect students' perceptions of culture, family, friends, nations, and issues and conflicts facing the global world. These instructional decisions are even more important when a family, a community, or a nation deeply feels the effects of an economic, sociocultural, environmental, political, or technological change at a personal or global level (Merryfield, 2002).

Teacher and student thinking is influenced and instructional decisions impacted when a new automobile assembly plant moves into town, refugees resettle into a suburban community, manufacturing plants exit to other countries, differing religious groups live in the same neighborhood, or a nation is invaded or attacked. These all have considerable impact on instruction and learning (Merryfield, 2002). In addition, consider that it is possible "be hanging around a pub in New Delhi that serves Lebanese cuisine to the music of a Filipino band in rooms decorated with barrels of Irish stout," (Zwingle, 1999, p. 12) or watching Olympic sporting events from the comfort of one's own home while the events take place thousands of miles away in a foreign country.

The world is in the midst of a "worldwide reformation of cultures, a tectonic shift of habits and dreams called...'globalization," (Zwingle, 1999, p. 12), the term used to define vast changes in politics, business, health, entertainment, and cultural patterns as the world expands into one gigantic global community. "The interdependence of people

around the world with respect to peace, security, environmental protection, and economic participation is now abundantly clear" (Cummins, 2004, p. xv). In addition, "it is now possible for more people than ever to collaborate and compete in real time with more other people on more different kinds of work from more different corners of the planet and on a more equal footing than at any previous time in the history of the world – using computers, e-mail, networks, teleconferencing, and dynamic new software" (Friedman, 2005, p. 8). As part of globalization, modern technology appears to have no boundaries in creating tools that continue to bring diverse cultures physically closer and the United States is currently a "leading nation in the increasingly interdependent, intercultural, and technologically oriented global society of the 21st century" (Futrell, Gomez, & Bedden, 2003, p. 381). However, as a milestone in world development, other countries, such as India, are beginning to challenge America in competing for global knowledge work (Friedman, 2005). This will generate great responsibility for America's educational institutions.

Friedman (2005) tells of three great eras of globalization. The first process of globalization was generally during the age of discovery through about 1800, was driven by power – wind, horse, and steam, and was led by governments and countries. The second era was from about 1800 to 2000, was driven by multinational companies and was characterized by falling transportation and telecommunication costs. The third era of globalization, which we are currently in, is driven by the capability of empowered individuals and groups to collaborate and compete globally.

In the current globalization era, economically, nations and diverse groups of people depend upon one another more than ever before. National economies are intertwined with each other and to speak of individual country domestic economies is outdated. Economic activity is essentially global and we all inhabit one economic world (Hartoonian & VanScotter, 1996). Nations can ill afford to preserve a sentiment of isolationism as countries continue to depend more and more on each other. Technology and economics are the driving catalysts forcing nations to become close, dependent neighbors.

Globalization of advanced technology, demanding economics, the movement of populations around the world, and increasing cross-cultural contact (Cummins, 2004) results in a "deepening ethnic texture" (Howard, 1999, p. ix) in the United States as reflected in our society and in our schools. Cultural diversity is becoming more commonplace both in rural and urban areas (Howard, 1999). Children of many different races, ethnic groups, religious persuasions, and nationalities are now attending America's schools – together. Diversity abounds in our classrooms of today, and, with little exception, those of tomorrow will not have the historical homogeneous Anglo culture previously present. Children today represent an incredible variety of cultures, abilities, attitudes, languages, and beliefs, which all impact the classroom environment and combine to bring a far more complex perspective to American classrooms than ever before. The United States is now "more accurately described as a wonderful mix of cultures and races, old and new immigrants, exceptionalities and talents. America's cultural diversity is one of its national treasures, providing democratic communities of

learning not available in many other nations" (Futrell, Gomez, & Bedden, 2003, p. 381). This diversity is reflected in American classrooms as well.

The trend of enrollment in the United States' public schools continues to be one of a growing diverse population, with expected percentages in some urban areas of a diverse student population being the majority of students. At the same time, the background of educators in public schools and those in teaching preparation programs remains primarily White, female, and monocultural (Bennett, 2003; Cushner, 1998; Futrell, Gomez, Bedden, 2003; Gay, 2000; Gollnick & Chinn, 2002; Grant & Sleeter, 2003; Larke, 1990; Latham, 1999; Marshall, 1996; McIntyre & Byrd, 1996; Nieto 2002/2003; Samovar & Porter, 2001; Warren, 2002; Zeichner, Melnick, & Gomez, 1996). This remains true across the nation, as well as in Iowa, which is typically considered to be a rural, monocultural state (Iowa Department of Education, 2004). "The picture is clear: For the foreseeable future, the vast majority of teachers will be White while the student population will grow increasingly diverse" (Howard, 1999, p. 2). This suggests a "demographic imbalance between non-minority teachers and minority students" (Larke, 1990, p. 23) and indicates, "teachers will be working with students whose cultural backgrounds are different than their own" (Larke, 1990, p. 23).

The changing demographics in American classrooms present significant challenges for classroom teachers and teacher preparation programs. Preparing practicing and preservice educators, who originally come from a monocultural background, to teach in multicultural classrooms is a serious task and extremely relevant in today's teacher education and professional development programs. Studies show a

high positive correlation between a teacher's cultural sensitivity and cultural awareness, and achievement levels of minority students (Banks & Banks, 1997; Davis & Turner, 1993; Dilworth, 1998; Gay, 2000; Gomez as cited in Zeichner, Melnick, Gomez, 1996; Larke, 1990; McCall, 1995; Nieto, 2004; Vavrus, 1994; Warren, 2002). In addition, teachers who are effective in diverse settings generally exhibit high levels of cultural sensitivity (Campbell & Farrell, 1985, as cited in Larke, 1990; Cruickshank, 1986, as cited in Larke, 1990). If the expectation truly is that all children will succeed, then it is critical that classroom teachers have a conceptual background of multiple cultures and can understand the perspective from which their students enter the classroom.

Multicultural education provides a structural framework from which to approach the teaching of diverse students in American classrooms. Research cites numerous characteristics of multicultural education that address diversity issues and support high levels of student achievement. Multicultural education is neither neutral nor apolitical, but is positioned from within a social justice perspective, and validates and affirms differences in class, race, sexuality, gender, ethnicity, and ability. It empowers students to be successful learners and contributing citizens, engages teachers and students to view society through the lens of multiple perspectives, and nurtures positive self-concepts in students. Multicultural education fosters intergroup understanding and respect for others' cultures, questions stereotypes, is antiracist, promotes understanding of the dynamics of dominance, supports equal distribution of power among groups, and teaches about how power, discrimination, and injustice shape the world (Banks, 1999; Gay, 2000; Howard, 1999; Leistyna, 2002; Merryfield, 2002; Nieto, 2004; Reissman, 1994). Multicultural

identity and the identities of others, are committed to student learning, and function within a broader community connection. (Banks, 1999; Howard, 1999; Leistyna, 2002; Nieto, 2004; Reissman, 1994; Zeichner, Melnick, & Gomez, 1996).

Within the multicultural school, parents and community members are active partners in the educational process (Banks, 1999; Leistyna, 2002; Zeichner, Melnick, & Gomez, 1996) and the school climate reflects the full dimensions of multicultural attributes. In addition, curriculum, instructional materials, instructional strategies, and language employed in classrooms are reflective of multiple perspectives to meet student needs (Banks, 1999; Iowa Department of Education, 1989; Gay, 2000; Leistyna, 2002; Merryfield, 2002; Nieto, 2004; Zeichner, Melnick, & Gomez, 1996).

As student populations in American classrooms continue to become more diverse, there exists a mismatch, or "demographic imbalance" (Larke, 1990, p. 23), between non-minority teachers and minority students, which can lead to lower student achievement. Evidence in the literature suggests that cross-cultural learning experiences add significantly to the preservice educator's understanding of diverse cultures (Cushner, 1998; Dilworth, 1998; Gollnick & Chinn, 2002; Gomez, 1996; Jarchow, McKay, Powell, & Quinn, 1996; Larke, 1990; Stachowski & Mahan, 1998; Valli, 1996; Zeichner, 1996; Zeichner & Melnick, 1996; Zeichner, Melnick, Gomez, 1996). However, literature is limited in its attention to the classroom teacher's understanding of diverse cultures and, more specifically, instructional practices in international schools. The problem for investigation in this study was to explore the cross-cultural effectiveness of educators

who teach within Iowa and educators who teach in international locations and to investigate the perceptions of international teachers regarding instructional practices in their classrooms.

Purpose of the Study

The purpose of the study was to investigate differences in cross-cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices international teachers use in their classrooms to meet the needs of a diverse student population. Literature reveals multiple effects gained for preservice educators when they experienced teaching in an international setting, but literature is limited in regard to the effects of an international setting and diverse student populations on classroom teachers' cross-cultural effectiveness and resulting instructional practices. This study will partially fill the gap that exists in literature regarding how international cultural experiences inform classroom teachers' instruction with diverse student populations.

In a broader respect, as classrooms in the United States continue to become more diverse, school districts will need teachers who are able to help close the gap in the "demographic imbalance" that exists between the diverse backgrounds of students and the monocultural background of teachers. This study explored and specifically highlighted not only cross cultural effectiveness of international and Iowa teachers, but also helped define active instructional practices of international teachers within the

multicultural education framework. The results will provide valuable information for administrators when seeking candidates for teaching in diverse classrooms.

The study was not intended to be a verification of a previously determined idea or theory, but was intended to be a discovery leading to new insights regarding how an international community with diverse student populations informs teaching practices of educators.

Research Questions

This study involved both quantitative and qualitative avenues of investigation.

The quantitative portion of the study investigated differences in cross-cultural effectiveness between teachers in international locations and teachers in Iowa locations.

Differences were investigated in the overall cross-cultural effectiveness and in cross-cultural effectiveness regarding emotional resilience, flexibility/openness, perceptual acuity, and personal autonomy. The qualitative portion of this study was intended to provide a deeper exploration of instructional practices of international teachers and was positioned within the multicultural education framework of research. The link between the international teachers' cross-cultural effectiveness and instructional practices was also examined in this study.

Specifically, the following research questions guided the study:

1. To what extent is there a difference in overall cross-cultural effectiveness (Kelley & Meyers, 1995) between teachers in international schools and teachers in Iowa schools?

- 2. To what extent is there a difference in emotional resilience (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 3. To what extent is there a difference in flexibility/openness (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 4. To what extent is there a difference in perceptual acuity (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 5. To what extent is there a difference in personal autonomy (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 6. What instructional practices, within the multicultural education framework, do teachers in international schools use in their classrooms?
- 7. How does the cross-cultural effectiveness of teachers in international schools link to the instructional practices they use in their classrooms?

Research Hypotheses

The differences between the international teachers and the Iowa teachers in regard to cross-cultural effectiveness were hypothesized as follows:

1. The teachers in international schools will be more cross-culturally effective overall (Kelley & Meyers, 1995) than the teachers in Iowa schools.

- 2. The teachers in international schools will be more cross-culturally effective in regard to emotional resilience (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 3. The teachers in international schools will be more cross-culturally effective in regard to flexibility and openness (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 4. The teachers in international schools will be more cross-culturally effective in regard to perceptual acuity (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 5. The teachers in international schools will be more cross-cultural effective in regard to personal autonomy (Kelley & Meyers, 1995) than the teachers in Iowa schools.

Definitions

For the purposes of this study, the following definitions were used.

<u>Classroom Teacher</u>: A certified provider of instruction in an international or Iowa school who practices the profession as defined in the Iowa Teaching Standards (Iowa Department of Education, 2002). Classroom teachers are also called practicing teachers.

<u>Cross-Cultural Effectiveness</u>: The interactions, connections, communications, and understandings between people of different cultures. This term can also be called cross-cultural competence, adaptability, or success and is used to define persons who have the skills, attitudes, and perspectives that honor and respect others who are from different backgrounds. Kelley and Myers (1995) identify four key components to cross-cultural effectiveness as having emotional resilience, being flexible and open, having perceptual acuity, and having personal autonomy.

<u>Cultural Pluralism</u>: The right to maintain group languages and cultures while combining with others to form a new society reflective of multiple differences (Nieto, 2004, p. 437).

<u>Cultural Sensitivity</u>: The attitudes, beliefs, and behaviors of teachers toward students of other cultures (Larke, 1990).

<u>Culture</u>: "The values; traditions; social and political relationships; and worldview created, shared, and transformed by a group of people bound together by a common history, geographic location, language, social class, and/or religion" (Nieto, 2004, p. 436).

<u>Demographic Imbalance</u>: A term used to indicate a mismatch, or a difference, between the cultural backgrounds of teachers and students (Larke, 1990; Latham, 1999).

<u>Diversity</u>: Differences in race, ethnicity, socio-economic status, language, culture, learning styles, and learning abilities (Futrell, Gomez, & Bedden, 2003). Interindividual differences such as "geographic race of origin, ethnic group, regionality, socioeconomic status, gender and sexual orientation, spirituality and religion, abilities and handicapping conditions, language, age, literacy and language development, worldviews, and experiences" (Huber, Kline, Bakken, & Clark, 1997, p. 136). "The appreciation of human differences and the belief that, in order for students to think critically – especially about life circumstances and opportunities that directly and indirectly impact their lives and the lives of their family members, community and country – they must affirm both social diversity (cultural pluralism) and human diversity" (Grant, 1995, p. 4).

Ethnic/Ethnicity: "A group that shares a common history, a sense of peoplehood and identity, values, behavioral characteristics, and a communication system. The members of an ethnic group usually view their group as distinct and separate from other cultural groups within a society" (Banks, 1999, p. 115).

Globalization: A conceptual situation in which world cultures continue to grow closer in understanding, interdependence, and physical connections, and multiple cultures live in a community or location.

Hidden Curriculum: "The unintended curriculum – what students learn from the school's culture and climate" (Glatthorn, Carr, & Harris, 2001, p.1). The "subtle and not-so-subtle messages that are not part of the intended curriculum, may also have an impact on students" (Nieto, 2004, p. 42). In this study it includes the perceptions, values, and beliefs teachers have regarding students that affect expectations teachers have of students.

<u>Instructional Practices</u>: The behaviors and activities of the teacher that promote student learning and improved achievement.

International School/Educational Setting: An accredited school district/system that is located outside the physical boundaries of the United States, uses an English-language, American curriculum and is listed in *The ISS Directory of International Schools*, 2003-2004, or is within the Department of Defense Dependents Schools system. This study included teachers from thirteen different international countries.

<u>International Teacher</u>: A teacher in a full-time teaching position in an international school located outside the United States.

<u>Iowa Teacher</u>: A teacher in a full-time teaching position in a public school located within the state of Iowa and, for the purposes of this study, teaching in a public school system within the (Iowa) Area Education Agency 267 (AEA267).

<u>Iowa Teaching Standards</u>: Eight standards and forty-two criteria that establish the qualities and expectations of an effective teacher in the state of Iowa. Developed by the Iowa Department of Education (2002).

Multiculturalism: "A philosophical position and movement that assumes that the gender, ethnic, racial, and cultural diversity of a pluralistic society should be reflected in all of the institutionalized structures of educational institutions, including the staff, the norms and values, the curriculum, and the student body" (Banks & Banks, 2005, p. 451). "A process of preparing students for meaningful participation in a diverse world and for assisting them in affirming their own unique cultural backgrounds while respecting others" (Reissman, 1994, p. 3). It is also known as "cultural diversity" and "ethnic studies" (Reissman, 1994, p. 2). The key tenets of multiculturalism include fostering "intergroup understanding, equity, excellence in subject mastery, student knowledge and respect for others' cultures, education for social justice, antiracist education, and the development of proactive community, national, and global citizenship" (Reissman, 1994, p. 2).

Multicultural Education: A process of basic education for all students that is antiracist, important for all students, pervasive and comprehensive in a total school environment, affirms ethnic, racial, linguistic, religious, economic, ability, and gender pluralism, and promotes democratic principles of social justice (Nieto, 2004, p. 346). "A

reform movement designed to change the total educational environment so that students from diverse racial and ethnic groups, both gender groups, exceptional students, and students from each social-class group will experience equal educational opportunities in schools, colleges, and universities" (Banks & Banks, 2005, p. 451). "Recognizing and respecting cultural differences and similarities in groups or individuals and incorporating such recognition and respect in educational policies, programs, and practices." (Garcia, 1984, p.6). It also includes the planned curriculum and instruction that educates students about society's diversity and includes all levels of the affective and cognitive domains, knowledge acquisition, higher order thinking skills, interpersonal and intergroup relations, and expectations for achievement (Iowa Department of Education, 1989).

<u>Preservice Teacher</u>: Students enrolled in institutions of higher learning majoring in education and intending to become a classroom teacher.

Student Teacher: Students enrolled in institutions of higher learning, in training to become educators, and engaged in practice teaching in a school under the direction of a classroom teacher.

Social Justice: "Teachers consider (and encourage student inquiry about) inequitable social structures; images of race, culture, class, and gender in popular cultures; and social action to bring about greater societal equity, both locally and globally" (Bennett, 2003, p. 2).

Teacher Efficacy: "A teacher's belief about the power he or she has to produce an effect on students" (Pang & Sablan, 1998, p.42) and the belief they can change a situation.

Significance of the Study

As classrooms within Iowa and across the nation continue to become more diverse, and as teachers continue to come from monocultural backgrounds, there will continue to be disconnect between those students of diverse backgrounds and the educators who teach them, which may impact academic achievement. Evidence in the literature suggests that cross-cultural learning experiences add significantly to the preservice educator's understanding of diverse cultures (Cushner, 1998; Dilworth, 1998; Gollnick & Chinn, 2002; Gomez, 1996; Jarchow, McKay, Powell, & Quinn, 1996; Larke, 1990; Stachowski & Mahan, 1998; Valli, 1996; Zeichner, 1996; Zeichner & Melnick, 1996; Zeichner, Melnick, Gomez, 1996). However, literature is limited in its attention to the classroom teacher's understanding of diverse cultures and, more specifically, instructional practices in international schools.

Research indicates a need for an investigative study of this type. Craft (1996) suggests, "there is very little evidence about practices and strategies that enable the development of greater intercultural teaching competence" (p. 153) and Dilworth (1998) states, "we know very little about how teachers approach issues of diversity in their own classrooms (p. 184). Zeichner, Melnick, and Gomez (1996) maintain, "another important issue related to teacher education for diversity is the question of teacher development over time" (p. 162). They continue by stating "there is very little discussion in literature of how this learning should be related to a teacher's career over time" (p. 162) and "much more work needs to be done to look at the process of teacher education for diversity developmentally" (p. 162). Lastly, Zeichner and Hoeft (1996) strongly state that "a

priority in research on teacher education for cultural diversity should be to investigate how particular kinds of experiences for teachers at the preservice or in-service levels are connected to the character and quality of their teaching" (p. 541).

This investigation partially answers the call for this type of research by providing information regarding the cross-cultural effectiveness of teachers in Iowa and in international locations and by providing information regarding the multicultural instructional practices in international schools beyond the preservice educator level. In addition, the results of this study will highlight instructional practices, within the multicultural education framework, in international schools, which will contribute to guiding professional development for classroom teachers in an effort to support skill enhancement for teaching students from diverse backgrounds. This, in turn, will improve student achievement.

<u>Limitations and Delimitations of the Study</u>

Limitations

Results of this study were limited to the self-perceptions of the participants. In regard to the elements being investigated in the study, the inventory, survey, and discussion questions were completed as self-response, volunteered information, as opposed to information gathered from observational techniques.

A second limitation of the study was reflected in the fact that no science teachers and no specialist area teachers were represented in the sample population of the Iowa teachers, while those teacher populations were represented in the international teachers.

This should not affect the results of the study, but reinforces that the results need to be generalized with care.

Delimitations

The study was limited to Internet communication techniques. Due to the international locations of the one sample population group of participants, circumstances prohibited synchronous time communication, and all participants had to possess or have access to Internet, online communication tools. This included email and Internet web access. Initial and continuing contacts with the participants and with the school directors and principals for the purposes of providing information, directions, and collecting responses were completed via technology tools. Initial contact was through email, the inventory and narrative survey questions were completed through an online survey tool, and the discussion questions were completed through the use of email. In addition, expenses associated with personal, real time interaction would have been prohibitive to the purposes of this study.

This study was limited to 20 participants teaching in various locations around the world and 20 participants teaching in Iowa. Because of the limited number of participants in comparison to the actual total number of teachers throughout Iowa and in international schools worldwide, the findings should be generalized with caution. In addition, the study was limited to volunteer classroom teachers who were currently teaching in an international school, in grades kindergarten through 12th grade, who serve diverse populations of students, and were willing to participate in the study. They were known to the researcher and were from locations identified in the *ISS Directory of*

International Schools, 2003-2004, or were from within the Department of Defense Dependents Schools system (DoDDS). This study was also limited to volunteer classroom teachers who were currently teaching in grades kindergarten through 12th grade in Iowa schools within the Area Education Agency 267 district.

This study was intended to investigate cross-cultural effectiveness of the participants and the perceptions regarding the instructional practices with diverse student populations. In view of the fact there is limited research available on the topic, this study was intended to be an investigative beginning point for further research.

Theoretical Perspective

To investigate the cross-cultural effectiveness of all participants and to examine instructional practices of the international teachers, this study employed the theoretical perspectives of social constructivism (Goodson, 1990; Kelly, 1963) and personal practical knowledge (Connelly & Clandinin, 1988). Goodson (1990) indicates that teaching practices are constructed and reconstructed over and over again within a variety of different experiential and interactive arenas. Kelly (1963) identifies this as building personal constructs, or patterns from which to view the world that are adjusted when major experiences occur and new constructs are incorporated. Connelly & Clandinin (1988) affirm that experiences build the kinds of meaning teachers construe and the ways they personally interpret the world. For that reason, it is important that teaching be looked at from the point of view of those involved – their personal practical knowledge.

As educators build constructs about teaching through their preservice and teaching experiences – background cultures, practice teaching experiences, university

classes, classroom encounters, encounters with others, etc. – they are constructing theoretical perspectives, or personal practical knowledge, which is changed as experiences change. As a participant noted in McKay and Montgomery's study (1996-97, p. 28), "teaching is learning." International setting and diverse student populations act as major experiences adjusting teachers' constructed ideas of teaching and accrued meanings of instructional practices. When teachers live and teach in communities that are much like the cultural backgrounds in which they grew up and attended schools, they are more likely to interact exclusively with cultures familiar to them. International settings and diverse student populations that are different than a teacher's monocultural background provide "ongoing opportunities to learn about another culture, to test the limitations of personal theories for teaching other cultures, and to deepen understanding about the influence of teachers' and students' cultural values on classroom instruction" (Jarchow, McKay, Powell, & Quinn, 1996, p. 183). Within the social constructivist framework, the international setting and diverse student populations provide teachers with experiences to acquire personal constructs for how cultures influence instructional practices (Jarchow, McKay, Powell, & Quinn, 1996).

Methods and Procedures Overview

This study compared perceptions about cross-cultural effectiveness of classroom teachers in international settings with classroom teachers in Iowa, and examined instructional practices perceived by international teachers to be effective in meeting the needs of a diverse student population within a multicultural education framework. Both quantitative and qualitative methods were used to provide a framework for establishing

the respondents' cross-cultural effectiveness and for providing deeper, rich description of teaching practices. The three data gathering methods included (a) a published inventory relating to cross-cultural adaptability, (b) narrative, open-ended survey questions specific to teaching practices and (c) in-depth discussion questions that further investigated the strands from the survey's narrative, open-ended questions.

Organization of the Study

The purpose of the study was to investigate differences in cross-cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices international teachers perceive to be effective in meeting the needs of a diverse student population. In Chapter 1, the introduction provides an overview of the study's foundation and purpose, research questions, definitions, significance of the study, limitations and delimitations, theoretical perspective upon which the study was based, and methods and procedures overview.

Chapter 2 provides the review of literature related to (a) historical foundations of multicultural education and international teaching, (b) diverse student populations compared to the teacher populations, (c) the link between teacher characteristics and student achievement, (d) effects of domestic and international field experiences on preservice and classroom educators, and (e) effective educational practices within a multicultural education framework. The research studies and literature reviewed in this chapter provided the historical and theoretical foundation upon which this study was based.

Chapter 3 describes the methodology and procedures used to complete the research study and answer the research questions. Included in this chapter is the design of the study, the measurement instruments and materials, the population of interest, the procedures that were used to collect the data, and explanation of how the data was analyzed.

Chapter 4 presents the data results including descriptions of the demographic characteristics of the participants, statistical analysis of the inventory, and analysis of the narrative, open-ended questions, and discussion questions.

Chapter 5 provides the summary of the results, conclusions and reflections, and implications that can be drawn from the analyzed data, including suggestions for further research.

CHAPTER 2

REVIEW OF LITERATURE

Introduction

The purpose of the study was to investigate differences in cross-cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices international teachers perceive to be effective in meeting the needs of a diverse student population. In this chapter the review of the literature relevant to the study was organized into five areas plus a summary. The five areas provide a foundation for the focus and purpose of the study and include (a) the historical foundations of multicultural education and international teaching experiences, (b) the statistical picture of a growing diverse student population and monocultural teacher population at both a national and state level and the mismatch which leads to the need for the study, (c) the link between teacher characteristics and student achievement, (d) effects of domestic and international field experiences at the preservice educator and classroom teacher levels, and (e) what constitutes effective educational practices within a multicultural education framework.

Historical Foundations of Multicultural and International Education Multicultural Education Foundations

From its very beginnings, the United States has been a very pluralistic and segregated society. The colonies were composed of a variety of European nationalities and religious sects and were surrounded by various land areas populated by indigenous people who were themselves different in regards to language, culture, and economic

activities. From this early beginning the notion of exclusion was closely entwined with democracy and the privileges of citizenship were bestowed only upon European-descent males.

During the colonial era, even though there were regional variations between the schools of the northern, middle, and southern colonies, the socialization and education of all children in the three regions were largely influenced by family, religious commitment, and English political ideas (Ornstein & Hunkins, 1993). Private educational institutions remained the most dominant form of schooling during the colonial period and served a population that was very literate (Sergiovanni, 1982).

Schooling was linked closely with popular government and political freedoms and leaders of the period recognized the need of having a literate and informed public to advance the cause of the new republic (Ornstein & Hunkins, 1993). Schooling, however, was not necessarily the only vehicle through which an education was obtained. One could learn several professions through a variety of ways including teaching oneself through self-study or studying with a mentor (Sergiovanni, 1982). This set the framework for the earliest field experiences in teacher education.

From the colonial era and through the early period of the nineteenth century, the original field experience in teacher preparation emerged as a type of apprenticeship program. The preservice teacher was usually indentured to a schoolmaster to learn the profession and did not receive a liberal arts education, as is the case today, nor studied subjects very much in depth. The indentured servant practiced the craft alongside the master until such time it was decided the preservice educator was qualified to be a

teacher (Slick, 1995). Even in this early time period, it is possible to observe how the ideas of the dominant societal group were passed on through the schools. The new teacher would be socialized through the mentor and would reflect those political and cultural ideas of the dominant societal groups, which were the local town leaders, and parents of students. In addition, early curriculum did not take into consideration the multicultural make-up of the student population at that time (Appelbaum, 2002).

In 1790, the Naturalization Act was passed, which denied citizenship and the right to own property to African and Asian immigrants and Native Americans. This law remained for over 150 years and was used by both state and federal governments to maintain a White dominant culture (Appelbaum, 2002). Schooling for the minority groups, if they were provided schooling at all, took place in legally established segregated schools (Seeberg, Swadener, Vanden-Wyngaard, & Rickel, 1998).

Within a strong nationalistic period that emerged during the Revolutionary period and continued into the 1800's, there was a shift in who controlled the school systems. State authority strengthened and public schooling became clearly defined as an institution that was controlled and financed publicly (Sergiovanni, 1982). As new knowledge increased and the amount of knowledge grew, education became more closely linked with schooling and the focal point of authority relative to education shifted from the parent to the state. An outgrowth of more state control was a bureaucratic, corporate structure of authority over the educational system. Standardization and efficiency became the primary educational institutional goals and state representatives worked at

carrying out the educational mandates. The schools, teachers, and supervisors became part of the emerging modern world of social engineering (Sergiovanni, 1982).

The early nineteenth century education included African American schools in Boston that were successful, Native American run schools that produced nearly 100 per cent literacy rates, segregated Chinese schools in California, German schools in the Midwest, and Swedish culture schools, among others. When decisions were made by Whites to take over the African American and Native American schools, performance decreased and literacy rates fell (Appelbaum, 2002). This reflects one of the first examples of the cultural gap between teachers and students resulting in lack of student achievement.

As the latter half of the nineteenth century brought about a significant change in the social structure of the country through industrialization expansion, the family began to lose its function as producer-consumer and instead became consumers within a factory system of employment. This strengthened control of the school to state authorities and away from the parent. In addition, at the same time, Horace Mann led the growth of the common school movement, which began in Massachusetts with the passage of state laws that required every town to form a school board that was charged with the responsibility for schools in the local area (Ornstein & Hunkins, 1993). The common school was attended by lower and middle class students and was supported by public funding. It provided a common framework of shared understandings, language, and common values believed to be necessary to bring about cultural homogeneity as a response to class and ethnic diversity (Hurn, 1993). In an age of increased immigration and westward

expansion, the common school primarily served an integrative purpose. The school would create a stable society, would socialize immigrants of ethnic and religious diversity, would favorably sustain the country's political and economic well being and would contribute to a democratic society (Ornstein & Hunkins, 1993).

The common school was believed to be a critical factor in the American system of equality and opportunity, supported a sense of community that all Americans could share, and promoted the national identity. Mann also believed this type of school system would be the great equalizer in providing a means of social mobility for children of workers and farmers and contribute to social democracy (Sergiovanni, 1982). This was the foundation of the American school system, as we know it today and thus the underpinnings of the socio-cultural context of our current educational and multicultural educational framework.

In Chicago, in 1889, Hull House was founded by Jane Addams and was one of the early twentieth century efforts toward intercultural education addressing pluralism and helping immigrants adjust to the American way of life (Appelbaum, 2002). It was at this time that Addams and John Dewey, along with others, wrote that along with learning the common culture, which included English, U.S. history, and U.S. political system, the home cultures of immigrants should be maintained. These culturalists believed the school curriculum should include "handicrafts and occupations, cultural traditions, folk songs and folklore, and the stories of nonwhite and immigrant students" (Appelbaum, 2002, p. 29.

Mirroring the social context of the times, the normal school, institutions of higher learning designed to prepare those who were interested in becoming public school teachers, emerged in the late 1800's and early 1900's (San Jose State University, 2005). Generally, the program was a two-year course of study that culminated in a field experience taking place in a model school that was associated with the normal school. Student teachers were placed to study, observe, and practice the teaching art (Slick, 1995). Following into the twentieth century, preservice education continued to follow the patterns and changes of education in general. This framework provided the basis for training preservice educators into the dominant culture of society, thus perpetuating the system.

The Progressive Era, characterized by the prominence of democratic principles in schools, dominated the thinking of the majority of the 1900's. Educational beliefs in this era were rooted in the work of Horace Mann and later John Dewey. Progressivism heavily promoted the ideology that education should promote democratic living by including problem-solving methods, scientific inquiry, cooperative behaviors and self-discipline. There was heavy emphasis on not what to think, but how to think and the teacher's role was that of a guide (Ornstein & Hunkins, 1993). Society was shaped by the incredible growth of corporations and the business world during this time. This established the American system of mass production, distribution, and consumption (Sergiovanni, 1982) and was the economic basis for the expansion of schools through the secondary level. Oakes (1986) points out that in the turn-of-the-century America and on the back of the industrial revolution, there was a population shift to urban centers and an

incredible increase of immigrants who were poor, unskilled, and could not speak English. It was believed by the dominant culture that these immigrants needed to be Americanized as quickly as possible and trained to fit into the rapidly expanding industrial society (Sergiovanni, 1982), which was accomplished, in part, through schools. According to Oakes (1986), students were in schools for two purposes; the advantaged group who were destined to become scholars and the disadvantaged group who needed more time to prepare before entering factory life. This reflects the marriage of the business world with schooling – a concept that is still prevalent today – and the foundational belief that education is related to economic production, work, the superiority of the dominant culture, and the value of meritocracy.

It is interesting to note the development of the international perspective in the American education system as it paralleled the social system. Early American leaders, Washington, Jefferson, Madison, and others, voiced strong opinions against involving America in the internal affairs of other nations. They also saw, however, the merit of staying on friendly terms with other countries. Thus, for nearly two centuries, foreign involvement was, for the most part, based on a policy of isolationism that was interrupted, basically, only by war. This policy was characterized through westward expansion and the idea of Manifest Destiny and in an attitudinal perspective of moral righteousness regarding the foreign policy of isolationism (Gutek, 1993).

The most significant international perspective connection to the evolving educational and social structure over the last two centuries of history is the common school movement and the theory of assimilation of immigrants. The common school

movement was based on a structural theory unique to America in that it was founded on state authority with local control. The influence of European educators, such as Pestalozzi and Montessori, impacted only in areas of methodology. The governance and organization depicted in the common school concept upheld a cultural isolation.

Certainly, there were those who were educated and had experiences with foreign cultures, but overall, American education had a minimized international dimension (Gutek, 1993).

The assimilation of immigrants could very well be subtitled "Americanization." As a predominant ideology of education, as previously indicated, the desirability of the dominant culture was to Americanize, or assimilate, immigrants into the English language and Protestant values, thereby separating immigrant children from their ethnic heritage. A homogeneous melting pot was the goal and pluralistic influences were not desired, thus shaping education and curriculum and establishing a hidden curriculum. Again, it is seen that American education minimized cultural diversity and international dimensions, and opportunities for cultural exchanges and expanding views that normally develop from multiculturalism were purposely lost (Gutek, 1993).

World Wars I and II eventually brought the United States into world affairs and the country has been very active from a global perspective since then, although educators did not necessarily take much of a lead in social reform movements during that era. In the decades of the 1930's and 1940's, some teachers participated in intercultural programs that endeavored against racism, prejudice, and discrimination, but most of the programs were instigated from outside the educational arena and imposed on the school bureaucracy (Sergiovanni, 1982). Intercultural and international education started to

become popular at this time, in a limited way, with social reformers and some educators due to the concern about dangers of international conflict. It was believed that, as the diversity of the population increased, it was important for people to learn about other cultures and ethnic groups and that this understanding would help reduce, or eliminate, international and domestic conflict (Appelbaum, 2002).

Even though public international efforts lagged, private efforts had been occurring since the beginning of the century. Philanthropic foundations established by industrial giants such as Carnegie, Rockefeller, Guggenheim, and Kellogg were responsible for supporting scholarships focused on foreign study and international exchanges. They also supported initiatives creating centers of international study and area studies programs (Pickert, 1992).

In 1928, the federal government commissioned the Merriam Report, which indicated a need to change the deculturalization approach to Native American education. Previously, it was believed that the education of Native American children needed to be accomplished by removing the child as far as possible from his/her home environment. The Merriam Report, however, emphasized the value of raising and educating children in the natural setting of home and family, which completely reversed previous beliefs. The idea of family involvement as an integral part of education reemerged in the late 20th century as part of multicultural school reform (Appelbaum, 2002).

At this time, various approaches are noted that started to address intercultural and multicultural education. "In 1933, the Service Bureau for Intercultural Education was developed in the United States, and thus was born the beginning of the intergroup

education movement" (Cushner, 1998, p. 5). Throughout the next several decades, this movement focused on reducing racial and ethnic tensions among citizen groups in the United States. Various activities included developing and presenting inservice programs and school assemblies to raise consciousness of minority and immigrant children. The movement introduced curricular methods and instructional units of study specific to minority groups, worked to ban books that reflected stereotypes and demeaned religious, ethnic, and racial groups, and founded many institutions and organizations whose focus was devoted to intercultural pursuits. One example is the American Field Service Intercultural Programs, which today is one of the largest international student exchange organizations worldwide. Its roots can be traced to the American Field Service, a volunteer ambulance service on the front lines after World War II and originating with the Service Bureau of the 1940's (Cushner, 1998).

After the World Wars, the creation of the United Nations gave great impetus to the interest in foreign cultures and the United States government also became involved. The Fulbright Act of 1946 was the first instance of an international educational exchange program sponsored by the government. It created an exchange of students and professors between 49 different countries and helped to bring an awareness of diverse cultures to the educational system (Michie, 1967).

Additional international educational activities increased until the Vietnam War era and the 1960's, and 1970's, which brought about great concentration on domestic issues and significantly limited foreign studies. International education emerged again in the late 70's and early 80's through issues of cross-cultural understanding, multicultural

sensitivity (Pickert, 1992), and a growing awareness of the global community. It is interesting to note, however, that these programs, for the most part, did not extensively influence public education and the hidden curriculum of the dominant culture thrived, continuing to support the cultural gap between teacher and students.

Domestically, the U.S. Supreme Court case Brown v. Board of Education (1954) heralded a landmark decision that called for an end to public school segregation and overturned previous legislation supporting separate but equal education. The decision provided that in public education there was no place for the doctrine of 'separate but equal' education and that separate schools are inherently unequal in their nature (Appelbaum, 2002). On the heels of this decision, the civil rights movement brought great focus to the racial injustice in the educational arena. Also, mass media had evolved in the 1950's and 1960's to the extent that it played a dominant role in bringing local civil rights issues to the forefront in national news. It was now possible to "force many people in the United States to confront racism and the legacy of oppression arising from slavery" (Appelbaum, 2002, p. 30).

At this time, in the 1960's and 1970's, the general nation-wide social ferment of civil rights movement supported the goal of African Americans to eliminate discrimination in education, employment, housing, and public accommodations. This movement had significant influence on the enactment of numerous educational reforms, including "school desegregation, multicultural and bilingual education, mainstreaming of students with special needs into regular classrooms, and gender-sensitive education" (Cushner, 1998, p. 6). Ethnic groups demanded that the curriculum in schools be

reformed to reflect the experiences, cultures, histories, and perspectives of the various ethnic groups. In addition, there was the demand that educational institutions hire more minority teachers and administrators, that more control of schools be given to local communities, and that textbooks be revised to more truly reflect the diversity present in the United States (Banks & Banks, 1997).

The civil rights movement increased in intensity, which drove federal legislation leading to the Civil Rights Act of 1964. The most important section of this Act, to this study, required that "no person, on the basis of race, color, or national origin, could be excluded from or denied the benefits of any program receiving federal financial assistance" and "established the precedent for using the disbursement of government money as a means of controlling educational policies" (Appelbaum, 2002, p. 30). This meant that if institutions did not comply with the mandates in the legislation then federal funding could be withheld. Also, the 1965 Immigration Act removed the quota system and earlier immigration restrictions. This opened the door to a new wave of immigrants at the same time that African Americans, Native Americans, Mexican Americans, and Puerto Ricans were demanding their cultures be included in the curriculum of public education (Appelbaum, 2002).

As alternatives to the traditional organized schools, Afrocentric and Latinocentric schools became visible and, as more Spanish-speaking refugees began arriving in Florida from Cuba, there was an increased interest in bilingual education. Eventually, this led to the Bilingual Education Act of 1968. In some areas, bilingual education evolved into bicultural education, which was designed to assist students to understand their own

cultures as well as the dominant culture. Segments of this idea supported multicultural curricula (Appelbaum, 2002).

In the mid 1970's, the United States Supreme Court (Lau v. Nichols, 1974) directed language education for students who were not proficient in English. The premise behind this mandate was that students needed to be able to speak and understand the language of the instruction in order learn and that providing instruction only in English was not equal treatment for these students. Requiring mastery of English skills before they could fully participate in the educational process unfairly handicapped them. Bilingual education programs and English as a Second Language programs grew out of this time period to meet the needs of students (Appelbaum, 2002).

The civil rights movement had significant success and in the face of a liberal national atmosphere other groups stepped forward to take actions to eliminate discriminatory practices and to demand that the educational process respond to their needs and differences. Women's rights activists, people with disabilities, senior citizens, and gay rights advocates all demanded, in various degrees, that institutions be reformed to eliminate discrimination, provide equal opportunities, open doors for employment, and acquire human rights. The Rehabilitation Act of 1973 protected employment rights for those with disabilities but only for businesses that received federal monetary assistance, so was limited in its applicability. The 1975 Individuals with Disabilities Education Act guaranteed public education for children with a broad range of disabilities, but only with the 1975 Education of All Handicapped Children Act was the right of equal opportunity for education guaranteed to all children with disabilities (Appelbaum, 2002). This act

also brought the idea of least restrictive environment and mainstreaming to the educational arena and was "the most significant legal victory of the movement for the rights of students with disabilities in education" (Banks & Banks, 1997, p. 6).

All of the programs, legislative acts, and changes in education and society as a whole "recognized the pluralistic nature of the society in a positive rather than a negative sense. Each attempted to help some educationally disadvantaged group to receive a better education within a pluralistic framework" (Cushner, 1998, p. 6). Society was changing to encompass a more pluralistic society. However, Appelbaum (2002) suggests that during the 1980's there existed a type of culture wars in which society actually experienced cultural ambivalence. From various perspectives, one school of thought maintained that society needed to preserve the standards and principles that are the foundation of essential national morals, suggesting one predominant culture needed to be preserved to the exclusion of others. Another perspective argued that a student's cultural ethnicity had little to do with defining self-identity and cultural meaning. Yet another perspective contended that students could learn about various cultures but that did little to foster growth in reading, writing, and mathematics. And still others argued that multiculturalism was morally important and teaching about others' cultures did not prevent students from participating in the mainstream culture (Appelbaum, 2002).

Education came galloping into the 1990's riding on the waves of the previous decades. Multicultural education emerged out of those previous decades and was defined from the "diverse courses, programs, and practices that educational institutions devised to respond to the demands, needs, and aspirations of the various groups" (Banks & Banks,

1997, p. 6). There is actually no specific course of study, no actual practice, nor an educational practice that can be identified as being multicultural education. Rather, it is a broad term that classroom teachers use that describes a wide assortment of practices and programs that support and are related to educational equity, those with disabilities, women, ethnic groups, those of low socio-economic status, and language minorities. The meaning and how the term is referred to are most likely to differ from one school district to another (Banks & Banks, 1997). However, the more comprehensive definition of multicultural education that has arisen out of the last several decades and has come to be accepted by those authors in the field is that multicultural education is "an educational reform movement whose major goal is to restructure curricula and educational institutions so that students from diverse social-class, racial, and ethnic groups – as well as both gender groups - will experience equal educational opportunities" (Banks, 2002, p. 123).

The past decade brought much attention to multicultural education and efforts to understand how to meet the needs of students. A multitude of organizations have arisen that support multicultural efforts in education. Included in this list is the National Association for Multicultural Education (NAME), founded in 1990. This organization was founded for the express purpose of bringing together educators who are interested in multicultural education and hosts an annual conference as well as publications and various resources for teachers (Appelbaum, 2002). Other organizations include the National Indian Education Association (NIEA), the Mexican American Legal Defense and Educational Fund (MALDEF), Teaching Tolerance Organization, National

Association for Bilingual Education (NABE), Regional Laboratories such as the North Central Regional Education Laboratory which houses various resources that address multicultural education, the National Council for Language and International Studies, and the Center for Research on Education, Diversity, and Excellence, which is federally funded through the Office of Education for Research and Improvement (OERI). The REACH Center (Respecting Ethnic and Cultural Heritage) has grown out of Gary Howard's work in diversity. One of the major focuses of the Center is "to help White educators reconceptualize their curriculum and their pedagogy from a multicultural perspective" (Howard, 1999, p. 18). Many, many organizations are now contributing to the issue of multicultural education.

Also gaining strength in the 1990's and continuing today, Gloria Ladson-Billings, Sonia Nieto, and Lisa Delpit write about and work in multicultural education from sociocultural and sociopolitical approaches. The basis for their work is that "both the content and structures of schooling connect with social class and racial stratification" (Appelbaum, 2002, p. 34). Beliefs, actions, and choices, based on cultural knowledge, habits, and tastes, have exchange values similar to the power of dollars. This is called cultural capital by sociologists. Cultural capital can buy acceptance or favored status in jobs, in schools, and in social interactions. Delpit (1995) identifies the culture of power, or the codes or rules for participating in power. These "relate to linguistic forms, communicative strategies, and presentation of self; that is, ways of talking, ways of writing, ways of dressing, and ways of interacting" (p. 25) Thus, students are learning particular codes of power in order to maneuver in society (Appelbaum, 2002).

Jeannie Oakes and Martin Lipton, also at this time, work from a sociocultural perspective that expects differences and celebrates those differences. Many of the aforementioned authorities, along with Cushner, Gay, Banks and Banks, Grant and Sleeter, Howard, and others in the field of multicultural education work from a social justice perspective, which advocates that educators have a responsibility to work for change that promotes effective and positive multicultural education.

Also in the decade of the 1990's, White studies and education for a global society took firm root. White identity became a research area by African American psychologists. The theoretical base included race as a dimension of personal identity as well as the idea that "race consciousness was an essential predisposition to eradicating racist policies and practices in schooling" (Appelbaum, 2002, p. 36). This was especially important since the vast majority of teachers were White (Appelbaum, 2002).

As the world continued to move at a rapid rate to becoming one global community, it was recognized in the 1990's that the rise of technology, especially the Internet, and globalization gave pause to educators to consider the need for students to understand global cultures (Appelbaum, 2002). As transportation and enhanced communications became more accessible, people around the world traveled more, industry and business became more diversified locating to various countries, and the need for understanding between cultures became more pronounced. As students graduated into the global working world, they inherited the need to be able to understand and work with diverse populations, creating an atmosphere of great need for the expansion of multicultural education.

The Americans with Disabilities Act (ADA) was adopted mid-year in 1990 and forbid unequal treatment of those who were disabled across a broad variety of contexts and circumstances. It also sought to establish the definition of what was meant to be disabled in order to be protected under the law. Later Supreme Court rulings more narrowly limited who could be considered disabled under the law. Also in this decade, the Individuals with Disabilities Education Act (IDEA) was passed and reiterated concepts from the legislation in the 1970's ensuring that all children have access to a free and appropriate public education within a least restrictive environment. These two pieces of legislation, the ADA and the IDEA, showed a growing awareness among the legislators and educators of the times "that separating children with disabilities from children without disabilities constitutes unequal education" (Appelbaum, 2002, p. 35).

As changing social and political forces gave impetus to the rise of multicultural education across previous decades in the American schooling experience, the model for field experiences in the preparation of American's teachers remained relatively the same. As previously noted, from the colonial era and through the early periods of the nineteenth century, the original field experience in teacher preparation emerged initially as a type of apprenticeship program. Eventually, in the normal schools between approximately 1830 and 1900, students learning to be teachers practiced under the direction of a master teacher in a model school after taking a collection of courses and field experiences. Into the 1900's, student teaching was later added and the preservice education of teachers was refined and improved over the decades to the point at which today, students are given a far better opportunity to not only practice the art of teaching, but are encouraged to

question, experiment, and reflect on their experience (Slick, 1995). It is interesting to note that even though the process has been refined from the colonial era to present day, the end result has remained relatively the same, basically, a culminating experience and a summative evaluation prior to receiving a teaching license.

It was in 1948 that the Flowers report (Flowers, Patterson, Stratemeyer, & Lindsey, 1948, as cited in Zeichner & Melnick, 1996), initiated the modern day idea of field experience placements in local communities for preservice educators rather than placements in model, or laboratory, schools. The report focused on various kinds of community field experiences that would support preservice educators to learn more about their students and families, foster a greater sense of community service, help them get in touch with community agencies and resources to augment their instruction, and to help dissolve barriers between schools and communities. However, regardless of the importance given to this report at this point in time, community experiences received minimal and sporadic attention in actual practice (Zeichner & Melnick, 1996).

Amongst a blitz of developing organizations, school and national reports and proposals that addressed cultural diversity and gained popularity in the 1970's, 1980's, and into the 1990's, "the idea that teachers needed to receive specific preparation in order to work effectively with a diverse population began to take hold as the education profession began to define goals and curricula for multicultural teacher education" (King, Hollins, & Hayman, 1997, p. 10). During this time, there were four well-known organizations that emphasized links between multicultural education and teacher preparation programs. The National Teacher Corps, founded in 1966 by the federal

government, functioned for sixteen years educating thousands of teachers for the purpose of improving the quality of teachers to serve low-income rural and urban areas.

Community experience was a central component in the education of the interns as was recruiting minority persons into the teaching field, thus establishing promotion of cultural diversity (Cruickshank, 1996; Zeichner & Melnick, 1996).

In 1973, the American Association of Colleges of Teacher Education (AACTE) adopted a policy statement directing that schools of education must be responsible to issues of pluralism and provided a clear objective regarding the need for educational institutions to seriously attend to race, ethnicity, difference, and social justice (Nieto, 2000). A few years later, in 1982, the National Council for Accreditation of Teacher Education (NCATE) issued standards addressing multicultural education requiring its member institutions "to pay more focused attention to diversity in their curriculum, instruction, and field placements" (Nieto, 2000, p. 182). The Renaissance Group, a consortium of 19 universities was established in 1989 for the purpose of improving teacher preparation programs. In their position paper, *Educating the New American Student*, published in 1993, one of the five areas of focus targeted minority and multicultural programs (Cruickshank, 1996).

After a declining period in the late 1970's in which the teacher education enrollments greatly declined, attention in universities shifted from the multicultural movement to sustaining weakening teacher education programs (King, Hollins, & Hayman, 1997). However, in 1983, *A Nation At Risk* was published by the National Commission on Excellence in Education and generated a great amount of discussion.

According to King, Hollins, and Hayman (1997), the emphasis turned from school curricula to a focus on teacher quality and the nature of teacher preparation programs in meeting a growing diverse population of students. International field experiences and placements within urban and Native American centers rose in popularity in the last two decades. Even with all the attention and concern, the pace of change has been slow over the last two decades and few changes of substance have occurred in teacher preparation programs. This has resulted in varying intensity and focus of efforts that address curriculum, program, and field experience transformation for preservice as well as classroom teachers (Nieto, 2000).

In January of 2002, President Bush signed into law the No Child Left Behind Act of 2001 (NCLB), reauthorizing the Elementary and Secondary Education Act (ESEA). The law represented the President's bipartisan education reform plan and made "the most sweeping changes to the Elementary and Secondary Education Act (ESEA) since it was enacted in 1965" (U.S. Department of Education, n.d.). The focus of the law was to improve the performance levels in America's schools while ensuring that all children learned to their highest potential. There are four basic educational principles upon which the law stands: "stronger accountability for results, increased flexibility and local control, expanded options for parents, and an emphasis on teaching methods that have been proven to work" (U.S. Department of Education, n.d.). The NCLB legislation forces emphasis on accountability on the part of states, school districts, administrators, and teachers to the achievement of various student subgroups. Prior to legislation, more often than not, poor performing students were veiled in average scores reported out by districts

and states. However, under the NCLB Act, all students, including those from low socioeconomic backgrounds, racial/ethnic groups, special needs, English Language Learners (ELL), and gender groups, will receive attention to ensure proficient levels of learning are reached. Discussions and research will continue as the nation progresses through the next decade under this law.

After the enactment of NCLB, Education Secretary Rod Paige, in an address celebrating International Education Week in November 2002 (Cardman, 2002), described new federal policies for international education priorities. Programs highlighted in his address included increasing expertise and knowledge about other cultures, regions, languages, and international issues, sharing educational practices and policies between the United States and other nations, providing leadership working with international partners, and promoting United States education. For the nation's public schools, Paige supported the development of partnerships between institutions of higher learning and K-12 programs for the purpose of funding high-quality programs that develop students' international skills and knowledge. Paige stated, "We are ever mindful of the lessons of Sept. 11 – one of which is that all future measures of a rigorous K-12 education must include a solid grounding in other cultures, other languages and other histories" (Cardmon, 2002).

International Education Foundations

Internationally, along with the Fulbright Act of 1946, which continues today to promote international education exchange programs, two major types of American Overseas Schools (AOS) emerged out of the 1940's in the international community and

still operate today: dependent and independent international schools (Simpson, 2000).

Dependent schools operate as part of the Department of Defense Dependents Schools (DoDDS), as an organizational element of the Department of Defense Education Activity (DoDEA), a civilian agency of the United States Department of Defense (Education America, 2000-2003). There are three subdivided areas, Pacific, Europe, and Americas/Cuba, in the DoDDS system and each area is organized into districts. Currently, the DoDDS system worldwide includes 153 schools in eight districts and in 13 different countries with 6,185 teachers and 71,100 students. DoDDS schools mainly serve children of military families and Department of Defense civilian employees: Army, 35%; Navy, 15%; Marine Corps, 6%; Air Force, 29%; and civilian, 15% (Department of Defense Education Activity, n.d.). In regards to race and ethnicity, 48% of DoDDS students are White, 17% are African American, 11% are Hispanic, 8% are Asian, 2% are reported as Other, and 14% are reported as Bi/Multi Racial. The majority of the teachers are White, 88%, with 7% reported as African American, 3% Hispanic, and 2% Other. Unlike stateside schools, 27% of the teachers are male and 73% are female. This is a higher percentage of male teachers than seen in the nation as a whole (Department of Defense Education Activity, n.d.).

The DoDDS system maintains a progressive, quality education for all of its students and supports the concept of setting high expectations for students.

Communications to parents are important for supporting high expectations and specific programs and initiatives are implemented to support the standards and goals of the system. The curriculum is patterned after stateside schools and includes structures for

students planning to attend college, those in special needs areas, and those in talented and gifted programs (Education America, n.d.).

Independent schools include those organized by expatriate communities, multinational companies, private entrepreneurs, and those supported in part by the U.S. Department of State Office of Overseas Schools. In addition, there are international schools that operate under a religious affiliation (Simpson, 2000). The various independent schools, with student enrollments from multiple nationalities and cultures, contain a far more international flavor than the DoDDS schools. Many of these schools, but not all, have received some form of assistance and support from the U.S. government under a program administered by the Office of Overseas Schools under the direction of U.S. State Department of State. These schools represent the American—Sponsored Overseas Schools (ASOS) also referred to as American Overseas Schools (AOS; U.S. Department of State, n.d.).

During the school year 2004-2005, the Office of Overseas Schools is assisted 191 schools with 12,788 teachers and administrators and 103,283 students in 132 countries. American teachers constituted 43% of the staff and American students were 27% of the student populations. The purpose of the assistance and support is twofold: provide adequate education to U.S. Government employees and demonstrate to foreign nationals the philosophy and methods of American education. The schools also strengthen understanding between Americans and people of other countries and cultures (U.S. Department of State, n.d.).

The independent AOS schools are a variety of different structures, from small schools to larger schools and with facilities that range from rented homes to million dollar campuses. The schools are controlled by associations of the parents of students enrolled and the U.S. Government is not part of the operation of the schools, nor does it have any control of the schools. The schools must abide by the host country laws and regulations in regard to importing educational materials, personnel practices, and other educational practices. Tuition is the major form of financial support and the schools reflect a highly academic core curriculum.

Historically, in 1821, Congress enacted a series of laws in support of schools on military posts, supporting their operation for approximately the next 75 years. Following the Spanish-American War in 1889 until post World War II, the military-based schools went through multiple changes of provided support and withdrawn support until the Department of the Army established and ran dependent schools in occupied countries of Germany, Austria, and Japan in 1946. Enrollment grew rapidly and by the late 1960's the three major military departments were operating 200 elementary and 100 junior and senior high schools in foreign countries. In 1964, the Secretary of Defense combined the three school systems into one worldwide system: the Department of Defense Overseas Dependents School System. At the same time, dependents assigned to U.S. embassies around the world also attended private American schools or international schools assisted by the U.S. Department of State. In 1976, the Office of Overseas Dependents Education was established and all international American military and government schools came under the operation of this office, still connected to the Pentagon and the Department of

Defense. Three years later, the name was changed to what is known today as DoDDS – Department of Defense Dependents Schools (Drysdale, 2000).

The DoDDS organization went through several expansions and draw downs to its present size today. After the end of the Cold War, in the 1980's and early 1990's, DoDDS was combined with the stateside schools on U.S. located military bases, initially referred to as Section 6 schools due to their establishment under Section 6 of Public Law 81-874, in 1994 under an umbrella agency DoDEA – Department of Defense Education Activity (Drysdale, 2000; Department of Defense Education Activity, n.d.). The school system remains under the direction of this office today.

In regard to the independent schools, in the early 1940's there were approximately 18 AOS schools, located mostly in Latin America. At that time, Nelson Rockefeller identified the need for creating and sustaining American-sponsored schools in other countries. Small amounts of financial support were provided as grants to a limited number of international schools and the system expanded across the next several decades (Miller, 2000).

By the early 1960's there were a considerable number, approximately 90, worldwide schools that had been "founded by parents, organizations, churches, and corporations" (Miller, 2000, p. 10) that served American students overseas. By 1963, three federal laws were established that were the foundation of support for American schools abroad. There were the Fulbright-Hays Act of 1961, the Foreign Assistance Act of 1961, and the Foreign Service Act of 1963. In 1964, the Office of Overseas Schools was established and charged with "planning, coordinating, and implementing the

overseas programs" (Miller, 2000, p. 11) of the Department of State, Bureau of Educational and Cultural Affairs, and Agency for International Development. There were nearly 100 AOS schools abroad that served approximately 23,000 American students and 18,000 students from other countries. These schools became know as the American-sponsored Overseas Schools. Over the last several decades, the AOS schools were linked to U.S. educational institutions, goals were established, and strong leadership assisted these schools with continued development. Regional associations were also organized and liaisons were maintained with professional development organizations and educational institutions, including accrediting agencies (Miller, 2000).

In 1998, Congress provided additional strength and support for the American education activities abroad by "returning the U.S. Information Agency and Cultural Affairs activities to the supervision of the Department of State" (Miller, 200, p. 14), which will help ensure the "American-sponsored overseas schools continue to expand, grow, and flourish" (Miller, 2000, p. 14).

Summary

What occurred over the past two centuries in the American educational system was reflected in several significant shifts. Education became equated with schooling, authority and control of school systems shifted from parent control to bureaucratic control, industrialization and corporate economics infused education with the ideas of efficiency and production control, immense immigration was reacted to with the idea for a need for Americanization, the population shifted from rural to urban centers, and school expansion increased to secondary and eventually post-secondary levels. These shifts

significantly redefined education as a part of the economic, social, and political forces of the country. Most importantly, they gave rise to the concept of schooling as being the great equalizer for the masses under a meritocratic system. However, as Oakes (1986) points out, equal results did not occur and the system, in actuality, still today, perpetuates inequities in the broader social structure and schools tend to offer advantages to the advantaged and fail to promote the disadvantaged. Even with rising awareness of the diversity issues in the last two decades, a growing number of organizations, materials, resources, and commitment to providing multicultural education, and available international teaching experiences, the most difficult questions relating to educational equity and access still are not being sufficiently addressed. It has been a long-stated goal of education in the United States that all children should benefit from a public education, regardless of family background. However, that has not been the case, nor is it currently the case. "School conditions in our society have been consistently, systematically, and disproportionately unequal and unfair, and the major casualties have been those students who differ significantly in social class, race, ethnicity, native language, and gender from what is considered the 'mainstream'" (Nieto, 2000, p. 181).

It is a time of critical concern in the nation's schools. Education systems need to be transformed to meet the needs of all the students and to embrace equity and multicultural education from a social justice perspective. It is a time when educators need to embrace a multicultural perspective and infuse all instructional practices with awareness and celebration of diversity. The question that remains is how to provide teachers with the tools to do so. This study will address that issue as it investigates

teachers' cross-cultural effectiveness and instructional practices within international settings with diverse student populations.

Diverse Student Populations Compared to the Teacher Population

Literature is replete in the affirmation of the deepening ethnic texture in American society. Even more diverse than its beginning 200 years ago, the nation's multicultural nature will continue to become even more diverse into the foreseeable future (Appelbaum, 2002; Bennett, 2003; Craft, 1996; Dilworth, 1998; Gay, 2000; Gollnick & Chinn, 2002; Grant, 1995; Grant & Sleeter, 2003; McIntyre & Byrd, 1996; Nieto, 2004; Samovar & Porter, 2001). Current immigration patterns, particularly with the influx of people from Southeast Asia, the Caribbean, and Latin America, substantiate that ethnic pluralism in American classrooms will also continue to grow (Bennett, 2003). Along with bringing ethnic and racial diversity to today's classrooms, new waves of immigrants also bring differing religions, customs, celebrations, priorities, languages, and expectations. However, that only begins the diversity challenges confronting classroom teachers. Added to cultural diversity are differences in regard to abilities, socioeconomic status, gender, sexual orientation, and family structure. Consider the following scenario from Grant & Sleeter (2003, p. 11).

- You have a new student who recently moved to the United States from Bosnia. She speaks only a handful of English words, and her life experiences are quite different from yours as well as from that of the other students in the class.
- Three of your students spend a portion of their school day in learning disabilities classes, one attends a class for the emotionally disturbed, and one is visually impaired.
- Most of the students in your class are Mexican Americans (and you are not). Some of their manners of responding to you are unfamiliar to you.

- You are transferred to a new school and most of your students live at or below the poverty level, unlike your own socioeconomic background.
- A group of students in your class seems tuned out, unmotivated, and academically behind.

These statements, or others like them describing the innumerable differences of students in one classroom, are accurate portrayals of the overwhelming diversity challenge confronting teachers across the country.

As seen in the statistics in Table 1, the trend of enrollment in America's schools continues to be one of a growing minority and diverse population. Of the current enrollment of approximately 48 million school-age children in public schools, 62% are non-minority and 38% are minority. It is projected that by the year 2040, student populations will reflect 49% non-minority and 51% minority. As can be seen, the

Table 1

National Public School Comparison of Current and Projected Pre-K – 12 Enrollment by Racial/Ethnic Group

Racial/Ethnic Group	% of Total School Enrollment			
	2003	2025 Projected	2040 Projected	
White	62	52	49	
African American	17	17	14	
Hispanic	16	22	28	
American Indian	1	1	1	
Asian/Pacific Islander	4	8	8	

minority will actually become the majority with the largest growth within the Hispanic group of children (Cetron & Cetron, 2003/2004; Education Week, 2000; Futrell, Gomez, & Bedden, 2003; Lewitt & Baker, 1994; National Center for Education Statistics, n.d.; National Center for Education Statistics, 2005; U.S. Census Bureau, n.d.).

In addition, 38% of school age children across the nation live in poverty and 33% have limited English language proficiency with 8% receiving direct services as English Language Learners. One-third of all African American and Hispanic students attend schools that have a minority enrollment of 90% or above. In one hundred of the largest school districts in the United States minority students are the majority population and 48 of those 100 schools have a poverty rate of 20% or higher (Futrell, Gomez, & Bedden, 2003; Iowa Department of Education, 2004; National Center for Education Statistics, n.d.).

In Iowa, which is generally considered to be a rural, monocultural state, enrollment statistics show similar trends indicating a shift in student demographics (Table 2). According to the Iowa Department of Education (2004), in school year 1985-1986, the Iowa public school student enrollment of 485,332 students reflected more than 95% White (non-Hispanic) students and less than 5% minority. In 2003-2004, while the public school student enrollment of 485,011 was similar to the total enrollment nearly 20 years earlier, there was a decrease in the White (non-Hispanic) student population and an increase in all other racial/ethnicity categories. In the fall of 1986, Iowa was one of 13 states in the nation with over 90% total enrollment in the White racial/ethnic group. By the fall of 2000, Iowa was one of only five states with a White student population of over

90% and by 2004 that had decreased to 88% (Iowa Department of Education 2004).

Thus, even in Iowa, a rural state, classrooms are becoming more diverse in regard to race and ethnicity grouping.

Table 2

Iowa Public School Comparison of Pre-K – 12 Enrollment by Racial/Ethnic Group

Racial/Ethnic Group	1985-1986 Enrollment %	2002-2003 Enrollment %	Change Enrollment %
White	95.3	89.0	- 7.2
African American	2.5	4.3	+ 67.6
Hispanic	0.8	4.4	+ 425.3
American Indian	0.2	0.5	+ 141.7
Asian/Pacific Islander	1.1	1.8	+ 61.0

The number of immigrants to Iowa continues to grow, paralleling the same trend at the national level. The number of English Language Learners (ELL) in the Iowa student population reached a 15-year high in 2002 with 15,452 students, which is a 391% increase since 1986 when there were 3,150 ELL students in Iowa (Iowa Department of Education, 2004). In comparison to the nation, the 2004 Iowa ELL public and nonpublic school enrollment was 3% of the total student population compared to 8% nationwide.

The top three identified home countries were Bosnia-Herzegovina, Mexico, and Vietnam. Immigrants from Bosnia-Herzegovina made up a little more than 29% of all immigrants to Iowa in 2001 and only three other states in the nation received more immigrants from Bosnia-Herzegovina than Iowa in that year (Iowa Department of

Education, 2002). Immigrants from Croatia and Sudan were two new student groups reported in Iowa in 2002 (Iowa Department of Education, 2004). During the 2003-2004 school year, over 75 languages were represented as the primary language for ELL students in the state The top ten languages, in rank order, were Spanish, Bosnian, Vietnamese, Laotian, Serbo-Croatian, German, Arabic, Chinese, Korean, and Russian (Iowa Department of Education, 2004).

In regards to socio-economic status, the national percent of students living in poverty was reported at 38% while Iowa reported 33%. The students eligible for free and reduced priced meals in Iowa was 30% of the student population, representing not only an increased percentage, but also indicating a 14-year high of students of low socio-economic status. When reviewing individual districts in Iowa, percentages ranged from a low of 5% to a high of 64% of students in districts eligible for free and reduced priced meals (Iowa Department of Education, 2004). This establishes socio-economic level as a significant factor in the diversity picture in classrooms in Iowa.

In the area of special education programs, students served under the Individuals with Disabilities Education Act comprise 13% of the enrollment of school age children nationwide. In Iowa, the number of students is 15%, higher than the national percent and one of the higher states in the nation in regards to the percent of students served under this act. Again, special needs students, including both special education and talented and gifted students, are a contributing factor in creating diverse classrooms in Iowa (National Center for Education Statistics, 2003).

Contrast the picture of the student diversity reflected in public school enrollment trends, both nationally and in Iowa, with the racial, ethnic, and background cultures of current and prospective classroom teachers. In 2001, of those persons with bachelor degrees or above in the field of education nationally, 90% were White (non-Hispanic), 5% were African American (non-Hispanic), 5% were Hispanic, and 1% were Asian and Pacific Islander, and 1% were American Indian (National Education Association, 2003). In large urban school districts, African American, Asian, and Hispanic teachers are more prevalent, sometimes six times as much, than in medium and small sized districts and the African American representation in the teaching field has decreased since 1971 from 8% to 6% in 2001, while White teachers have increased in number (National Education Association, 2003).

In the decades of the 1960's through the early 1980's male representation in the teaching field was generally 33%. Since 1981, the percentage of male teachers has steadily been dropping. Currently, at a 40-year low, males represent only 21% of the teaching force while females represent 79% (National Education Association, 2003).

In Iowa, similar trends are seen as well as differences. There has been a decrease of male teachers from 37% in 1986 to 28% in 2004, which also indicates an increase of female teachers. Minority representation in the teaching field in Iowa remains relatively low, but has actually increased slightly from 1.2% in 1986 to 1.8% in 2004. More minority teachers are clustered in larger, urban centered school systems as opposed to medium and small sized systems (Iowa Education Department, 2004).

On a last review point of population statistics, it can be noted that at the college and university level a majority of full-time instructional faculty and staff is White (non-Hispanic) at 85%. Asian and Pacific Islanders are represented at 6%, African Americans (non-Hispanic) are represented at 5%, Hispanics are represented at 3%, and American Indians are represented at a little less than 1% (National Center for Education Statistics, n.d.). Thus, practicing and prospective teachers are products of predominantly White colleges of teacher education and come from a monocultural, White neighborhood (Howard, 1999).

In summary, statistics paint a picture in which male teachers and teachers from non-majority groups are declining in numbers, while the population trends show significant and continuing increase in students from diverse groups. As we progress into the nation's education future, the vast majority of teachers will be White, monolingual, and from the middle socio-economic class while the student population will grow increasingly diverse (Howard, 1999; Melnick and Zeichner, 1998). This suggests a "demographic imbalance between non-minority teachers and minority students" and indicates, "teachers will be working with students whose cultural backgrounds are different than their own" (Larke, 1990). The next section will explore the implications of this situation.

The Link Between Teacher Characteristics and Student Achievement

Culture "refers to the dynamic system of social values, cognitive codes,

behavioral standards, worldviews and beliefs that give meaning and order" (Gay, 2000,

p.8) for students, teachers, parents, and various groups in society. Culture is the

determinant in the thinking of teachers, in what they believe, and how they behave, which in turn affects how they teach and learn. Teachers' and students' cultures are foundational to the educational process due to the sociocultural nature of education (Gay, 2000). Teachers carry their personal cultural background into the classroom and perceive students and student abilities from their own background lens that contains preconceptions and possible prejudices. Teaching is based on the teachers' social identities, and complex, intertwined beliefs, which contain issues of race, ethnicity, and social class (McCall, 1995; Warren, 2002) and teachers filter curriculum through personal cultural experiences and generally teach in the same way that they were taught (Latham, 1999; Mahan & Rains, 1990).

Students also bring their cultural contexts into the classroom and come to school with their own perceptions and cultural backgrounds that provide the foundational component for the construction of their knowledge and how they draw meaning from the behaviors of teachers, administrators, and peers. The mismatch between the background cultures of teachers and students impacts learning (Latham, 1999).

The climate of schools has a foundational nature born out of the early colonial period that reflects a European, middle-class, White perspective and contains specifically prescribed ways and beliefs that are value-laden (Gay 2000). "Teacher training and textbooks have tended to attribute educational failures to deficiencies in children" (Persell, 1997, p. 97) and often times teachers believe that these deficiencies are a result of social characteristics such as ethnicity, language, social-economic status, or behavior,

rather than attributing deficiencies to the social structure of schools and the classroom (Persell, 1997).

Teachers who exhibit low expectations and lack of efficacy often times lower their teaching standards, produce less teaching effort, and use less rigorous curricula for low-achieving students in poor urban schools. At times, the belief is exhibited that some students will not learn or cannot learn and that there is little, if anything, that a teacher can do to change the situation. This affects student achievement and academic performance (Warren, 2002). Some teachers who lowered expectations expressed the belief that they are not responsible for helping students overcome what is perceived as deficits, they viewed students' families and cultures as deficits rather than strengths, they lacked confidence to help students overcome barriers to learning, and they lacked determination to carry out the work needed to teach these students (Warren, 2002).

Teacher perceptions and beliefs about non-majority students determine the level of expectations set in the classroom and the kinds of treatment students receive (Gilbert & Gay as cited in Larke, Wiseman, & Bradley, 1990; Gollnick & Chinn, 2002; Sleeter & Grant as cited in Larke, Wiseman, & Bradley, 1990). When teachers use subjective evaluations, it is more likely that those students with a similarity to the teacher's race, gender, and ethnicity will be more likely to receive a more positive subjective evaluation (Latham, 1999). McCall (1995) cites a study by Eggleston indicating that teachers often have lower expectations for girls, African Americans, and lower socio-economic students. White, middle-class males receive the most attention from teachers and African American boys receive the least (American Association of University Women as cited in

McCall, 1995; Banks as cited in McCall, 1995). Larke, Wiseman, and Bradley (1994) cite numerous studies indicating that African American, Hispanic, and Native American are perceived to be less competent in academic activities, while White and Asian students are perceived to be more successful and more competent. In addition, male students receive more positive and negative praise than females and that teachers perceive that higher socio-economic status is linked to higher achievement and lower socio-economic status is linked to lower expected achievement performance. Persell (1997) suggests that social class "may influence teacher expectations directly, or indirectly, through test scores, appearance, language style, speed of task performance, and behavior" (p. 97).

Warren (2002) noted that ineffective teachers do not value the special talents of the culturally diverse students and their families. For example, some teachers view being bilingual a deficit rather than a strength in that the bilingual student has access to two cultures. Larke (1990) indicated that the value teachers place on the student's language is an important component in the expectation level the teacher sets for the student. Because language is a very important part of children's lives and culture, if teachers devalue the language, then they are devaluing both the student and the student and family's culture.

Teachers' expectations affect the frequency of interaction they have with students as well as the kinds of behaviors exhibited toward different children. "Teachers spend more time interacting with pupils for whom they have higher expectations" and "students for whom teachers held high expectations were praised more frequently when correct and were criticized less frequently when wrong or unresponsive than were pupils for whom teachers had low expectations" (Persell, 1997, p. 98). Persell (1997) also cites

Rosenthal's beliefs that teachers show more overall warmth to those students they expect to do well, that educators teach more to those students for whom they hold high expectations, and that high-expectancy students are called on more often, given more chances to reply, and asked more difficult questions. It is clear to see that teachers hold different expectations for different students and that they behave differently in accord with those expectations. The implications are that when "teachers hold definite expectations and when those expectations are reflected in their behavior toward children, these expectations are related to student cognitive changes" (Persell, 1997, p. 98).

Larke (1990) summarizes multiple studies with findings that indicated teachers who are culturally sensitive are more effective in ensuring that all students achieve to high levels of expectations. Their cultural sensitivity can be seen by how they modify curriculum and instruction and break down barriers in the learning process and develop understanding of their own personal cultural contexts and those of their students.

Successful teachers demonstrate the belief that all children can succeed, respect the students and the communities they come from, help students make connections between their learning, their community, and the world, have high expectations of all students, rethink and inform their instructional approach by listening and learning from students, and have a sense of efficacy in that they believe they can have a dramatic impact on student achievement (Ladson-Billings, 1994; Pang & Sablan, 1998). In addition, successful teachers perceive students not only as members of a group, but also as individuals (Locke as cited in Marshall, 1996), utilize both content and instructional strategies that fit students instead of the other way around, and view themselves as part of

the students' community instead of being separate from it (Ladson-Billings, 1994).

Lastly, successful teachers develop trusting and caring relationships with students,
believe in teaching students how to cope with social inequities and to work toward the
betterment of their community, and are totally committed to facilitating academic and
social growth in their students (Ladson-Billings as cited in Pang & Sablan, 1998).

Recent studies in Tennessee and Texas (Carey, 2004) confirm that teacher effectiveness is the most important influential factor in student achievement gains; teachers have greater influence than poverty, race, parent's education, and additional factors often attributed to student failure in schools. Effective teachers know their subject matter, are more experienced, are fully certified, are more likely to score higher on various types of standardized tests, and were engaged in a high quality and competitive undergraduate education program. "While none of these qualities is a perfect proxy for teacher effectiveness, research shows that those teachers who demonstrate some or all of these qualities in combination are, on average, more effective than those who do not" (Carey, 2004, p. 36). The disturbing information is that "poor students, low-performing students, and students of color are far more likely than other students to have teachers who are inexperienced, uncertified, poorly educated, and under-performing" (Carey, 2004, p. 8). The lack of distribution of effective teachers to low-income and minority students has had and will continue to have crushing impact on student achievement. The poor students, students of color, and those who are low-performing are more likely to have teachers who are under-performing, poorly educated, uncertified, and inexperienced. Carey (2004) continues with:

...the more we know about effective teachers and how important they are, the more we also know where they are — mainly in the schools and classrooms of America's better-off white students. Where they are not is teaching low-income and minority students, the students who have traditionally been short-changed by the education system, the students who are most dependent on our public schools (p. 8).

Harmon (2002) completed a study regarding the experiences of gifted African American students who were bused in a desegregation initiative. The study was completed from the perspective and through the voices of the talented and gifted African American students involved in the initiative. Within the results of the study, students identified characteristics of effective and ineffective teachers. Students reported that ineffective teachers "lacked an understanding and appreciation of African American culture and behavior" (p. 71), set low academic expectations, did not believe students were capable of learning or of being gifted students, gave students less challenging work, "did not explain concepts in a way students could understand, did not relate content or concepts to students in meaningful way, relied on auditory teaching methods, and gave instructions using language that was difficult to understand" (p. 72). In addition, ineffective teachers were believed to be prejudiced, disrespectful, and uncaring.

Conversely, Harmon (2002) also highlights that students believed effective teachers to be respectful, caring, introduced new concepts in ways that could be understood, used relevant and interesting content, employed differentiated instruction, used multicultural materials, and employed various teaching styles. When interviewing the effective teachers identified by the students, the author found the teachers who were motivated to teach minority populations, "created classroom environments that did not tolerate any form of prejudice" (p. 73), taught positive social skills, provided a risk-free,

safe classroom environment, did not tolerate prejudice, and set expectations high for academics and behavior.

In summary, teachers are the most influential factor in student learning and effective teachers are content knowledgeable, educated in competitive programs, perform well on tests, are certified, set high expectations for students, provide a risk-free learning environment, form positive, caring relationships with students, provide relevant lessons, use various teaching strategies to meet learning styles of students, are motivated, and exhibit efficacy. From a cultural perspective, teachers see their world from their own demographic lens and may unconsciously form different values and expectations about students who are different in their backgrounds from their own. This impacts the expectations teacher may hold for students, how they respond to students in the classroom, and how they view the student's family. The lack of cultural sensitivity on the part of teachers builds barriers to positive student achievement. The next section describes one approach to building preservice and classroom teachers' cultural awareness and sensitivity that supports high expectations for student achievement and contributes to teacher effectiveness: cross-cultural and international teaching experiences.

Effects of Cross-Cultural Field Experiences on Preservice and Classroom Teachers

For the field of education, and applicable to other fields as well, Eleanor (1997) suggests that literature studies reveal five cross-cultural training models that have been used at various times and in various situations to address the need for individuals and groups to become more culturally sensitive; the social exchange model, the intellectual model, the awareness model, and the environmental training model.

The fifth model, the environmental training model, is the model of focus for this study and is significantly related to the findings of studies on cross-cultural immersion field experiences at the preservice teacher education level as well as with classroom teachers. In this model, "the trainees are active participants in the learning process from within the foreign culture or a simulated foreign culture" (p. 82). When applied to the field of education, the environmental, or experiential learning, model can easily be seen through the various field experiences of preservice teachers and in the various cross-cultural experiences classroom teachers may have outside their employment or within their employment. One of the three ways this model is most effective is when it is "used by a university that can promote a stable, ongoing relationship with a cultural group" (Eleanor, 1997, p. 82) and is within a supervised cultural immersion experience.

Various programs and teacher educators have implemented cross-cultural field experiences that challenge preservice teachers to reflect on their own perspectives regarding students and families who are unlike themselves. The more well known programs include Cultural Immersion Projects through Indiana University (Stachowski & Mahan, 1995, Stachowski & Mahan, 1998), the Minority Mentorship Project (MMP) at Texas A & M University (Larke, Wiseman, and Bradley, 1990), the Urban Education Program of the Associated Colleges of the Midwest (Melnick & Zeichner, 1998), and the Consortium of Urban Professional Development and Technology (Ligons, Rossado, & Houston, 1998).

Stachowski and Mahan (1998) have been implementing the Cultural Immersion Projects since the early 1970's, which includes both domestic and international field experience placements. The projects place student teachers in schools across the Navajo and Hopi Indian nations to work in Native American classrooms, in schools on the border of Arizona and Mexico and in the Rio Grande Valley of Texas to work with Latino children, and in schools near Indianapolis where low-income African American children attend. Important new learnings resulting from the cross-cultural experiences as reported by the participants were compiled into the categories of teaching strategies, curriculum, knowledge acquisition, discoveries about self, and world or global issues. The most important sources of learning for the participants came from community sources, both community members and students. When preservice teachers are involved in this type of experience, they better understand how different cultures live, what they think, and what they value (Gomez, 1996; Stachowski & Mahan, 1998). These are all critically important if educators are to make connections with children from diverse cultures that will yield improved student achievement.

Larke, Wiseman, and Bradley (1990) completed a study with the Minority

Mentorship Project (MMP) of Texas A & M University. This is a program that integrates
multicultural education, human relations training, and cross-cultural mentoring for the
purpose of "changing attitudes of preservice teachers about minority students" (p. 10). In
this program, pre-service education majors mentored minority students through early
field experience placements. Results of the study revealed that "the perceptions and
attitudes of the mentors toward their mentees had changed" (p.7). Preservice teachers
had moved from a more ethnocentric point of view to one with some degree of cultural
understanding. From the interactive experiences with their mentees, the university

preservice educators' attitudes regarding minority students were changed to a more positive, warm, and knowledgeable point of view. The preservice teachers saw strengths and capabilities in their mentees by the end of the experience as compared to the initial feelings revealing negative attitudes and perceptions towards the minority mentees. Thus, cross-cultural mentoring "can be used positively to enhance the attitudes of preservice teachers to work with diverse students" (p. 10).

The Urban Education Program of the Associated Colleges of the Midwest (UEP) has been in existence since 1963 and community field experiences to preservice educators from small colleges that cannot offer this opportunity themselves. The program places preservice teachers, who mostly come from monocultural backgrounds, in urban settings in Chicago's ethnic communities with the purpose of developing various skills and attitudes related to the field of teaching. One focus is the "understanding of the cultural dimensions of people's lives and their impact on learning" (Melnick & Zeichner, 1998, p. 91). The immersion in the Chicago setting is radically different from the setting they are used to on their home campuses. Results indicate that participation profoundly affects the thinking of the participants and are transformative in nature in regard to building cultural sensitivity (Melnick & Zeichner, 1998).

In one Texas urban center, the Consortium of Urban Professional Development and Technology places preservice educators in urban professional development schools for various field experiences. From the studies conducted within this program, several major findings emerged. Prospective teachers, classroom teachers, and university faculty involved in these programs taught differently from those in more traditional programs,

the professional knowledge of the preservice educators tended to be greater than in the traditional placement programs, and student achievement in the public schools where the placements occurred tended to increase when these schools were transformed into professional development schools for increased placement of preservice teachers (Ligons, Rossado, & Houston, 1998).

Additional studies and cross-cultural experiences located within the United States yielded supportive evidence of the importance of cross-cultural field experiences.

Cooper, Beare, & Thorman (1990), in a comparison between student teachers placed in monocultural schools in Minnesota and those placed in multicultural schools in Texas, found that those in the multicultural experience demonstrated more multicultural competencies. This included being more comfortable discussing racial issues, believing they had the necessary training to teach in culturally diverse settings, maintained racial and ethnic openness, had high expectations for all students, were more likely to encourage a variety of viewpoints, helped their students develop skills needed to survive in a society dominated by a White culture while uplifting their students cultural values, were more likely to visit the families of their students, and provided opportunities to celebrate differences. Huber and Kline (1993) reported from a longitudinal study that field experiences contribute to an increase in preference for social diversity and that they create significant differences in preservice teacher attitudes toward diversity.

In regards to international cross-cultural experiences, in 1985 West (as cited in Jarchow, McKay, Powell, & Quinn, 1996) indicated there were 100 educational institutions that conducted student teaching programs in 55 different countries

worldwide, providing a vast number of cross-cultural experiences for student teachers. Wilson (1993) reported that cross-cultural international experiences contributed significantly to participants' global perceptions and overall knowledge of cultural lifestyles. Individuals experienced personal growth and built meaningful interpersonal relationships, which contributed to the development of effective teacher qualities. In the area of perceptual understanding, impact of a cross-cultural international experience contributed to "open-mindedness, anticipation of complexity, resistance to stereotyping, inclination to empathize, and non-chauvinism" (Wilson, 1993, p. 22). Development of knowledge included "knowledge of other cultures, and a general awareness of world issues, global dynamics, and human choices" (Wilson, 1993 p. 22). Traits in the area of personal growth that were impacted by the international experience included increased self-confidence, independence, and acceptance of responsibility. Lastly, interpersonal connections and cultural perceptions were greatly enhanced. (Wilson, 1993).

Kissock (1997) identified two similar programs that place preservice educators in overseas locations for student teaching field experiences; Moorhead State University Student Teaching Abroad (STA) program, in existence since 1969, and the Global Student Teaching (GST) program, located at the University of Minnesota, Morris (UMM), in existence for the last sixteen years. Together, the programs work cooperatively placing more than 120 students each year in almost 70 different schools on five continents to complete student teaching cross-cultural field experiences. From these placements, students learned about curriculum, instructional practices, classroom management, and assessments that are inherently different than what they experienced in

their own educational upbringing and university classrooms. Findings indicated these students developed not only their teaching abilities, which would be expected from a student teaching experience, but they also learned about alternative approaches to education, questioned traditional assumptions, and broadened their global understanding and world view (Kissock, 1997).

The Overseas Student Teaching Project (Mahan & Stachowski, 1990; Stachowski & Mahan, 1995; Stachowski & Mahan, 1998) places student teachers in schools in England, Wales, Scotland, the Republic of Ireland, New Zealand, and Australia. The participants live with families in the community and fully engage in the various aspects of teaching in their schools while becoming totally involved in the communities in which they live and teach. Study results indicated that the student teachers in the cross-cultural setting reported a greater number of learnings than those in conventional student teaching placements and that these learnings were heavily weighted in the areas of more extensive and diverse factual and affective learnings and encompassed more community and world perspectives. The sources of the learnings in the cross-cultural experiences were attributed more to community experiences and community people than the learnings of student teachers in conventional placements and those placed in cross-cultural experiences became significantly more involved in their host communities leading to a "better understanding of how citizens in the host community live, what they think, and what they value - vital learnings for educators serving the community's children" (Stachowski & Mahan, 1995, p.104). Basically, the overseas experiences provided a greater number of important learnings overall, those learnings were of a broader

significant scope, the learnings encompassed more community and world perspectives and influences, and learnings delved into facts, issues, and relationships that students placed in conventional experiences rarely considered. In addition, cross-cultural learning was far more successful when student teachers were required to become immersed in the local community, which contributed to more culturally sensitive teachers in our nation's schools.

In a Canadian study, Finney and Orr (1995) found that White, Canadian students who were involved in a cross-cultural experience in Australia with the Aboriginal culture became more conscious of bias and stereotyping in their own way of thinking and became more conscious of how they interact with others. The students became more aware of what it felt like to be a minority, gained appreciation for the strength of character exhibited by minority groups, and learned the meaning of oppression as it was expressed in the lives of others through a personal context.

Additional studies report that for those involved in international student teaching placements the international experiences expanded world views, increased professional competence, deepened reflection on the educational practices in other countries, developed a stronger and in-depth global perspective, promoted deeper understanding of students' and families' cultural values, and reconstructed personal beliefs, perspectives, and attitudes about teaching in diverse classrooms (Quinn, Jarchow, & Powell, 1995-1996; Quinn, Jarchow, Powell, Barr, & McKay, 1995). Implications for the field of education are numerous. Living and interacting with children and families different than their own heightens awareness of other cultures as well as the limitations of their own

culture, developing multicultural awareness. International student teaching helps participants confront and diminish their own ethnocentricity and leads to respecting and honoring diversity in others. Deep and enriching international experiences vicariously enrich the lives of classroom students because of the teacher's firsthand knowledge of differing cultures (Jarchow, McKay, Powell, and Quinn, 1996). These factors are critical in effectively understanding the diverse students who enter today's classrooms and to personalize education to meet student needs and ensure academic achievement.

As prolific as the research is on the effects of international field experiences on preservice educators, the literature is limited in addressing the preparation and professional development of classroom teachers for teaching in a culturally diverse classroom. "Most scholars who have examined the literature on inservice teacher education for insights about how to prepare teachers better for cultural diversity have concluded that little attention is given to issues of diversity" (Zeichner & Hoeft, 1996, p. 538). In addition, this researcher was unable to find evidence of studies that followed the preservice educators who had international field experiences into their first years of teaching for the purpose of discovering the lasting effects of their experiences. Most of the literature studies on staff development for classroom teachers examined the structural variations of professional development delivery systems and their effectiveness. However, a few specific studies have given limited insight into effective professional development for classroom teachers in regard to diversity (Zeichner & Hoeft, 1996).

Mahan & Rains (1990) described a study regarding diversity training for classroom teachers that was a summer cultural immersion experience located on Navajo

reservations in southwest United States. From this cross-cultural experience, teachers reported they experienced changes in themselves personally as well as in their professional practices. They reported that they listened better and much more before speaking, had become more nonaggressive, and they incorporated more Native American information and perspectives into existing coursework as well as created new courses for the same purpose. "The authors believe that a cultural immersion practicum...can be both personally rewarding and professionally beneficial in enhancing the growth and development of inservice teachers for a culturally pluralistic country" (Mahan & Rains, 1990, p. 22).

Craig, Hull, Haggart, & Perez-Selles (2000) reported that the use of Teacher Assistance Teams, a process that has been used in identifying accommodations or other changes in support of the needs of special education students, can be effectively applied to increase teacher cultural awareness and competence in classroom instruction and practices. The process is based on creating a problem-solving atmosphere through inquiry and teacher teams can use this process to integrate cultural discussions as part of the problem solving process for students. In this type of learning community, "teachers and administrators learn from on another and expand knowledge and teaching strategies to increase their ability to effectively teach children" (p. 9).

In a study of regarding attitudes of middle school teachers toward English

Language Learners (ELL), Youngs and Youngs, Jr. (2001), identified specific training,

personal contact with diverse cultures, prior contact with ELL students, demographic

characteristics, and personality as predictors of teacher attitudes. Results, specific to this

current study, indicated that teachers are more likely to have positive attitudes toward ELL students if they have lived or taught outside the United States and have interacted with a culturally diverse population of ELL students.

In a study in which teacher educators who were recognized for being effective in preparing teachers in multicultural and global education were asked to reflect upon their practices, Merryfield (2000) found that among the variables that contributed to their effective teaching, several teachers identified their experience living in foreign countries as important. It was reported that long-term cultural immersion in a cross-cultural overseas experience opened the door for rich reflection and rethinking of personal identity as well as new perspectives actions that America takes overseas and its role in the international world forum. It also was "a critical step towards their understanding what being 'different' from the majority of people" (p. 437) was like on a daily, longterm basis. Thus, they began to understand what it meant to be an outsider and recognized the cultural capital and privilege of power of those in the majority culture. This learning was transferred to their work with preservice educators in helping to further the understandings about multicultural and global education and the dynamics of culture and power. As an application to public school systems and universities alike, Merryfield (2000) stated, "If experiential knowledge of diversity and equity is a quality needed in teacher educators, recruitment and hiring of such people is probably a much more efficient and productive strategy than trying to effect changes in current faculty" (p. 441).

In addition to physical cultural immersion experiences for classroom teachers, there are also in existence a variety of programs available through different institutions, organizations, and universities that focus on providing classroom teachers with particular coursework or professional development aimed at examining various component of diversity and the application of learning to personal lives and classroom structure, curriculum, and relationships. However, there is basically limited empirical evidence to support the success of different approaches of multicultural professional development for classroom teachers. Zeichner and Hoeft (1996) concluded that "the kinds of personal transformations that are discussed as necessary in the multicultural teacher education literature, such as the elimination of racist attitudes and practices and the changing of culturally encapsulated people into multicultural people, can only be accomplished by an approach to staff development that goes beyond the skill training and informational or curriculum package delivery workshops that have dominated practice to date" (p. 539). These types of professional development experiences do not support the truly personal nature of changes that are needed to develop teachers of great intercultural sensitivity that will promote academic achievement for all students. Zeichner and Hoeft (1996) further indicated that the "priority in research on teacher education for cultural diversity should be to investigate how particular kinds of experiences for teachers at the preservice or inservice levels are connected to the character and quality of their teaching" (p. 541). This study approaches that call for research.

In summary, as previous sections in this study have pointed out, teacher expectations are shaped not only by their own cultural upbringing and experiences, but also by student ethnicity, family income, and previous identification of students' perceived abilities, which in turn influences how they teach which has a direct impact on

how students learn. Cross-cultural field experiences, within the United States or in international locations, have positive impact in contributing to the understandings and expectations of students whose backgrounds differ from preservice and classroom teachers. This in turn, promotes enhanced classroom instruction and family connections to meet individual student needs and advance high academic achievement. Gomez (1996) indicates that "among the most promising practices for challenging and changing the preservice teachers' perspectives was their placement in situations where they became the 'Other' and were simultaneously engaged in seminars or other ongoing conversations guiding their self-inquiry and reflection" (p. 124). In regard to classroom teachers and their perspectives toward diversity, there is limited research available on classroom teachers and professional development and multicultural practices.

Effective Educational Practices within a Multicultural Education Framework

Seminal research and theory foundational to the multicultural education

framework specific to this study can be attributed to James A. Banks (2002), Carl Grant and Christine Sleeter (2003), Christine Bennett (2003) and Sonia Nieto (2004).

Banks has defined multicultural education as a reform movement designed to promote major change rather than as an added-on initiative in schools and school systems. Key goals of multicultural education are identified by Banks (2002) as (a) "to help individuals gain greater self-understanding by viewing themselves from the perspectives of other cultures" (p.1), (b) "to provide students with cultural and ethnic alternatives" (p.2), (c) "to provide students with the skills, attitudes, and knowledge needed to function within their ethnic culture, within the mainstream culture, and within

and across other ethnic cultures" (p.2), (d) "to reduce the pain and discrimination that members of some ethnic and racial groups experience because of their unique, physical, and cultural characteristics" (p.2), and (e) "to help students to master essential reading, writing, and math skills" (p.4).

The purpose of these goals is to support educational reformation in a pluralistic society that continual works to help students understand their own cultures and communities, helps students understand multiple cultures, and affirms those cultures. It also ensures that students learn the necessary skills, attitudes, and knowledge needed to function across multiple cultures, while maintaining and affirming their own culture. This frees students from cultural boundaries enabling them to function across society for their own good, for the common good in support of a democratic citizenry, and for the purpose of creating a more equitable and just society.

Banks (2002) identifies five dimensions of multicultural education. The content integration dimension includes the use of examples, information, and data that teachers use from a variety of cultures and groups to support content key ideas, principles, theories, and generalizations in the teaching of the content. Banks indicates this as one of five of the major components and yet some schools and school districts accept the practice of using only this dimension, which significantly narrows the scope of multicultural education in those schools.

The *knowledge construction process* dimension consists of how teachers help students understand the "cultural assumptions, frames of references, perspectives, and biases" (Banks, 2002, p. 15) that are evidenced in the content area and influence the

construction of knowledge within that discipline. Teachers help students understand the influence of race, gender, social status, and ethnicity in how knowledge is constructed and represented in the materials presented and the decisions made within the content area (Banks, 2002).

The *equity pedagogy* dimension deals with the techniques and teaching strategies that support academic achievement for all students. This includes using techniques that support specific cultural and learning styles for diverse groups and techniques that increase learning opportunities. A critical aspect of this dimension is the assumption that teachers have a full understanding of the appropriate learning strategies, as well as the cultural contexts of students and the match between learning strategies and cultures (Banks, 2002).

The *prejudice reduction* dimension involves the "characteristics of children's racial attitudes and strategies that can be used to help students develop more positive racial and ethnic attitudes" (Banks, 2002, p. 14). Children learn at a very young age the biases and preferences associated with race. However, racial attitudes can be positively impacted with appropriate teaching materials and strategies that are realistic in nature in regard to racial and ethnic images (Banks, 2002).

The last dimension of Banks' (2002) five dimensions is an empowering school culture and social structure, which is the culture and organization restructuring process that delivers a school environment in which all students experience educational equity, equality of opportunity, and empowerment. Such a process ensures multiple

opportunities for learning success, fair assessments, and normalizes the belief of teachers and staff that all students can learn.

In regard to curriculum integration in schools, Banks (2002) summarizes the information to four basic approaches used to integrate multiple perspectives and cultural aspects into the school curriculum. Level one is the contributions approach, which basically entails the limited content inclusion of holidays, celebrations, foods, specific time periods for study, heroes from that culture, etc. The additive approach involves the addition of cultural concepts, content, and themes to the current curriculum structures. documents, and procedures, without changing any of these in regard to current purposes and characteristics. The transformation approach far more comprehensively changes the curriculum than the two previous approaches at levels one and two. At this level, "the canon, paradigms, and basic assumptions of the curriculum (Banks, 2002, p. 31) are changed and students are enabled to "view concepts, issues, themes, and problems from different perspectives and points of view" (Banks, 2002, p. 31). It is through this approach that students are exposed to multiple cultural perspectives and voices of "both the victors and the vanquished" (Banks, 2002, p. 31). Students are encouraged to use critical thinking in analyzing what they read and hear, and are also guided to develop their own conclusions and generalizations through critically thinking skills. Level four is the decision-making and social action approach, which is analogous to social justice. This level extends the transformative approach and takes the students beyond the classroom walls. Students are enabled to pursue projects, activities, and tasks "that allow them to take personal, social, and civic actions related to the concepts, problems, and

issues they have studied" (Banks, 2002, p. 32). This teaches students to learn about multiple perspectives, care about their community, and take civic-minded action that promotes equity and justice.

Grant (1994) and Grant and Sleeter (2003) outline five approaches for multicultural teaching and meeting the needs of diverse students based on race, ethnicity, gender, social class, disability, and sexual orientation. The approaches are based on the various perspectives of student differences and how teachers handle those differences in the classroom. The first approach is *teaching exceptional culturally different students*. This approach focuses on how to bridge the capabilities of the student with the demands in classrooms, the school, and society as a whole. The aim is to help the student function successfully in multiple contexts (Grant, 1994; Grant & Sleeter, 2003).

The second approach is the human relations approach and is based on students getting along, appreciating, and respecting each other and themselves. This approach helps students build positive feelings (Grant, 1994; Grant & Sleeter, 2003).

The third approach, the *single-group studies* approach, supports the teaching of groups to students through various content and contexts. Groups studies includes women, Arab Americans, African Americans, those with disabilities, those alternate sexual orientations, etc (Grant, 1994; Grant & Sleeter, 2003).

The fourth approach is the multicultural education approach and is actually somewhat of a combination of the first three approaches. The specific aim is to affect systemic change in schools and school systems in which diversity, equity, and equality are embedded in the school climate as well as in the individual lessons and units teachers

organize for learning (Grant, 1994; Grant & Sleeter, 2003). It is this approach that most closely matches with the work of Banks (2002), Bennett (2003), and Zeichner (1996).

The fifth approach is comparable to the social justice perspective (Nieto, 2004) and the *decision-making and social action approach* of Banks (2002). This approach addresses social inequalities within the society and between groups of people and charges students and adults alike with the responsibility of addressing those inequities, to work productively for the betterment of groups, and to constructively and responsibly respond to social problems (Grant, 1994; Grant & Sleeter, 2003).

Nieto (2004) has developed a model of multicultural education that is based on "a variety of levels of support for pluralism" (p. 384). The four levels of multicultural education are classified as "tolerance, acceptance, respect, and affirmation, solidarity, and critique" (p.384). The first level is tolerance, which is interpreted in the school setting as meaning that differences are endured as a burdensome part of a diverse population. Programs and activities are generally superficial at this level. Bulletin boards, assemblies, and units may be about Black History Month or a specific special person. An English as A Second Language program might be in place.

Acceptance is the next level of diversity support and shows a school's acknowledgement that differences are important. Programs and activities at this level might include a transitional bilingual program, multicultural fairs and programs, or communicating with parents through newsletters translated into their native language (Nieto, 2004).

The third level is *respect* and includes programs, lessons, and activities in which diversity is highly respected. The native language(s) of students is used in different ways throughout the school, interactions with parents would be frequent and positive and parents would have a voice in the school decisions. Students' cultural values and experiences would provide a significant connection to lessons and learnings in the classroom. In addition, multiple perspectives would be a pervasive throughout the curriculum of the school (Nieto 2004).

The fourth level in Nieto's (2004) model of multicultural education is *affirmation*, solidarity, and critique. This reflects the highest level of support for diversity. At this level, the race, culture, ethnicity, and language of the students, parents, and community are accepted as valid bridges for learning. Students learn about conflict, equity, and social justice as a way to act in order to support a more humane society (Nieto, 2004).

Bennett (2003) also indicates that multicultural education is an approach to teaching and learning and involves affirmation of cultural pluralism and democratic values and beliefs. The primary goal is "to foster the intellectual, social, and personal development of *all* students to their highest potential" (Bennett, 2003, p. 14)

The conceptual framework includes the four interactive dimensions of curriculum reform, equity pedagogy, multicultural competence, and teaching toward social justice. These are similar to Banks' work. *Curriculum reform* basically involves restructuring the curriculum to move a traditional monocultural perspective to one of multiple perspectives that includes active inquiry, development of new constructs and understandings about differences and various groups of people and their historical contributions. In addition,

this involves confronting the bias reflected in textbooks, media, and other types of educational materials. *Equity Pedagogy* is based on a total restructuring of the school environment to address meeting student needs based on equity and providing equal opportunity for all students. It involves a total transformation of the school climate including addressing the hidden curriculum that is most often articulated in teacher expectations of students learning. *Multicultural competence* encompasses the major ideas of development of personal ethnic identity, reducing prejudice, and negotiating competencies of other cultures. In addition, it involves helping students to understand how to navigate within various cultures while maintaining and affirming their own culture. This promotes the idea of cultural pluralism as the normal state of affairs. *Teaching toward social justice* combats prejudice and discrimination, stresses basic human rights, dignities, and similarities, focuses on developing awareness of historical aspects of cultural groups, and addresses current situations of racism, sexism, and classism. The very basic and profound aspect of this dimension is the resulting actions designed to end inequity, prejudice, and discrimination in everyday life.

Bennett (2003) also puts forth four core values and supporting goals that are "rooted in democratic ideals and American Indian philosophy" (p.16) and provide "the strong ethical foundations of multicultural education" (p. 16). The core values include responsibility to a world community, acceptance and appreciation of cultural diversity, reverence for the earth, and respect for human dignity and universal human rights. The goals supporting these values include (a) "to develop multiple historical perspectives," (b) to strengthen cultural consciousness," (c) "to strengthen intercultural competence,"

(d) "to combat racism, sexism, and other forms of prejudice and discrimination," (e) "to increase awareness of the state of the planet and global dynamics," and (f) " to build social action skills" (Bennett, 2003, p. 33).

Synthesizing the concepts in the models previously presented with additional informational pieces from experts in the field provides the foundational characteristics of multicultural education that address diversity issues and support high levels of student achievement. These concepts are the foundation of the instructional practices queried in this research study and outlined in the following components: multicultural education, multicultural teachers, multicultural schools, multicultural instructional materials, and multicultural instructional practices.

Multicultural education validates and affirms differences in class, race, sexuality, gender, ethnicity, and ability. It empowers students to be successful learners and contributing citizens, engages teachers and students to view society through the lens of multiple perspectives, and nurtures positive self-concepts in students. Multicultural education fosters intergroup understanding and respect for others' cultures, questions stereotypes, is antiracist, promotes understanding of the dynamics of dominance, supports equal distribution of power among groups, and teaches about how power, discrimination, and injustice shape the world (Banks, 1999; Gay, 2000; Howard, 1999; Leistyna, 2002; Merryfield, 2002; Nieto, 2004; Reissman, 1994).

Multicultural education is neither neutral nor apolitical, but is positioned from within a social justice perspective. This means that teachers and students alike are involved in learning how to think in more inclusive ways applied to real life, and seek to

develop a proactive citizenship directed at nurturing passion for justice to achieve a more humane society within community at local, national, and global levels (Howard, 1999; Nieto, 2004; Reissman, 1994; Zeichner, Melnick, & Gomez, 1996).

As multicultural educators, teachers are knowledgeable and competent in their content areas, know themselves in regard to their own racial, cultural, and ethnic identities, see themselves as members of a community, see their profession as a way to give back to society and encourage their students to do the same, show respect to all students, are reflective regarding professional practices, develop personal bonds with students, are committed to equity for all students, and believe they can make a difference through their teaching. Teachers, administrators, and counselors exhibit high expectations for all students and help students set and achieve positive career goals (Banks, 1999; Howard, 1999; Ladson-Billings as cited in Webb, 1998; Leistyna, 2002; Nieto, 2004; Reissman, 1994; Zeichner, Melnick, & Gomez, 1996).

Within the multicultural school, parents and community members are encouraged to be a part of school activities, programs, and planning, and have a voice in making important decisions within the school context (Banks, 1999; Leistyna, 2002; Zeichner, Melnick, & Gomez, 1996).

A multicultural school reflects a transformative, multidimensional pervasive school culture in which the physical environment, curriculum, and relationships among teachers, administrators, students, parents, and community reflect a total atmosphere of responsiveness to ethnic, racial, language, cultural, and ability differences. Curriculum and materials, the learning context, instructional techniques, assessments, and policies

proactively encompass diversity. Programs are continually monitored, bulletin board displays and activities reflect diversity, the staff is multiethnic and multicultural, the lunch program serves various ethnic foods, holidays and assemblies reflect cultural and ethnic diversity, and professional development programs are reflective of a continuous, comprehensive, and systematic multicultural approach. Lastly, teachers seek to teach the students the culture of the school while maintaining the students' sense of their own cultural identity and pride (Banks, 1999; Gay, 2000; Leistyna, 2002; Nieto, 2004; Zeichner, Melnick, & Gomez, 1996).

Instructional materials in a multicultural environment are inclusive of racial, ethnic, and cultural perspectives, incorporate multicultural information and resources in all content areas and through various skills, and acknowledge diverse heritages as legacies that affect how students learn, their attitudes, dispositions, and their worth to be taught. Materials are free from negative biases and can be used to help develop more positive racial and ethnic attitudes. Assessments do not contain stereotypical language, are reflective of multicultural perspectives, and procedures are culturally sensitive. Classes for talented and gifted students proportionately represent students from diverse backgrounds (Banks, 1999; Zeichner, Melnick, & Gomez, 1996; Gay, 2000; Leistyna, 2002).

Instructional practices in a multicultural learning environment provide for crosscultural experiential, authentic, collaborative, challenging, and interactive learning. The learning is seen as relevant and meaningful by students and teachers scaffold instruction to connect with the frames of reference, learning styles, prior experiences, motivational styles, and cultural knowledge of the students. Instructional practices empower, engage, create personalized learning, and are constructivist for students. Lessons are designed to use student strengths to construct knowledge and teachers recognize that the students' cultures and experiences are a worthwhile resource from which to use examples, data, or information in the teaching process (Banks, 1999; Gay, 2000; Leistyna, 2002; Merryfield, 2002; Zeichner, Melnick, & Gomez, 1996).

Briefly, to review, several models of multicultural education have been defined in the research and contain elements specific to aspects in the school environment. One commonality to the different models is that they are all basically leveled in accordance with a continuum based on the approach to diversity - from an awareness level to a level of social justice and advocacy. Specific to this study is the delineation of instructional practices within the framework of multicultural education and is the model for identifying the multicultural instructional practices employed by the international teachers in the study.

Specific to effective teaching strategies, Marzano, Pickering, & Pollack (2001) have completed a meta-analysis of the instructional strategies that have a positive major impact on student achievement. Their work combines the results "from a number of studies to determine the average effect of a given technique" (p. 4). The categories that are identified and further explained are (a) identifying similarities and differences, (b) summarizing and note taking, (c) reinforcing effort and providing recognition, (d) homework and practice, (e) nonlinguistic representation, (f) cooperative learning, (g) setting objectives and providing feedback, (h) generating and testing hypotheses, and (i)

questions, cues, and advance organizers. These strategies "have a high probability of enhancing student achievement for all students in all subject areas at all grade levels" (p. 7).

Summary

The purpose of this study was to compare perceptions about cross-cultural effectiveness of classroom teachers in international settings with classroom teachers in Iowa, and to examine instructional practices international teachers' perceived to be effective in meeting the needs of a diverse student population within a multicultural education framework. Specific to that purpose, Chapter Two reviewed the foundational research studies and available literature regarding the (a) historical foundations of multicultural education and international teaching, (b) diverse student populations compared to the teacher populations, (c) the link between teacher characteristics and student achievement, (d) effects of domestic and international field experiences on preservice and classroom teachers, and (e) effective educational practices within a multicultural education framework.

The past two centuries in the American educational system reflects a long-stated goal of education in the United States that all children should benefit from a public education, regardless of family background. However, as stated previously in this study, that has not been the case in the past, nor is it currently the case. Schools and school systems continue, in varying degrees, to be systematically unequal and unfair and students who are different in social class, race, ethnicity, language, and gender from what is considered the mainstream majority, have not been given an equitable, nor an equal

education. At the same time, international teaching programs increased in number and dimension or remained stagnant depending on the political atmosphere of the country at the time. However, there continued to remain various aspects of international teaching connections throughout the history of the nation while the population grew more and more diverse in its complexity of race, ethnicity, religious faith, culture, language, and abilities.

Currently, in regards to the statistical picture of students and teachers in the nations' schools, student population trends show a significant and continuing increase in student enrollment reflecting a wide variety of diverse groups. At the same time, male teachers and teachers from non-majority groups are declining in numbers, indicating a continuing increase of teachers who are White, female, monolingual, and from the middle socio-economic level. This indicates a mismatch between the backgrounds of students and teachers and has profound effects on student achievement. As teachers see their world from their own demographic lens, they may unconsciously form different values and expectations about students who are different in backgrounds from their own. Teacher expectations are shaped not only by their own cultural upbringing and experiences, but also by previously identified perceived abilities of students based on student ethnicity and family income. This, in turn, influences how they teach, directly impacting how students learn. This impacts the expectations teacher may hold for students, how they respond to students in the classroom, and how they view the student's family. The lack of cultural sensitivity on the part of teachers builds barriers to positive student achievement.

One approach to building preservice and classroom teachers' cultural awareness and sensitivity that supports high expectations for student achievement is cross-cultural domestic and international teaching experiences. Cross-cultural field experiences, within the United States or in international locations, have positive impact in contributing to the understandings and expectations of students whose backgrounds differ from preservice and classroom teachers. This in turn, promotes enhanced classroom instruction and family connections to meet individual student needs and advance high academic achievement. However, the research is limited in regard to classroom teachers and their instructional practices specific to multicultural education, which frames the expected behaviors and strategies teachers need to be using in order to support high expectations for student achievement in diverse classrooms. Thus, this study examines the cross-cultural effectiveness of Iowa and international teachers and investigates the instructional practices of international teachers within the multicultural education research framework.

CHAPTER 3

METHODS AND PROCEDURES

Introduction

The purpose of the study was to investigate differences in cross-cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices international teachers perceive to be effective in meeting the needs of a diverse student population. The purpose of this chapter is to describe the design of the study, describe the measurement instruments, identify the population of interest, and delineate the specific procedures regarding data collection and analysis.

Design of the Study

Both quantitative and qualitative elements (Miles & Huberman, 1994) were used in this study as a way to provide a comparative framework for establishing the respondents' cross-cultural effectiveness for two sample groups of teachers, international teachers and teachers in northeast Iowa (Appendix A), and for providing deeper, rich description of teaching practices, within a multicultural education framework, specific to international teachers. The three data gathering methods included (a) a published inventory relating to cross-cultural adaptability, (b) narrative, open-ended survey questions specific to teaching practices, and (c) in-depth follow-up discussion questions that further investigated the strands that arose from the survey's narrative, open-ended questions.

In adding to the limited research that focuses on the perceived effects of international settings and diverse student populations on teaching practices, the study first established the foundation of the respondents' cross-cultural effectiveness through the published inventory and, secondly, through the survey open-ended questions and the follow-up discussion questions, the study emphasized discovery and exploration, rather than validation and confirmation, to investigate international teacher perspectives, within a multicultural education framework, regarding instructional practices. The study used thick description of teaching practices, which added substance to the results of the inventory, a main function of qualitative research (Krathwohl, 1993) and expanded the conceptual framework (Miles & Huberman, 1994) of the multicultural education research from within the reality of the international teachers.

Demographic Questions

The initial section of the survey was comprised of demographic questions (Appendix B). Results from this section were used to describe characteristics of the participants and were also used to provide a matched sample of Iowa teachers to the international teachers.

Inventory

The Cross-Cultural Adaptability Inventory (CCAI; Appendix C), the quantitative data source, provided numerical data that established cross-cultural effectiveness results for comparison analysis. Two groups of teachers completed the inventory; a sample of classroom teachers from international schools and a matched, sample of classroom

teachers from Iowa schools. The inventory was posted for a specific period of time on an internet site via an online survey tool for teachers to access and complete.

Narrative, Open-Ended Questions

The international classroom teachers completed narrative, open-ended questions (Appendix D), based on the research framework that identifies effective multicultural teaching practices. These questions were posted for a specific period of time on an Internet site via an online survey tool for teachers to access and complete. The narrative, open-ended questions were specific to gathering information regarding the international teachers' instructional practices. The responses were analyzed and then synthesized into themes for further probing through the discussion questions.

Discussion Questions

From the synthesized themes that arose out of the narrative, open-ended questions, further inquiry was completed through discussion questions (Appendix E) completed by the international classroom teachers. The discussion questions were posted via online email communication in a discussion type format. The researcher posed questions and the teachers initially responded to each of the four questions and were also be able to respond to the answers posted by the other international teachers. The questions were posted for a limited time frame. This method created asynchronous discussions in email and print format and provided further depth to identified themes regarding instructional practices from the narrative, open-ended questions. To provide robust research, the discussion questions combined exploration of the identified strands, while narrowing the focus to specific in-depth discovery (Krathwohl, 1993).

The narrative, open-ended survey questions and the follow-up discussion questions comprised the qualitative data sources for the study and provided the deeper, rich description of specific teaching practices, within the multicultural education framework, related to international settings and diverse student populations. This data source provided well-grounded, rich descriptions and explanations of which Miles and Huberman speak (1994) and that, in this study, supported the investigation of instructional practices of international teachers that are identifiable within multicultural education contexts.

Measurement Instruments and Materials

For the purposes of this study, the term survey was used to denote the (a) demographic questions, (b) the Cross-Cultural Adaptability Inventory (CCAI), and (c) the narrative, open-ended questions. These three components were all included within the online survey tool and divided as three sections with separate directions for each section. The discussion questions were not regarded as part of the survey, but were completed separately from the survey.

Demographic Data

The initial portion of the survey contained questions specific to demographic data. These questions elicited variable information from the teachers regarding gender, race, age, predominant grade level and content subject currently taught at the time of the survey, total years of teaching experience, total years of international teaching experience, level of education, cultural context of student teaching experience, cultural context of school attended when growing up, Iowa or international setting of current

teaching assignment, enrollment size of the school or school system, and cultural composition and racial composition of the current school population in which they were teaching.

Cross-Cultural Adaptability Inventory (CCAI)

The Cross-Cultural Adaptability Inventory, (CCAI; Kelley & Meyers, 1995), the second part of the survey, was a self-awareness instrument designed to provide information about the respondents' cross-cultural effectiveness and was completed as an online inventory by both the participants in Iowa and those in international locations. The inventory contained 50 items that assessed cross-cultural adaptability in four dimensions (subscales): emotional resilience (ER), flexibility/openness (FO), perceptual acuity (PAC), and personal autonomy (PA).

Emotional resilience refers to an individual's emotional capacity to deal effectively with the frustrations and stress of living in different, unfamiliar cultures and their reactions to that new culture. Flexibility/openness refers to an individual's attitudes toward a different culture in regards to their curiosity, natural respect, readiness to listen to other viewpoints, seeking to become acquainted with others in order to understand their point of view. Perceptual acuity refers to an individual's ability to empathize with those who are different from oneself, being receptive to others' ideas and behaviors, and not attributing negative elements based on perceived differences. This usually involves a shift in one's cultural worldview. Perceptual acuity also includes the ability to be competent in communication skills, comprehending both verbal and non-verbal contextual cues. Personal autonomy refers to the ability to preserve self-concept, or self-

identity, with a high sense of self-esteem while respecting a different culture and amid changing world perspectives within oneself. Basically, it means that one can be open to new cultural experiences while maintaining one's own identity without feeling threatened by the differences between the cultures (Kelley & Meyers, 1995).

The CCAI inventory was designed to be culture-general. "The culture-general approach assumes that individuals adapting to other cultures share common feelings, perceptions, and experiences. This occurs regardless of the cultural background of the person or characteristics of the target culture" (Kelley & Meyers, 1995, p. 2). The CCAI can be used in numerous ways for individual respondents or as a basis for understanding various research findings. In addition, it assists in helping individuals understand the factors and qualities foundational to cross-cultural effectiveness (Kelley & Meyers, 1995). For the purposes of this study, the CCAI was used to investigate the respondents' relative weaknesses and strengths in relation to the four dimensions of the inventory and in regard to respondents' overall cross-cultural adaptability. Results were compared between the two groups of teachers, international teachers from various worldwide locations and Iowa teachers from the Area Education Agency 267 area. The results from the CCAI were also used in establishing a perspective when analyzing the narrative, open-ended questions and discussion questions.

Each item response in the CCAI was based on a type of Likert scale that included six response options ranging from "Definitely True" to "Definitely Not True." Each option range was given an assigned value from 1 and 6, beginning with "Definitely True" being 6. Nine of the items were reversed scored. "I can function in situations where

things are not clear" (Kelley & Meyers, 1995, p. 57) and "I am not good at understanding people when they are different from me" (Kelley & Meyers, 1995, p. 59) are sample items from the CCAI and reflect the types of statements to which participants responded.

Items for each respondent were totaled to provide a raw score for each dimension and a total overall score. The total score provided the most reliable overall indicator of cross-cultural adaptability. An individual, or group, with a high total score has the most critical skills that are considered to be needed for effectively interacting with people from different cultures. Individuals who have a sense of self, display emotional equilibrium, have mental flexibility, are open to others' differences, and have solid communication skills are more likely to do well in cross-cultural situations (Kelley & Meyers, 1995).

Statistically, in determining reliability and validity, the authors, Kelley & Meyers (1995), subjected the CCAI to a factor analysis. Revisions were made and norms and reliability were established on a group of 653 respondents. On each of the four dimensions, internal consistency (standardized alpha) was established. Using Cronbach's Alpha, the emotional resilience dimension scale was .82, the flexibility/openness dimension scale was .80, the perceptual acuity dimension scale was .78, the personal autonomy dimension scale was .68, and the overall reliability total was .90. Correlations between the scales ranged from .27 to .59. The emotional resilience dimension contained eighteen items, the flexibility/openness dimension contained fifteen items, the perceptual acuity dimension contained ten items, and the personal autonomy dimension contained seven items. Additional statistical information from the authors indicated that the CCAI

was considered to have evidence of face, content, and construct validity (Kelley & Meyers, 1995).

Narrative, Open-Ended Questions

The third component of the online survey followed the CCAI and contained narrative, open-ended questions that were completed only by the international teachers. The questions were directed specifically toward instructional practices of the international teachers and were derived from the literature framework defining the teaching practices of multicultural teachers. The questions were stated to elicit explanatory, in-depth responses and analytic induction was used to find commonalities, differences, patterns, and themes in the data and provided the foundation for the in-depth discussion questions. For the purposes of this study, analytic induction was defined as a process that includes finding common elements in the data, describing and explaining the elements and linking the findings with research found in literature (Krathwohl, 1993; Miles & Huberman, 1994). The strands, or themes, that arose out of the narrative, openended questions served as the foundation for additional data collection in the form of indepth discussion questions (Miles & Huberman, 1994).

Discussion Questions

From the findings in the survey's narrative, open-ended questions, and from the research framework identifying effective multicultural teaching practices, in-depth discussion questions were formulated and used in a semi-structured discussion format with the 20 international teachers via online, email messaging in a discussion type format. The discussion questions combined exploration of the identified strands from the

narrative, open-ended questions, while narrowing the focus to specific in-depth discovery (Krathwohl, 1993), which allowed the researcher to focus the discussion to the agenda of the study. The answers were also checked against current literature findings and matched to constructs (Krathwohl, 1993; Miles & Huberman, 1994).

Population of Interest

The subjects involved in this study were from two non-probability, purposive sample groups of teachers, chosen to match the relevant theory and research base in which this study was grounded (Krathwohl, 1993; Miles & Huberman, 1994). The teachers making up the first sample group were known to the researcher and included 20 teachers at various international locations who were currently teaching in an accredited, international school that utilized American curriculum and provided instruction in English. The group included elementary through high school level teachers, grades K-12, males and females, various years of teaching experience, wide range of age levels, and different content areas in general education, special education, and specialist areas. The second group of 20 teachers was a sample from schools located in Iowa within the Area Education Agency 267 district. The Iowa teachers were matched to the international sample of teachers in regards to gender (male, female), years of teaching experience (less than 5 years to more than 25 years), and level of teaching (elementary, middle, and high school levels).

Non-probability, purposive sampling was used, rather than random sampling, primarily for the purpose of specific exploration and discovery and to extend and densify understanding, find the limits of the generalities, and to ensure validity of the information

from the study (Krathwohl, 1993; Miles & Huberman, 1994). The international teachers in the study were selected "for their sensitivity, knowledge, and insights" (Krathwohl, 1993, p. 326) regarding their international experiences and "their willingness to talk about it" (Krathwohl, 1993, p. 326). In addition, due to the difficulty of identifying the larger population of international teachers from which to randomly sample, the twenty international teachers were known to the researcher and identified to participate in the study.

Data Collection and Analysis

The effects of international setting and diverse student populations on instructional practices as identified through cross-cultural adaptability and specific multicultural teaching practices was the significant focus of this study. Demographics of the two teacher sample populations were summarized. Descriptive analyses were completed regarding the CCAI overall results, as well as in regard to the four dimensions to investigate cross-cultural effectiveness between the two groups of teachers; means and standard deviations were computed. Inferential statistical analyses were conducted to investigate the differences between the international teachers and Iowa teachers.

Qualitative data from the narrative, open-ended survey questions and discussion questions were analyzed for revealed patterns regarding multicultural teaching practices of the international teachers.

Data Collection

Step 1. Twenty individual teachers, known to this researcher and teaching in thirteen various international schools districts, were contacted for participation in the

study. Introductory, explanatory emails (Appendix F) were sent to each of the teachers requesting their participation. After participation was agreed to, via response to the email invitation, the Informed Consent was secured via email (Appendix G).

Step 2. Due to copyright limitations, a hard copy of the CCAI was mailed to each participating international teacher via United States postal services global express. The mailed packet contained the published CCAI, directions for accessing the online Internet survey tool to complete the survey items, and the time frame for completing the survey (Appendix H). Each international teacher was able to complete the demographic items, the CCAI, and the narrative, open-ended questions through the online Internet survey site that was accessed asynchronously by the participating international teachers at their convenience. The survey was available for approximately a four-week period of time. The deadline for completing the survey was extended due to delays with the delivery of the packet containing the directions and the CCAI to some international locations. Demographic data was included at the beginning of the survey, the CCAI was the second part, and the narrative, open-ended questions were the last segment of the survey. The narrative questions explored information specific to teaching practices and were derived from the research framework that identifies effective multicultural teaching practices.

Step 3. The last segment for the international teachers was participation in a discussion phase. The discussion questions were derived from the identified strands that arose out of the narrative, open-ended survey answers specific to effective instructional practices identified in the multicultural education research. Each of the four questions was posted individually at different times and not all at once. Participants were able to

respond in a limited discussion style when questions were posted. The discussions were asynchronous and utilized email messaging as a discussion tool and format. The period to answer and respond to in-depth discussion questions was approximately four weeks. Responses to the discussion items were analyzed for qualitative data patterns and linked to research.

Step 4. Upon completion of the online survey by the international teachers, a matched sample of 20 Iowa teachers from Area Education Agency 267 school districts was identified to complete the demographic questions and the CCAI portion of the survey. The Iowa teachers were matched to the international sample of teachers in regards to gender (male, female), years of teaching experience (less than 5 years to more than 25 years), and level of teaching (elementary, middle, and high school levels). Eighteen superintendents and two principals in the Iowa Area Education Agency 267 were contacted via a phone call eliciting their assistance in identifying teachers in their district who met the matched criteria and were willing to participate in the study (Appendix I). An email and a mailed packet that included information regarding the study and a chart for the superintendents to complete regarding identified teachers were sent after the phone call (Appendix J). In the larger metropolitan area, the principals of specific schools were contacted rather than the superintendent. Reminder phone calls were also made if needed. Once the superintendent, or principal, returned the information identifying teachers in their district willing to participate, an email was sent to the identified teachers further explaining the study and eliciting an email reply

indicating their agreement to participate (Appendix K). After participation was agreed to, the Informed Consent was secured via email (Appendix L).

Step 5. A hard copy of the CCAI was mailed to each participating Iowa teacher via priority mail. The mailed packet contained the CCAI and directions for accessing the online Internet survey tool to complete the survey items (Appendix M). Each Iowa teacher was requested to complete the demographic items and the CCAI through the online Internet survey site that was accessed asynchronously by the participating Iowa teachers at their convenience. The survey was available for approximately a four-week period of time. Phone contact was made when necessary to encourage teachers to complete the survey. The Iowa teachers were required to complete only the demographic and CCAI survey sections, and not the narrative, open-ended questions nor the discussion questions, which were specific only to the international teachers.

Because all data was collected through electronic means, all teachers, both in Iowa and in international locations, had access to the researcher via email throughout the time of data collection for the purpose of clarifying questions the respondents may have had, to assist with technology challenges, and to clarify concerns. No teachers contacted the researcher during the study.

Data Analysis

Demographics of the two teacher sample populations were reviewed and summarized. Descriptive statistical analyses were employed in regard to the study. Data results from the CCAI for the two sample populations of teachers were used to examine overall cross-cultural effectiveness as well as cross-culture effectiveness in regard to the

dimensions: emotional resilience, flexibility/openness, perceptual acuity, and personal autonomy.

Inferential statistical analyses were conducted to investigate the differences between the international teachers and Iowa teachers. Independent sample t-test was used on each of the dimensions and the total score.

Using analytic induction, the qualitative data from the narrative, open-ended survey questions and the discussion questions were coded and sorted to find commonalities, differences, patterns, and themes leading to description and generalizations and compared to current literature findings regarding multicultural teaching practices (Krathwohl, 1993; Miles & Huberman, 1994). The strands found in the narrative, open-ended questions led to the four discussion questions for deeper analysis that contributed to and extended the multicultural education research framework (Miles & Huberman, 1994).

All forms of data were synthesized to identify teachers' cross-cultural effectiveness and to identify effective instructional practices of the international teachers. Triangulation was achieved through compilation and synthesis of the data from the CCAI (a normed inventory), the narrative, open-ended survey questions relating to teaching practices, the follow-up discussion questions, and review of supporting literature research (Krathwohl, 1993).

CHAPTER 4

RESULTS OF THE STUDY

Introduction

The purpose of the study was to investigate differences in cross-cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices international teachers perceive to be effective in meeting the needs of a diverse student population. To achieve this purpose, twenty international teachers and twenty Iowa teachers completed the Cross-Cultural Adaptability Inventory, which provided statistical data regarding cross-cultural effectiveness of the two groups. As part of their survey, the international teachers completed narrative, open-ended questions regarding instructional practices, which were analyzed for strands that led to explicit discussion questions focusing on instructional practices. The international group of teachers also answered the discussion questions. The purpose of this chapter is to present the data results which include descriptions of the demographic characteristics of the participants, the statistical results of the inventory, the results of the narrative, open-ended questions and results of the discussion questions.

Demographic Characteristics of the Participants

The participants in this study (N = 40) were comprised of two specific sample groups of teachers. The first group of teachers (n = 20) were known to the researcher and were teaching at various international locations in accredited, international schools that utilized American curriculum and provided instruction in English (Table 3). Of this

group of teachers, 55% were from countries in Asia, 15% were from Central America, 10% were from Europe, 10% were from South America, 5% were from the Middle East, and 5% were from Africa.

Table 3
School Locations of International Teachers

Location	Region	n	%
Aviano, Italy	Europe	1	5
Baku, Azerbaijan	Asia	2	10
Beijing, China	Asia	1	5
Caracas, Venezuela	South America	2	10
Hong Kong, China	Asia	. 1	5
Johannesburg, South Africa	Africa	1	5
Monterrey, Mexico	Central America	3	15
Mumbai, India	Asia	. 1	5
New Delhi, India	Asia	4	20
Okinawa, Japan	Asia	1	5
Taipei, Taiwan	Asia	1	5
Wiesbaden, Germany	Europe	1	5
Yanbu, Saudi Arabia	Middle East	1	5

The second group of teachers (n = 20) were from schools located in Iowa within the Area Education Agency 267 district (Table 4). Of the 18 superintendents and two principals contacted and asked to participate in identifying teachers in their district willing to participate who matched the specific criteria of gender, level of teaching, and years of teaching, 17 responded in the affirmative. Eleven (65%) of the 18

Table 4
School Locations of Iowa Teachers

Location	n	%	
Charles City, Iowa	1	5	
Clarksville, Iowa	1	5	
Dysart, Iowa	1.	5	
Denver, Iowa	. 1	5	
Fairbank, Iowa	. 1	5	
Iowa Falls, Iowa	3	15	
Janesville, Iowa	1	5	
LaPorte City, Iowa	1.	5	•
Shell Rock, Iowa	1,	5	•
Tripoli, Iowa	1	5	
Waterloo, Iowa	5	25	
Waverly, Iowa	3	15	

superintendents and principals actually submitted information. Of the information submitted, 75 Iowa teachers were contacted and invited to participate and more than the needed number of 20 teachers initially responded who met the matched criteria. Those who responded first and met the criteria were accepted to participate in the study.

Of the international group of teachers and the Iowa teachers total (N = 40), 55% were male and 45% were female and all were of the same race, Caucasian.

For the most part, the ages of the teachers in the two groups were fairly similar (Table 5), however, the international teachers tended to be more in the upper ranges of age than the Iowa teachers and less in the 40 to 49 years range than the Iowa teachers. The majority of the international teachers (50%) were between the ages of 50 and 59 years of age, and the majority of the Iowa teachers were equally divided (40% each) in the age categories of 40 to 49 years and 50 to 59 years. The majority of both groups of teachers were more than 40 years of age (77.5%).

Table 5

Age in Years of Teacher Participants

Years	Iowa	International	n	%	
29 years or less	2	1	3	7.5	
30 – 39 years	2	4	6	15.0	
40 – 49 years	8	3	11	27.5	
50 – 59 years	8	10	18	45.0	
60 years and more	0	2	2	5.0	

In both groups of teachers, international and Iowa, the highest percent of teachers were teaching at the high school level, (45%; Table 6). This included grades nine through 12. The next highest percent (40%) were teaching at the elementary level. This included grades kindergarten through either 5th or 6th grade. The smallest percent of teachers (15%) were teaching at the middle school level. This included grades 5 or 6 through 8th grade.

Table 6

Levels of Teaching (Matched Variable)

Teaching Level	Iowa teachers	International teachers	n	%
Elementary	8	8	16	40.0
Middle School	3	3	6	15.0
High School	9	9	18	45.0

Note. N = 40; International group n = 20, Iowa group n = 20.

Content subject areas in which the teachers were instructing varied between the two groups (Table 7). The elementary (generalist) content subject area was similar between the two groups with 37.5% of the total number of teachers instructing in that subject area. There were more reading/English teachers and social studies teachers in the Iowa group. Math and the category of 'other' were similar between the two groups. There were no science teachers and no specialist teachers (music, art, physical education, industrial technology, etc.) in the Iowa group. The lack of science and specialist teachers participating in the study from the group of Iowa teachers is a limitation of the study. In

a larger sample of the teaching population, science and specialist teachers would be represented.

Table 7

Content Subject Areas of the Teachers

Content area	Iowa teachers Int	ternational teachers	n	%	
Elementary	8	7	15	37.5	_
Math	1	2	3	7.5	
Reading/English	4	. 1	5	12.5	
Science	0	4	4	10.0	
Social Studies	5	3	8	20.0	
Specialist areas	0	2	2	5.0	
Other	2	1	3	7.5	

Note. N = 40; International group n = 20, Iowa group n = 20.

The total years of teaching experience (Table 8) was intended to be a matched criterion between the two groups of teachers, but when the data was reviewed the match was not an exact fit. However, the match was similar enough between the two groups to support the intent of the study. The majority of the teachers (55%) had more than 21 years teaching experience. Eighteen of the teachers (45%) had 15 years or less of teaching experience. There were no teachers in either group that had between 16 and 20 years teaching experience.

Table 8

Total Years of Teaching Experience (Matched Criteria)

Years	Iowa teachers	International teachers	n	%
Less than 5	2	2	4	10.0
6 – 10	1	2	3	7.5
11 – 15	5	6	11	27.5
16 – 20	0	0	0 ,	0.0
21 – 25	3	2	5	12.5
More than 25	9	8	17	42.5

Note. N = 40; International group n = 20, Iowa group n = 20.

In regard to the number of years teaching in an international location (Table 9), all of the Iowa teachers had taught five years or less in an international school. This reflects what was expected with the design of the study. Of the international teachers, 15 of the 20 teachers (75%) had taught 15 years or less in an international setting. Five of the teachers (25%) had taught in an international location more than 15 years and only one had taught more than 25 years in an international location.

When looking at the level of education data in both groups, more of the international group of teachers had advanced degrees (Table 10). Five teachers (25%) in the Iowa group held master degrees and 15 teachers (75%) in the international group held masters degrees or above. Conversely, 15 of the Iowa teachers (75%) held bachelor degrees and five of the international teachers (25%) held bachelor degrees.

Table 9

Years Taught in an International Location

Years	Iowa teachers	International teachers	n	%	
5 or less	20	4	24	60.0	
6-10	0	6	6	15.0	
11 – 15	0	5	5	12.5	
16 – 20	0	2	2	5.0	
21 – 25	0	2	2	5.0	
More than 25	0	1.	1	2.5	

Note. N = 40; International group n = 20, Iowa group n = 20.

Table 10

Level of Education

Degree level	Iowa teachers	International teachers	n	%	
BA	4	4	8	20.5	
BA + 15	3	0 2 2	3	7.5	
BA + 30	8	1	9	22.5	
MA	1	5	6	15.0	
MA + 15	1	1	2	5.0	
MA + 30	3	8	11	27.5	
EdD/PhD	0	1	1	2.5	

In regard to their student teaching experiences, the international teachers had more experiences that were multicultural and international than the Iowa teachers (Table 11). Even with that observation, 26 of the 40 teachers in the study (65%) indicated their experience was in a single cultural setting in the United States and in that number there were far more Iowa teachers (18) than international teachers (eight) in this category. In those experiences, 25 of the 26 teachers indicated the predominant culture was Caucasian and one teacher indicated the predominant culture in the school was African-American.

Table 11

Cultural Environment of Student Teaching Experience

Туре	Iowa teachers	International teachers	n	%
United States				
multicultural setting	2	7	9	22.5
single culture setting	18	8	26	65.0
International location	0	4	4	10.0
Other	0	1	1	2.5

Note. N = 40; International group n = 20, Iowa group n = 20.

There were more than twice as many international teachers (seven) than Iowa teachers (two) who reported their student teaching experience was in a multicultural setting in the United States. Of these teachers, the cultures reported as being present in the schools included Caucasian, African-American, Asian, Hispanic, Hawaiian, Filipino, and Japanese. Four international teachers indicated experiencing student teaching in an

international location with experiences in Denmark, United Kingdom, and Canada. One international teacher, in the 'other' category, indicated not having a student teaching experience at all.

The two groups of teachers, international and Iowa, were similarly dispersed in regards to the cultural environments of the schools they attended while growing up as children and students (Table 12). Eighty per cent of the two groups experienced schooling in single culture settings with Caucasian as being the reported culture. One teacher reported that African-American was the dominant race of the school attended while growing up. Of the 20% reporting growing up in a multicultural school setting, the cultures and races reported to be represented in that setting included Caucasian, African-American, Hispanic, Euro-American, and Native American. One international teacher reported growing up in international schools.

Table 12

Cultural Environment of School When Growing Up

Туре	Iowa teachers	International teachers	n	%	
Multicultural setting	5	3	8	20.0	
Single culture setting	15	17	32	80.0	

Note. N = 40; International group n = 20, Iowa group n = 20.

Student enrollment data (Table 13) indicates the participants in the study were teaching in schools that ranged from smaller schools, 100 – 300 students to larger schools, more than 1,000 students. Numbers of teachers reporting in each category were

similar between the two groups. The category with the largest number of teacher reporting (30%) was the category of more than 1,000 students. Approximately half of the teachers (47.5%) were teaching in schools between 100 and 500 students and 32.5% were teaching in schools between 700 students and more than 1,000.

Table 13
Student Enrollment of Current School

Number of students	Iowa teachers	International teachers	n	%	"
Less than 100	0	0	0	0.0	
100 – 300	6	3	9	22.5	
300 – 500	6	4	10	25.0	
500 – 700	1	3	4	10.0	
700 – 1,000	2	3;	5	12.5	
More than 1,000	5	7.	12	30.0	

Note. N = 40; International group n = 20, Iowa group n = 20.

Racial compositions at the participants' schools were distributed as would be expected. Students in the Iowa schools were far more likely to be Caucasian, or White, and less likely to be African-American, Hispanic, Asian, or Native American. Average percentages were Caucasian 91%, African-American 5%, Hispanic 5%, Asian 1%, and Native American 1%. The range percentages for each category was interesting to note in the Iowa schools. The percent of Caucasian students in the schools ranged from 74% to 100%. The African-American ranges were from 1% in some Iowa schools to 23% of the

student population in one school. The Hispanic student populations ranged from 1% in some schools to as much as 29% of the student population. With both the Asian and Native American student populations in Iowa, the ranges were from 1% to 3% in schools.

Students in the international schools were more likely to be from a variety of races. The international schools, as a whole, reflected similar average racial percentages for Caucasian (34%), Hispanic (31%), and Asian (39%), with each accounting for about one-third of the school populations. African-American students accounted for an average of 10% of student populations in the international schools and Native Americans accounted for about 3% of student populations. The ranges in the percents of the student racial groups in different schools were quite broad. For example, in some schools only 1% of the students were Caucasian while in others 98% of the students were Caucasian. This is a similar reality for the percents of Hispanic students in different schools – from 1% to 98% of the student population were Hispanic. Also, the same situation was reflected for Asian students - from 1% to 96% of students in different schools were Asian. The percent range of African-American students was from 1% to 28% in various schools. This wide range of variations of racial distributions in the various international schools can be attributed to the location of the schools and the culture reflected in the larger social community, which is mirrored in the cultural composition of the local schools. One would expect to find these varying ranges and percentages in international schools.

When teacher participants were asked to indicate the cultural composition of their respective schools a variety of answers were provided that for the most part represented

national cultures and/or racial groups; however, the results support the intent of the study. As expected, the Iowa teachers reported large percentages of American (or United States) culture, with ranges from 75% to 100% of students who were American in culture. Additional cultures that were included reflected small percentages of school populations. These were Bosnian, Mexican, Chinese, Asian, Korean, Filipino, Native American, and African. Three Iowa teachers indicated that foreign exchange students from Germany, Korea, and other countries were also attending their schools. Racial groups that were reported as culture groups included Hispanic and African American.

As would be expected, the cultural compositions of the international schools looked like an attendance list for an event at the United Nations. Nineteen of the international teachers indicated American students were attending their schools and the percentages of those students ranged from less than one percent to 100 percent. Seven teachers indicated Korean students attended the school, six indicated student from the United Kingdom attend their schools, four indicated students from India attended their schools, four indicated students from Canada attended their schools, three indicated European students attended their schools, and three indicated Mexican children attended their schools. Additional culture groups listed in the response included Venezuelan, Chinese, Australian, New Zealand, South African, Irish, British, Canadian, Scottish, Norwegian, Spanish, Colombian, Vietnamese, Trinidadian, French, German, Thai, Belgian, Samoan, Singaporean, Swiss, Bulgarian, Georgian, Ukrainian, Nigerian, Italian, Azeri, Pakistani, Sri Lankan, Filipino, Arabic, and Malaysian. On several responses,

teachers indicated they had students from "more than 50 different countries," or "from 54 different countries," etc. attending their school.

Statistical Analyses of the CCAI Results

To answer the quantitative research questions, descriptive and inferential statistical analyses were used to ascertain the differences in cross cultural effectiveness between the international teachers and the Iowa teachers as measured by the CCAI (Appendix N). The analyses related to the following research questions:

- 1. To what extent is there a difference in overall cross cultural effectiveness (Kelley & Meyers, 1995) between teachers in international schools and teachers in Iowa schools?
- 2. To what extent is there a difference in emotional resilience (Kelley & Meyers, 1995) in regard to cross cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 3. To what extent is there a difference in flexibility/openness (Kelley & Meyers, 1995) in regard to cross cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 4. To what extent is there a difference in perceptual acuity (Kelley & Meyers, 1995) in regard to cross cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 5. To what extent is there a difference in personal autonomy (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?

The hypotheses related to the research questions were stated as follows:

- 1. The teachers in international schools will be more cross culturally effective overall (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 2. The teachers in international schools will be more cross culturally effective in regard to emotional resilience (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 3. The teachers in international schools will be more cross culturally effective in regard to flexibility and openness (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 4. The teachers in international schools will be more cross culturally effective in regard to perceptual acuity (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 5. The teachers in international schools will be more cross cultural effective in regard to personal autonomy (Kelley & Meyers, 1995) than the teachers in Iowa schools.

Means and standard deviations were determined for both sample groups (Table 14) and an independent samples t-test (Table 15) was executed to investigate cross-cultural effectiveness between the international teachers and the Iowa teachers. Results indicated that in the total score, the international group of teachers were more cross-culturally effective with a marginally significant difference of, t (38) = 1.73, p = .09. Vogt (1999) explains the term marginally significant as "research results that fail to exceed the critical value needed to be statistically significant but that come close enough" and that the researcher finds important to discuss in the study (p. 169). In this particular study marginally significant will be defined as $0.05 \le p \le 0.10$ and the overall difference

indicating the international teachers were more cross culturally effective than the Iowa teachers is important to mention. In the dimension of emotional resilience (ER), the difference between the international teachers and the Iowa teachers was statistically significant, t (38) = 2.84, p = .01. In the dimension of flexibility and openness (FO), the difference between the international teachers and the Iowa teachers was not statistically significant, t (38) = .72, p = .48. In the dimension of perceptual acuity (PAC), the difference between the international teachers and the Iowa teachers was not statistically

Table 14

CCAI - Descriptive Statistics

CCAI Scale	Group	M	SD	SE Mean	
Total	International	240.40	15.21	3.40	
	Iowa	231.80	16.18	3.62	
ER	International	86.50	5.98	1.34	
	Iowa	80.70	6.91	1.55	
FO	International	71.35	5.91	1.32	
	Iowa	70.05	5.55	1.24	
PAC	International	47.75	3.42	.76	
	Iowa	46.95	4.59	1.03	
PA	International	34.80	2.80	.63	
t	Iowa	34.10	3.31	.74	

significant, t (38) = .63, p = .54. In the dimension of personal autonomy (PA), the difference between the international teachers and the Iowa teachers was not statistically significant, t (38) = .72, p = .48.

Table 15

CCAI - Independent Samples t-Test Between International and Iowa Groups

CCAI Scale	t	df	Sig.	M difference	
Total	1.73	38	.09	8.60	
ER	2.84	38	.01	5.80	
FO	.72	38	.48	1.30	
PAC	.63	38	.54	.80	
PA	.72	38	.48	.70	

Note. N = 40; International group n = 20, Iowa group n = 20. Difference is international minus Iowa.

It was decided to use Cohen's conventions (1988, p. 25, 26) in determining effect size of the results. The difference between the international teachers and Iowa teachers on the emotional resilience (ER) dimension yielded a large effect size (.83) and the difference between the international teachers and Iowa teachers on the total CCAI scores yielded a medium effect size (.53). Differences between the international teachers and Iowa teachers in regards to the flexibility/openness (FO), perceptual acuity (PAC), and personal autonomy (PA) dimensions yielded small effect sizes.

Internal consistency reliability of the CCAI inventory responses was measured using Cronbach's alpha coefficient analysis. Reliability results reflected .79 on the emotional resilience dimension, .71 on the flexibility/openness dimension, .73 on the perceptual acuity dimension, and .57 on the personal autonomy dimension. Thus, on three of the four dimensions, ER, FO, and PAC, results reflect that the individual items measure the constructs as intended within the dimensions and within the overall inventory (Krathwohl, 1993, p. 207) and that the items are consistent and measure the same underlying characteristics in the dimensions (Huck and Cormier, 1996, p. 80). On the PA dimension, it would appear questionable that the items measure the same construct to a high degree of reliability.

Thus, the overall findings indicate that the international teachers appear to be more cross culturally effective than the Iowa teachers using the CCAI (Kelley & Meyers, 1995), but in regard to statistical significance, the findings support only the second Hypothesis. Hypothesis 1 is marginally supported and Hypotheses 3, 4, and 5 were not statistically supported.

Narrative, Open-Ended Question Results

To answer the qualitative research questions, the narrative, open-ended survey items were completed by the international teachers only and not by the Iowa teachers, reflecting the design of the research study. All twenty of the international teachers answered all eight of the narrative, open-ended questions that were the last part of the survey. Responses (Appendix O) were analyzed and coded to investigate for specific strands relating to instructional practices of the teachers (Table 16) and as defined in the

Table 16

Themes Revealed in the Narrative, Open-Ended Questions

	Narrative, Open-Ended Question	Themes From Responses	# of Teacher Responses
1	What types of activities, lessons, do you use to bridge, or connect, the students' cultures to classroom learning?	 (a) specific curriculum units and activities (b) various content or subject matter connected to culture (c) holidays, ceremonies, and festivals (d) connection to the students' home and cultures 	7 13 7 5
2	How do you personalize instructional delivery and your teaching to students' cultures, learning styles, and diversity in your classroom?	(a) types of instructional strategies (b) language strategies (c) connecting to the students' home cultures	15 7 10
3	How do you modify tests to provide fair assessments for diverse student populations in regard to race, ethnicity, language, and/or ability?	 (a) ESL and special education students (b) student cultures (c) vocabulary (d) general modification strategies 	8 4 4 12
4	How do the materials, used in your lessons/instruction reflect cultural non-bias? What process is used in your school to ensure the use of culturally non-biased materials?	 (a) textbooks (b) how teachers augment materials (c) specific programs or approaches (d) the process used to ensure non-biased materials are used 	9 6 3 5
5	How do you provide meaningful tasks for students that are responsive to race, culture, ethnicity, and language, in a culturally respective classroom?	 (a) tasks that reflected a global perspective and supported an overall classroom climate of respect, (b) tasks connected to the students' home cultures (c) tasks related to curriculum and instruction 	7 4 11
6	In your class/school, how are academic expectations differentiated in regard to cultural identity or ability of the students?	(a) high expectations are not differentiated in regard to ethnicity (b) academic expectations are differentiated for students when necessitated by ability (special needs students) or language (ESL students)	16 14
7	What types of professional growth opportunities do you participate in that are pertinent to teaching a diverse student population?	(a) types of opportunities (b) types of topics	10 11
8	What activities are you engaged in within the local community?	(a) extracurricular school activities (b) volunteer work (c) social and cultural events	9 12 11

multicultural education research. These strands were then used to formulate the discussion questions that followed later. The analysis of the narrative, open-ended survey questions related to the following research questions in this study:

- 6. What instructional practices within the multicultural education framework do teachers in international schools use in their classrooms?
- 7. How does the cross-cultural effectiveness of teachers in international schools link to the instructional practices they use in their classrooms?

All twenty teachers in the international group answered all eight of the narrative, open-ended questions. Responses ranged from short phrases, such as, "I don't," or "Learning Spanish," to longer phrases, short sentences, multiple sentences, and informal paragraphs. The directions in the survey stipulated that the questions required only short answers and that they did not need to be excessively long. The responses received were what was expected for this section of the study.

Question 1

What types of activities, lessons, do you use to bridge, or connect, the students' cultures to classroom learning?

There were four general themes that arose out of the responses to question one.

These were (a) specific curriculum units and activities, (b) various content or subject matter connected to culture, (c) holidays, ceremonies, and festivals, and (d) connection to the students' homes and cultures.

Responses from seven teachers articulated the theme of specific curriculum units and activities (participants 2, 6, 7, 12, 14, 18, and 19). Biographical units were listed,

such as a naming unit (participant 2) that focused on how cultures name children and the meanings of student names, and biographical student scrapbooks (participant 14).

Cooperative learning and lessons specific to brain research were named as instructional strategies (participant 6). Also mentioned were specific curriculum units that studied different countries (participant 7), bringing guest speakers from different cultures into the classroom (participants 6 and 18), and completing journal exercises that involved students with current events from their own countries and other countries (participant 12).

Responses from 13 teachers articulated the theme of how content connected to culture in their classrooms (participants 1, 3, 5, 6, 7, 8, 10, 11, 13, 14, 16, 18, and 19).

Art, science, social studies, math, reading, vocabulary, writing, the International Baccalaureate program and English as a Second Language (ESL) curriculum were specific subject or content areas listed. It was suggested that artwork often times reflects student ethnicities (participant 1). It was stated that the teachers emphasize the global, international, and universal nature of science with students (participant 8), that science involves multiple individuals from many countries who have contributed to the field (participant 3), and that science history comes from various countries and cultures (participant 19). Math algorithms come from various countries (participant 19), and reading, vocabulary, and writing (participants 5 and 14) were indicated as connecting text to students' cultures and promoting reflections in noting likenesses and differences. In the area of the social sciences, economics and world geography were specific course areas listed (participants 13 and 16). In the area of economics, participant 13 indicated he/she incorporated the local conditions that students may experience on the way to

school and the interactions students may have in the local markets into the classroom lesson. In world geography, the teacher engaged students in lessons that focused on discussions about the students' own countries and heritages and the teacher kept "the world map open to reference" (participant 16) at most times. Generally, in the social science area, comments included that teachers created their own curriculum, that students and teachers often times together learned about culture, customs, and holidays, that students sometimes assisted the teacher in learning about local customs and events, and that the teacher connected the theme of community to students' own experiences (participants 7, 10, and 11).

Seven teachers alluded to the theme of holidays, ceremonies, and festivals as a way to connect student cultures to the classroom (participants 7, 10, 11, 14, 16, 18, and 19). Specific holidays mentioned were Festival of Nations school celebration (participant 18) and "sharing of homelands on United Nations Day" (participant 19) This theme was prevalent on its own, as well as was occasionally interwoven within other themes.

Five teachers responded with answers that articulated the theme of connecting the classroom to students' home and parents (participants 2, 4, 5, 10, and 17). The teachers' practices included using parents as guest speakers to "help explain, give examples of culture, customs, and traditions" (participant 10), drawing upon the child's cultural background when comparing differences and similarities between cultures (participant 17), using open-ended assignments to encourage students to "incorporate aspects of their own culture," and framing questions within a "context of students' cultures and experiences" (participant 4). One teacher indicated that allowing students to share their

own cultural view and experiences helps other students understand a global perspective of events (participant 2). In addition, one teacher reported encouraging students to talk about their cultures and bring examples from home that add to the classroom activities, supporting elaboration of class discussion to ensure students understand the reasons behind certain actions within a culture and why certain values are honored in a culture (participant 10).

Question 2

How do you personalize instructional delivery and your teaching to students' cultures, learning styles, and diversity in your classroom?

There were three general themes that arose out of the responses to question two.

These were (a) types of instructional strategies, (b) language strategies, and (c)

connecting to the students' home cultures.

Fifteen teachers articulated answers that created the theme types of instructional strategies (participants 1, 2, 5, 6, 7, 8, 9, 10, 12, 14, 16, 18, 19, and 20). Teachers indicated they work one-on-one with students (participants 5 and 7), arrange for teaching assistants to work with individual and/or small groups of students (participant 19), and/or schedule after school tutoring for students (participant 19). One-on-one conferencing was also reported (participant 13). Recognizing, being aware of, and attending to student learning styles were indicated often by the teachers as ways to personalize education in the classroom (participants 2, 8, 9, 10, 16, 17, and 20). Several teachers mentioned they use differentiated instruction in regard to learning styles, cultural backgrounds, and ability levels (participants 9, 10, 18, and 19). Learning modes was

addressed and teachers indicated they provide instruction, activities, and lessons that utilize visual, auditory, and kinesthetic modes (participants 1 and 14). Teachers also indicated they use cooperative learning (participant 7), verbal discourse (participant 7), and/or random pairing (participant 20), and that flexibility and variety of lessons are "key elements to meet individual needs" (participant 14). Adapting, modifying, reducing, and altering assignments were cited as being used to personalize instructional delivery (participants 2 and 19). Also referenced was prefacing lessons with objectives as a way to support the relevance of the lesson and relating new learning to previous knowledge (participant 6). One teacher stated that if he/she connects new learning to previous learning, the students "are more likely to be motivated in learning the newly presented materials" (participant 6).

A second theme revealed in this question was that of personalizing learning for students in an English as a Second Language (ESL) program. Seven teachers addressed this area (participants 1, 3, 10, 11, 14, 19, and 20). Extra time, peer assistance, and monitoring progress were cited as examples (participant 1). Collaboration between general education and ESL teachers was cited and included providing assistance, sharing lessons, and reducing or altering assignments in general education courses (participants 3 and 19). Specific strategies for ESL students included providing "a variety of opportunities to use written and spoken language throughout the school day" (participant 11), providing ESL students questions they can answer verbally (participant 20), allowing them to present their thinking verbally (participant 19), accommodating ESL students through cooperative learning groups (participant 19), and/or using dictionaries for

translating activities (participant 14). Lastly, one teacher indicated, "Language differences in particular demand a lot of practical discovery activities, visibly posted vocabulary, graphic aids and organizers, repetition of key ideas, restatement in simple language and questioning" (participant 10).

The third theme that arose in question two focused on home cultures and the connection to personalized instruction. Half of the twenty international teachers made statements that expressed this theme (participants 2, 4, 5, 10,11, 12, 14, 15, 16, and 18). Many of the statements revealed teacher efforts to learn about, or become personally aware of, students' cultures. Teachers stated they want to "understand the values of students and parents" (participant 16), try to "be sensitive to parents' views as well, although they may also be reluctant to express them" (participant 10), try to understand social behaviors (participant 10), "try to be conscious of my students' backgrounds, nationalities, and cultures" to individualize teaching (participant 4), or try to bring in cultural contributions through family histories (participant 2). One teacher noted, "I try to be aware of any 'teachable moment' related to the students' cultures where I can personalize the instructional delivery" (participant 5). Teachers also revealed that they try to connect curriculum lessons to students' home cultures as a way to personalize instruction (participants 4 and 5). Through their comments, it appeared that teachers sought out connections between cultures and lessons, encouraged students to discuss and share their home cultures and cultural perspectives, connected the home cultures and perspectives to lessons, and highlighted the cultures represented in the classroom. It was interesting to note that one teacher indicated that students' cultures didn't impact the

classroom, but learning styles were addressed in the activities and lessons. This teacher reported "we are more alike than different" (participant 17).

Question 3

How do you modify tests to provide fair assessments for diverse student populations in regard to race, ethnicity, language, and/or ability?

The answers expressed by the international teachers in regard to this question revealed 4 general themes. The themes focused on modifications in regard to (a) ESL and special education students, (b) student cultures, (c) vocabulary, and (d) general modification strategies.

In working with ESL and special education students, nine teachers indicated using various measures when modifying tests and providing fair assessments (participants 1, 2, 3, 8, 9, 10, 11, 19, and 20). Teachers indicated they differentiated according to ability (participants 2, 4, 8, and 9), were sensitive to language in the tests (participants 1, 3, 8, 9, 10, 11, 19, and 20), allowed students to receive assistance in a resource room or ESL classroom when taking a test (participants 1, 3, and 10), allowed students to use dictionaries (participant 10), worked with the special needs and ESL teachers to "develop fair, reasonable assessments" (participant 10) or modified items for language needs (participant19). One teacher indicated that specific modifications for language included "rewriting exams published by textbook companies, reading instructions, discussing answers before giving written questions – to set students up for success" and providing fewer modifications as language skills improve (participant 11). One teacher does "not

assign grades to any descriptor on the report card which relies on first language competence" (participant 19).

When modifying tests in regard to culture, four teachers made comments (participants 9, 13, 16, and 18). Teachers indicated they were sensitive to culture and modified tests accordingly (participant 9), that they ensured tests were culture free (participant 9), and/or that they proofread tests for cultural bias and would change them if needed (participant 18). In addition, one teacher indicated that students were encouraged to use examples from their own culture to help answer test items (participant 13).

Four teachers revealed ideas about modifying tests in regard to vocabulary, which is also linked to culture and language (participants 3, 4, 6, and 14). Teachers provided further assistance with vocabulary words when students took tests (participant 3), tried not to use vocabulary on assessments that "might be specific to a particular region or country and that not all students would understand" (participant 4), allowed students to use dictionaries when taking tests (participant 6), ensured that tests were provided using vocabulary students understood to enable testing of concepts rather than vocabulary (participant 14), and ensured that tests used "the common terms that students have been exposed to" (participant 14).

Generally, when modifying tests to provide fair assessment, teachers indicated they used a variety of general strategies. Twelve teachers had comments that articulated this theme (participant 2, 4, 5, 7, 10, 11, 12, 13, 15, 16, 18, and 19). Teachers reported they used alternative assessments (participant 5), a variety of assessments (participants 5 and 16), tests that contained multiple formats with different types of tasks, or

performance or project based assessments that allowed students to demonstrate learning and that utilized rubrics for grading (participants 15 and 18). Extra time was provided if needed to complete the test (participants 7, 12, and 13), sometimes questions were illustrated (participant 15), verbal answers were allowed (participant 10), group problem solving was used for assessments, (participant 19), and tests are allowed to be taken home for completion (participant 12). Other teachers reported that they adapted the level of text and types of questions to fit the students' abilities (participant 5), they used positive reinforcement (participant 7), provided background information or cultural explanations (participant 5), and sometimes read the directions and items of a test aloud for students (participants 5, 10, and 11). One teacher stated, "I generally give students a copy of the test beforehand. That way they can ask questions about it before actually taking it. If necessary, I modify the test based on their questions and ability to understand it" (participant 4).

Several responses to this question highlighted that in some instances teachers did not modify tests. One teacher reported providing no modifications at all (participant 17). Another teacher didn't modify, but made "a point of asking ESL or new kids what questions they have, or things they don't understand" and tries "to work with weak kids before the test" (participant 20). A third reported not having any special needs students in class and that students in the class were homogenous, thus modifying was not needed in this teacher's perspective (participant 7). Another teacher modified for ability and language, but not for race or ethnicity (participant 8).

Question 4

How do the materials used in your lessons/instruction reflect cultural non-bias?

What process is used in your school to ensure the use of culturally non-biased materials?

Four themes were revealed in the teacher responses to these two questions. The themes were about cultural bias in materials in regard to (a) textbooks, (b) how teachers augment materials, (c) specific programs or approaches, and (d) the process used to ensure non-biased materials are used.

Nine teachers made comments regarding the textbooks used in the schools and their classrooms (participants 2, 3, 7, 9, 10, 11, 15, 17, and 20). Most comments reflected concern about textbook bias. Even though, as one teacher noted, "our textbooks have become more globally written, to reflect stories, contributions, and pictures of all ethnic/racial groups" (participant 2), other teachers indicated a belief that most textbooks used in the international schools were based on American curriculum and therefore reflected an American bias (participants 3, 7, 10, 11, and 20). Then again, one teacher reported, "Due to the British background of the school, there is a bias towards British-based materials. With a growing population of non-British teachers, resources are being ordered that are more culturally balanced" (participant 15). Whatever the bias reflected in the textbooks, most teachers were aware of the possibility of bias and took steps in their classrooms to reduce the bias, wherever it came from. As stated, teachers appeared to be "sensitive to and aware of examples, words, geographic allusions, etc. that may cause confusion" (participant 3). Those were discussed in class to highlight differences and similarities and to bring clarity to student understanding.

One teacher indicated that the American textbooks included concepts that were not always relevant for students living and attending school outside the United States (participant 11). For example, as this researcher has experienced, in the area of measurement, American textbooks were based more in American units of measure, while the students in international schools lived in, and were from, countries that operate with the metric system.

Two teachers reported no bias in materials (participants 1 and 13): one perspective was stated that art is discussed from multiple perspectives and in relation to elements and principals of art and another perspective indicated that bias in the content area of economics comes not from the text, but "in the interpretation by students and teachers" (participant 13). A teacher in a Department of Defense Dependents' School (DoDDS) indicated, "textbooks chosen and used by DoDDS will always reflect cultural diversity. No one culture or race or whatever is depicted as superior nor inferior to others" (participant 17). On the other hand, a different teacher stated, "Instructional material is the least culturally sensitive area of the classroom and we remind ourselves who wrote the textbook and who it is for and then adapt it to our particular circumstances" (participant 16). Thus, the opinions regarding textbook materials were different from one another, but the teachers appeared to understand the issue of bias and dealt with it in their classrooms by screening, clarifying, adjusting, and modifying concepts presented in the materials. One teacher summed up the difficulty with textbooks in international schools quite nicely by expressing, "It is impossible to find

perfect materials, especially as an international setting (Third culture) is so different from a U.S.-based classroom in cultural and national make-up" (participant 10).

The augmentation of materials, the second theme of this question, was articulated through teacher comments that they added additional books, materials, problems and information to help address the American bias, provide richer cultural depth to lessons, and ensure a balanced cultural approach in the classroom. Six teachers reported that they added problems to counteract American bias in materials (participant 9), adapted or created their own materials "using either the host country, or the students' own backgrounds as a basis" (participant 10), brought materials with them to the teaching position to provide more cross-cultural experiences with literature (participant 7), ordered additional resources to promote a cultural balance (participant 15), or connected the materials to relevant local situations or events (participant 20). One teacher commented that "it is up to the individual teacher to adjust his/her teaching style and methods" to ensure materials are not culturally biased (participant 12).

The third theme expressed through teacher responses to this question focused on culturally non-biased materials and specific programs or approaches. The International Baccalaureate (IB) and Middle Year Program (MYP) were programs that "aim to promote intercultural understanding and tolerance of cultural differences" (participant 5) and that much effort has been put into the programs to eliminate cultural bias from the materials used in the program (participants 4 and 5). One teacher reported using a literature-based approach and brought in culturally rich literature materials that "expose

children to a wide variety of cultures and lifestyles in an informative and entertaining way" (participant 7).

In regard to the fourth theme of question four, the process that ensures non-biased materials are used, five teachers (participants 2, 3, 14, 17, and 19) indicated a process did exist to ensure the selection of non-biased materials and four teachers (participants 6, 7, 12, and 18) indicated there was no procedure used in their schools to ensure the use of non-biased materials. In three cases, committees or teams, were organized to review and select appropriate materials that were culturally non-biased, ensured the text supported a global view, and/or reflected cultural diversity. One teacher indicated that materials were "evaluated and selected by vertical curriculum teams and are in keeping with the mission of our school to provide education for international students who will meet the needs of a global society" (participant 19).

Question 5

How do you provide meaningful tasks for students that are responsive to race, culture, ethnicity, and language, in a culturally respective classroom?

In regard to providing a culturally respective classroom through meaningful tasks that are responsive to various aspects of diversity, there were three themes that rose out of the teachers' responses. They were (a) tasks that reflected a global perspective and supported an overall classroom climate of respect, (b) tasks connected to the students' home cultures, and (c) tasks related to curriculum and instruction.

There were seven teacher responses that articulated the first theme in this question: tasks that reflected a global perspective and supported an overall classroom

climate of respect (participants 2, 3, 10, 14, 16, 19, and 20). The idea of a global perspective in the classroom permeated many of the teacher's comments and is foundational in creating a classroom that is respectful of multiple facets of diversity. Reflected in the teachers' comments was the creation of a risk free classroom in which "students feel free to express their differences" (participant 16), that cultures and religions were respected through the daily classroom routines (participant 19), students welcomed new students to the classroom (participant 3), and there was a climate built upon trust between the teacher and students to enable both to learn from each other and help each other navigate unfamiliar cultures (participants 10 and 16). Other comments indicated that it was important for students to see themselves as part of the global whole (participant 2), that multiple groups are contributors to the global society (participant 2), and that "tasks are centered around concepts that are universal...on the things we all have in common" (participant 14). In a risk-free climate, teachers reported that diversity was celebrated as good fortune (participant 3), and that tasks promoted various cultural connections; i.e. "population density of India vs. Botswana" (participant 20).

The theme of connecting to home cultures was also articulated in this question.

Teachers reported that meaningful tasks encouraged students to "use aspects of their own backgrounds in their work" (participant 4), encouraged students to be proud of their culture and heritage (participant 17), and connected real world experiences to students' backgrounds (participant 16). Another example of meaningful tasks that created a respectful classroom involved allowing students to use research sources in their native

language and to use examples in their assignments that connected to students' native countries and cultures (participant 13).

Curriculum and instruction was the last theme revealed in this question. Eleven of the twenty teachers had one or more responses that connected meaningful tasks in a respectful classroom to the curriculum and instruction in the classroom (participants 2, 5, 6, 8, 9, 10, 11, 12, 15, 16, and 18). Instructional practices teachers reported related to this theme were role-playing, journal discussions, cooperative learning, projects, sharing, cultural events, and current events that were relevant to students' lives. One teacher remarked that a "broad range of texts and activities that are responsive to race, culture, ethnicity, and language" (participant 5) support meaningful tasks in a respective classroom and another teacher indicated that giving teachers flexibility in adapting and modifying resources and supplemental materials supported meaningful tasks (participant 5). Specific to curriculum subjects, science-based investigations (participants 8 and 15), the history of math from different parts of the world (participant 9), and the IB and MYP programs were listed as providing support for teachers to create meaningful tasks (participant 5).

Question 6

In your class/school, how are academic expectations differentiated in regard to cultural identity or ability of the students?

The overall response to this question by all teachers indicated that high expectations were not differentiated in regard to ethnicity; however, academic expectations were differentiated for students when necessitated by ability (special needs

students) or language (ESL students). There was a great deal of emphasis by the teachers, parents, and school personnel in general that high expectations were the norm, and that they were met through a variety of methods including individual progress, traditional grading system, the IB and MYP programs, and/or an emphasis on development of skills needed to compete in today's global and ever changing world. When differentiation occurred teachers reported it as extra time, preferential seating, reduced content, adjusted grading, modified curriculum, and focus on skills rather than grades for special needs and/or ESL students. Assignments and evaluation of assessments were also differentiated as needed.

Question 7

What types of professional growth opportunities do you participate in that are pertinent to teaching a diverse student population?

Teacher responses revealed two general themes when answering this question.

They articulated various types of opportunities and various topics involved in professional development opportunities pertaining to teaching a diverse student population. Teachers listed conferences, workshops, seminars, and university courses and degree programs as types of opportunities available to them in international locations. Multiple topics focused on specific subject matter specific to diversity: adapting lessons for global classrooms, methods and courses specific to diverse students and classrooms, connecting lessons to students' cultural backgrounds, courses to learn the local language, and training in working with ESL students. General topics included learning styles, gender issues, and general teaching techniques.

Other opportunities for professional growth were identified as discussions of issues with colleagues and educational journals and books. One teacher indicated that the school had offered "very little cultural sensitivity training" (participant 6) and four other teachers indicated there were no opportunities available to them for professional growth (participants 7, 12, 17, and 20). One teacher suggested that communication with students and parents provided the best opportunity for professional growth and stated, "there is not one workshop or conference that provides better professional growth than your own students" (participant 18).

Question 8

What activities are you engaged in within the local community?

Three themes emerged from the teacher responses regarding the activities they are engaged in within the local community. These are (a) extracurricular school activities, (b) volunteer work, and (c) social and cultural events. Eight teachers indicated they were involved in school activities through membership on faculty committees, coaching school sports, and advising various types of student clubs (participants 2, 3, 8, 11, 15, 16, 17, and 20). These types of involvement included membership on school improvement and advisory committees, sponsorship of new families and teachers, serving as department chair, organizing school functions and productions, and advising student clubs, such as drama activities, science clubs, National Honor Society, etc.

Twelve teachers, more than half of the international teachers, reported participation in volunteer work in the local community (participants 2, 5, 6, 7, 8, 10, 12, 14, 15, 18, 19, and 20). This covered involvement in outreach/service projects, fund

raising events, and general volunteer work. They volunteered at the local USO, military post office, and as a scout leader. They walked and ran in various events to raise money for cancer research, environmental awareness, and AIDS research. They supported outreach programs for the local underprivileged populations, completed service learning projects with students, supported refugees, and sponsored children in schools with economically disadvantaged students.

More than half the international teachers, eleven of the twenty, reported participation in social and cultural activities in the local community (participants 1, 4, 5, 6, 7, 8, 9, 11, 17, 18, and 20). This ranged from taking pottery courses with a local artist, to traveling, to attending or visiting cultural exhibits, art galleries, museums, concerts, theater, dance, and orchestral events. Other activities in the community included dining in local establishments, attending church, participating in sports with expatriates, socializing with friends from the local community, attending local weddings and other celebrations, and shopping on the local economy. These activities contributed positively to teachers' learning about the local culture and provided avenues for teachers to engage in social justice actions.

Discussion Question Results

To further answer the qualitative questions of the study and probe deeper into the instructional practices of the international teachers, themes from the narrative, openended questions were used to create four in-depth discussion questions. Responses to the four discussion questions (Appendix P) were coded and analyzed for meaningful insights that further defined the instructional practices of the international teachers from within

the multicultural education research framework and linked to the cross-cultural effectiveness of the international teachers. The analysis of the discussion questions related to the following research questions of this study:

- 6. What instructional practices within the multicultural education framework do teachers in international schools use in their classrooms?
- 7. How does the cross-cultural effectiveness of teachers in international schools link to the instructional practices they use in their classrooms?

The discussion questions were presented to the international teachers through email in an asynchronous electronic discussion format. Participants were asked to answer each of the four questions and were also encouraged to respond to the answers provided by other participants. The questions were posted individually every three to four days and were left posted for approximately four weeks. Instructions indicated that participants were to use the 'reply all' command in the email format, which allowed the teachers to read and respond to the answers of the other participants.

Discussion question one was based on responses from narrative, open-ended question eight and probed deeper into the activities teachers were involved in outside the classroom in the school and local community and the teacher perceptions as to why these activities are important to being an effective teacher. Discussion question two was based on the responses from narrative, open-ended questions two, three, and five and more deeply explored teacher perceptions on how specific instructional strategies support learning in a diverse classroom. Discussion question three was derived from a blend of responses from narrative, open-ended questions one, two, three, five, and eight and

investigated deeper teacher beliefs in the importance of the connections between the classroom and students' home cultures in supporting the learning process. Discussion question four was based on multiple statements expressed in various ways across the narrative, open-ended questions that highlighted the sense of personal beliefs, passions, and commitment the teachers expressed regarding educating students from various cultures in international schools. Discussion question four probed the personal values the teachers brought to the international classroom and the beliefs in how those personal values affected student learning.

Not all of the answers to the narrative, open-ended questions provided a basis for further exploration. For the most part, the answers to narrative, open-ended questions four (materials used in the classroom), six (differentiating academic expectations), and seven (professional growth opportunities) were fairly straightforward answers and did not appear to lead to further exploration in this study. It is suggested, however, that in future research these topics might be explored through a revision of the narrative, open-ended questions four, six, and eight so as to elicit more in-depth answers for further investigation.

In regard to the four discussion questions, eight participants (n = 20) answered all four (100%) of the discussions questions. One participant answered three of the four questions (75%), six participants answered two of the four questions (50%), three participants answered one of the four questions (25%), and two participants did not answer any of the four discussion questions. Sixteen of the 20 participants (80%) provided an answer for discussion question one, twelve participants (60%) provided an

answer for question two, and eleven participants (55%) responded to discussion question three and to discussion question four.

Table 17 provides a listing of the four discussion questions with the themes that arose out of the responses related to each question and the number of teachers whose answers fell within that theme.

Question 1

Within the multicultural educational models in research literature, one of the characteristics of effective teachers is their ability to negotiate other cultures. In addition, the CCAI inventory that you completed gauges a participant's ability to empathize with those who are different from oneself. Involvement with, and understanding of, students' home backgrounds is part of this process, which implies understanding and negotiating community/local cultures. In the survey/narrative question responses, nearly all the participants indicated they are involved with 1) sponsorship of school activities (clubs, sports, drama, PTA, committees, etc.), 2) volunteer work in the local community (fundraising, service work, outreach work, etc.) and/or 3) cultural activities (art, music, travel, local events, etc.). Why are these types of involvement important to you as an international teacher? How does it contribute to your effectiveness as a teacher in a diverse classroom?

There were four interwoven themes that arose out of the respondents' answers regarding why school activities, volunteer work, and cultural activities were important to them and how they contributed to their teaching effectiveness. The teachers described (a) how their involvements impacted their learning of cultures, (b) how it helped them

Table 17

Themes Revealed in the Discussion Questions

	Discussion Question		Themes From Responses	# of Teacher Responses
1	[Multicultural education and CCAI]Involvement with, and understanding	(a)	how their involvements impacted their learning of cultures	12
	of, students' home backgrounds is part of this process, which implies	(b)	how it helped them connect to students and families	8
	understanding and negotiating community/local cultures. In the survey/narrative question responses, nearly all participants indicated they	(c)	how it helped their students and themselves connect to the real world	7
	are involved with [school activities, volunteer work, and cultural activities]. Why are these types of involvement important to you as an international teacher? How does it contribute to your effectiveness as a teacher in a diverse classroom?	(d)	how the involvement contributed to the development of their own personal characteristics and attributes	6
2	[Survey/narrative question responses and effective teaching strategies from	(a)	cooperative learning	9
	research]In your classroom, which of these [instructional strategies] do	(b)	identifying similarities and differences	5
l	you find to be the most powerful effective teaching strategy that supports	(c)	additional instructional strategies	8
ļ	cross-cultural experiential, authentic, and interactive learning? Why?	` ′		1
3	In the research within the multicultural education framework, effective teachers are seen as those who seek to teach the students the culture of the	(a)	values students' cultures and expands student thinking about other cultures	6
	school while maintaining the students' sense of their own cultural identity and pride. In the survey/narrative question your responses indicated a	(b)	helps teachers better understand students and supports positive interactions between teacher and student	3
	variety of ways of using students' home cultures in the instruction of the classroom. Explain your beliefs in how important this practice is in helping students construct knowledge and learn.	(c)	a way to connect new learning to known cultural concepts and to a sense of community	4
4	Research indicates that one of the most powerful factors in student	(a)	personal attributes	8
	achievement is what the teacher believes about students and what the	(b)	beliefs and values in regard to students	6
	teacher expects of students. In addition, the teacher's perspective is shaped by their own upbringing, their life experiences, their beliefs and values, and	(c)	beliefs and values in regard to students' families, home cultures, and local culture	3
	their sense of efficacy (the belief they can change a situation). What personal values do you hold regarding teaching in an international and diverse setting? What do you believe about how you affect student learning?	(d)	beliefs and values in regard to the classroom and teaching	5

connect to students and families, (c) how it helped their students and themselves connect to the real world and (d) how the involvement contributed to the development of their own personal characteristics and attributes.

The teachers indicated that being involved in school activities and the local culture actually helped them learn more about the culture. By actively participating in school functions, teachers new to the local culture were able to socialize with those who knew the culture and could learn from the interaction (participant 6). In addition, the school was a familiar environment for the teacher and school functions outside the school day mixed the familiar (school environment) with the new (local culture) and this mix contributed to the teacher's learning (participant 6) in a non-threatening way.

It was reported that involvement in the local culture contributed greatly to the teacher's learning by aiding the teacher to expand existing cultural knowledge (participants 4, 6, 7, and 20), build new constructs (participants 7 and 9), and assist the teacher to learn to be sensitive to the local culture and learn the nuances of the culture so as to not offend either socially or politically by inappropriate behavior (participants 7 and 9). Thus, involvement in activities helped teachers build assets to negotiate various cultures and cultural contexts.

Multiple comments by participants highlighted how the teachers brought their learning into the classroom. Teachers indicated they shared their traveling experiences with students (participants 1 and 7), they participated in school-sponsored field trips with students (participants 1 and 18), and they encouraged families to participate in cultural experiences (participant 17). One teacher revealed that students taught her/him a great

deal about the culture when they were involved in school and local activities together (participant 7). Others indicated that they brought their learning into the classroom to share with students (participant 4) to "help students grow, develop, learn about themselves and others" (participant 5) and another teacher indicated that school organized cultural events helped students learn, understand, and appreciate cultures other than their own (participant 3). In addition, participant 10 stated that a "teacher who has spent time out in the community can help explain unfamiliar customs and habits in a way that makes them seem less strange" to students in the classroom who are from cultures different than the local culture.

It was evident that teachers believed involvement in activities was deeply rewarding and added a great deal of richness to their own lives and to their teaching (participants 10 and 14). As participant six stated, "You cannot learn this from a book." They continued to relate that even though experiences with the culture can sometimes be frustrating, the "positive and delightful experiences that take place every day" led to deeper learning and understanding of the culture (participant 10). This teacher continued with that he/she was "constantly learning that tolerance, openness, and knowledge go a long way to creating trust and understanding." Through involvement in school activities, volunteer work, and cultural activities, teachers accepted, tolerated and adjusted to differences while examining their own beliefs and behaviors. Ultimately, through their school and cultural experiences and activities, these teachers changed their personal constructs and acquired new constructs that influenced instructional practices (Connelly

& Clandinin, 1988; Goodson, 1990; Jarchow, McKay, Powell, & Quinn, 1996; Kelly, 1963).

The second theme that emerged out of question one focused on the teacherstudent and teacher-family relationship building that occurs when teachers were involved in school, volunteer, and local cultural activities. School-sponsored activities allowed teachers to spend time with students outside the school day developing relationships with students that were not necessarily academic based. Interacting with students in activities beyond the school day and interacting with the local culture helped teachers know their students and understand students' home cultures which in turn enabled teachers to make connections with students, more effectively relate to students in the classroom, and build better class rapport (participants 11, 18, and 20). Teacher involvement in various activities helped create a sense of community for students who were living in a foreign culture and encourages student participation in the local culture, which may have been very different than the students' home cultures (participant 2). Activity involvement by students, and with teachers, helped students accept new ideas, adjust to new cultures, and grow "in diversity, global knowledge, and becoming world citizens" (participant 2). As one teacher succinctly stated it, involvement in activities "helps to connect me to my students and it helps to connect us all to the global family" (participant 15).

Teachers also reported that involvement in activities provided an avenue for interacting with parents in a venue outside the daily classroom. This encouraged invitation to new opportunities and experiences with parents (participant 6), enhanced communication with parents (participant 18), and created bonds with parents that were

less stressful than can sometimes occur within the classroom (participant 18). Through the bonds that were created, teachers related that they invited and encouraged parents to work in the classroom as volunteers or as guest speakers to share their cultures and geographic backgrounds with students (participant 17). Again, this created enhanced communication and encouraged parents to become involved with their child's education.

The third theme that emerged out of the answers to discussion question one focused on how active involvement helped teachers and students connect to the real world through volunteer work and defined and supported underlying characteristics of the teachers' personalities within this connection. Multiple teachers (participants 4, 5, 8, 11, and 15) indicated that the involvement in volunteer work and community service was a commitment of theirs no matter the location – stateside or international, and that the participation was similar between the locations. The major distinction noted by participant eight was the difference in diversity, in depth of experience, and in the variety of locales in regard to the volunteer work. The teacher noted that community service in international locations compared to stateside locations were "different experiences yet the same manifestation of who we were and are." Another teacher noted that "international teachers seem to be more involved and committed to extracurricular activities and community service," but that it "may reflect the type of people who pursue an international teaching career" (participant 15). The same teacher also indicated that there were "more visible opportunities" and "more available resources" for community service overseas.

Participants 3, 15, and 20 noted the connection between their involvement in community service and the importance to student involvement in community service. It was perceived that when teachers were involved, it helped get students involved, and once students were involved with community work, they often continued as adults. Participant 20 stated that volunteer work developed career awareness for students for those who may later choose service oriented/outreach careers. The same teacher noted that their new school mission statement "stresses that all students will become motivated global contributors." Teachers also indicated that volunteer work provided experiences for privileged children to understand the global contexts in which they live and to offer opportunities for them to help those less fortunate (participant 3). One teacher reflected that community service taught "students that there is more to life than a pampered existence" (participant 20).

The fourth and last theme revealed in question one revolved around the impact of school and community involvement on personal qualities. Teachers indicated that being actively involved in school and community strengthened their personal beliefs and values through self-reflection, helped them learn, grow and know themselves better, and helped them reflect a more positive attitude in their teaching (participants 5, 8, 10, and 11). Teachers also indicated that experiences in school and community activities assisted them in understanding and negotiating local cultures and helped them more easily embrace diversity (participants 5 and 14). One teacher revealed personal growth away from ethnocentrism as seen in the statement,

The difference, for me, lies in the personal challenge of understanding and negotiating cultures – from the language, customs, the beliefs to the everyday

sounds and sights on the streets. I find this process to be incredibly stimulating and enriching....the daily opportunity that living overseas provides to grow as a person, to learn about other cultures, to realize that my way is not the only way (participant 5).

Question 2

In the survey/narrative question responses, nearly all participants indicated they use a wide variety of instructional strategies from one-on-one to the use of graphic organizers and multiple others. Research highlights the most effective instructional strategies as 1) identifying similarities and differences, 2) summarizing and note taking, 3) reinforcing effort and providing recognition, 4) homework and practice, 5) nonlinguistic representations, 6) cooperative learning, 7) setting objectives and providing feedback, 8) generating and testing hypothesis and 9) questions, cues and advance organizers. In your classroom, which of these do you find to be the most powerful effective teaching strategy that supports cross-cultural experiential, authentic, and interactive learning? Why?

In regard to the use of specific instructional strategies, there was a general consensus that all of the listed strategies in the question were used to some degree and that often the use of a specific strategy depended on the desired outcome and learning that was targeted. Table 18 indicates the number of times each strategy was commented on by the teachers.

Teachers saw cooperative learning as a strong strategy for group processing, building group dynamics, and supporting ESL students in the learning process. As participant 14 noted, when ESL students are grouped in cooperative work groups, they

"help each other understand instructions, study new words, practice pronunciations, write stories, etc." Often times, for ESL students, English, which was the language of

Table 18

Teacher Perceptions of the Most Powerful Instructional Strategy

Instructional Strategy	Number of Times Commented On		
Cooperative Learning	9		
Identifying Similarities and Differences	5		
Reinforcing Effort and Providing Recognition	n 3		
Homework and Practice	3		
Setting Objectives and Providing Feedback	3		
Generating and Testing Hypothesis	3		
Questions, Cues, and Advance Organizers	3		
Summarizing and Note Taking	2		
Nonlinguistic Representations	2		

classroom instruction, was one commonality for students who speak various native languages. In cooperative learning, the students needed to speak to each other and needed to speak English to help each other. In this sense, cooperative learning "is the open door that allows [students] to build that common vocabulary, scaffold meaning, and answer questions that are important to their learning" (participant 14). In another sense, it was also helpful for ESL students to work with fluent speakers of English as a way to

learn and for peers to help each other (participants 1, 14, and 18). One teacher (participant 20) related that he/she organized cooperative learning groups based on ability levels rather than cultural differences and placed translators in specific groups as needed to support language needs.

Teachers also highlighted benefits to students of using cooperative learning groups. These perceptions included that cooperative learning allowed students to be more responsible and take ownership for their own learning (participant 4), that cooperative learning provided social learning opportunities for students (participant 11), and that cooperative learning lowered anxiety levels for students: "students don't seem to be as wary of asking for further clarification from a peer" (participant 20).

The second instructional strategy that teachers highlighted as a powerful instructional strategy to maximize learning in a multicultural classroom was the strategy of identifying similarities and differences. This strategy was reported as an excellent way for students to clarify differences and similarities between cultures and to gain understanding of the elements being compared. In addition, it was reported as an effective way for "students to put their life in perspective with the rest of the world" and "differences between cultures is easy to see" (participant 11). One teacher stated that for higher level classes, identifying similarities and differences served as an excellent writing prompt and helped students see how a particular problem fits into a bigger picture (participant 20).

In regard to the remaining strategies, various comments were made. The strategy of summarizing and note taking was cited an excellent tool for assessing comprehension

in the area of reading (participant 18), homework and practice was indicated as a great strategy for getting parents actively involved in a child's education which created a "win-win situation" (participant 17), and setting objectives and providing feedback was perceived to help students understand what they needed to know and do and helped them focus (participant 10). Generating and testing hypotheses was perceived as a way to help students learn better by developing and proving their own conjectures "rather than simply mimicking exercises out of the book" (participant 20) and as an effective tool for formative and summative assessment (participant 18). Questions, cues, and advance organizers were thought of as a type of scaffolding process important in the learning of a language, thus was effective for ESL learners.

Questions and cues...also support student interaction as the students have a base to work from and a framework with which to organize their communication. As a result, students are well supported in the language learning process and can effectively build their skills and confidence as they increase their cross-cultural understanding (participant 5).

Lastly, reinforcing effort and providing recognition was perceived as an instructional strategy that helped create a safe and supportive environment, motivated students, and helped them feel valued, which is a necessary precursor to academic learning (participant 10).

Ouestion 3

In the research within the multicultural education framework, effective teachers are seen as those who seek to teach the students the culture of the school while maintaining the students' sense of their own cultural identity and pride. In the survey/narrative question your responses indicated a variety of ways of using students'

home cultures in the instruction of the classroom. Explain your beliefs in how important this practice is in helping students construct knowledge and learn.

This question seemed more of a challenge for teachers and their answers were quite varied and reflected much personal identification to the importance of using home cultures in classroom instruction to help students construct knowledge. There were three brief themes highlighted from the responses to this question.

Teachers perceived that using home cultures in the classroom valued students' cultures and expanded student thinking about other cultures. By valuing student cultures in the classroom, students felt accepted, comfortable and secure, appreciated for their diversity, and also felt a sense of belonging (participants 5 and 14). When student cultures were valued, students felt free to share their home cultures and experiences. This helped student keep in touch with their home values and customs while learning about other cultures and added value to their own lives (participants 10 and 12). In addition, connecting cultures to curriculum helped students learn various perspectives and taught tolerance of others' views (participants 4 and 17).

Teachers noted that using home cultures in the classroom helped the teacher better understand students and supported positive interactions between teacher and student (participant 20). At times, the teacher learned more from the student and was able to make connections between the student's culture and classroom learning (participant 7). This promoted teacher awareness of cultural differences and supported allowing student choice in demonstrating understanding; students chose "in which manner they feel most

comfortable demonstrating their understanding" (participant 20). This ensured tolerance of cultural nuances in the classroom. For example, from participant 20:

My time in the mid East taught me that most Arab cultures are decidedly more vocal than written. I notice that my Arab students are much less successful having to perform a task in writing, than they are in explaining the same concept orally. On the other hand, naturally quiet cultures, such as my Japanese students, are just the opposite. This year, my students with the least English are all Japanese, and with them I notice that they are able to express very complex ideas in writing, but struggle (or are reluctant) to verbalize them.

The third theme in question 3 focused on using home cultures in the classroom as a way to connect new learning to students' known cultural concepts and to a sense of community. "Students construct knowledge on the foundation of what they already know or believe to be true" (participant 10) and, as participant 4 stated, "By helping students maintain their own culture within the context of a course they are able to more easily connect with the curriculum. Once they make the connection, they 'construct knowledge and learn." When students shared culturally specific concepts, the teacher was also able to provide activities that made sense to the student and helped the student learn (participant 10). In addition, new knowledge built on the student's cultural foundation provided the student not only new conceptual learnings, but also a greater understanding of the local community (participant 13). Participant 1 related that he/she allowed students to make choices in creating an art portfolio and they often chose artists from their own home cultures. Students connected to the home culture artist, used the artist's pictures in their student portfolio, wrote commentaries on the artist's work, and created their own work in the artist's style. This provided a level of comfort for the student, and maintained a known element in their learning while constructing new knowledge.

Question 4

Research indicates that one of the most powerful factors in student achievement is what the teacher believes about students and what the teacher expects of students. In addition, the teacher's perspective is shaped by their own upbringing, their life experiences, their beliefs and values, and their sense of efficacy (the belief they can change a situation). What personal values do you hold regarding teaching in an international and diverse setting? What do you believe about how you affect student learning?

The teachers were prolific in their response to this question. It was obvious that they had significant interest and passion in their personal values and beliefs about themselves and about how they affected student learning. Four themes were highlighted in the responses. Within the first theme, teachers identified a long list of attributes that described their personal values: enthusiasm, curiosity, honesty, integrity, openness, sense of humor, dedication, tolerance, self-confidence, perseverance, persistence, hard work, punctuality, being prepared, dependable, willingness to make mistakes, look at things with a fresh or new attitude, be open to the wonders of the universe, lighten up, being human, believing in education, taking pride in own cultural values, caring about others, respecting others, think before speaking, being creative, and having a giving spirit.

Participant 4 identified the joy of learning as a belief; joy of learning about self, about others, and about a subject, and remaining a life-long learner. One teacher very eloquently described personal values and beliefs that were very much from a social justice framework (participant 10). This teacher believed that all should have a life free

from pain, threat and suffering, that it is the responsibility of those more fortunate to help others, that one should be aware of environmental issues, that the earth should be protected, that all living creatures are worthy or respect, and that one should be tolerant of others while seeking understanding of those who are different from oneself.

A second theme embedded in question four were the beliefs and values the teachers held in regard to students. They indicated belief in the inherent and intrinsic value of each child (participants 8 and 13) and that all students have talents (participant 13), that it is important to understand students (participant 7), and to have expectations of them (participant 20) while motivating and encouraging each student to reach their full potential as successful contributors to their families and society and lifelong learners (participants 13 and 20). In order to do this, teachers indicated they need to listen to students (participant 7), let students know they care about them (participant 7), provide students with challenges and choices (participant 20), and encourage students to feel positive about themselves so they can approach learning confidently and can master it successfully (participant 10). Lastly, teachers indicated the need to help students "to see the best in each other and become more aware of the value of another's thoughts" (participant 10).

The third theme in question four were the beliefs and values the teachers held in regard to students' families, home cultures, and the local culture. It was indicated that teachers definitely need to know the world from which their students come, the activities in which they are engaged, and must show interest in the cultural values different from their own (participants 7 and 12). In addition, a teacher needs to believe in families and

the importance of connecting to the students' parents and should encourage parents to experience, accept, and embrace diverse cultures (participant 17).

The last theme in question four described the teachers' beliefs and values in relation to the classroom and teaching. The classroom should be risk-free, safe, and comfortable to afford students an environment in which they are free to learn and grow (participants 7 and 13) and the teacher should model the beliefs and values to which he/she aspires (participant 5). The classroom should be a place where "differences are explored, discussed, and celebrated," (participant 10) so students can learn to negotiate multiple cultures and contribute to society.

Summary

Chapter 4 reported the demographic information regarding the participants and the data results from the (a) Cross-Cultural Adaptability Inventory (CCAI) completed by 20 Iowa teachers and 20 international teachers, (b) the narrative, open-ended survey questions completed by the 20 international teachers, and (c) the follow-up discussion questions also completed by the 20 international teachers.

The CCAI compared cross-cultural effectiveness between the two teacher groups. The international teachers had higher means in the overall score as well as in each of the four dimensions of emotional resilience, flexibility/openness, perceptual acuity, and personal autonomy. However, the only dimension score that yielded a statistically significant difference between the two groups was in the emotional resilience dimension.

The narrative, open-ended questions yielded multiple themes in each question that connected to the multicultural education framework. These themes were used to design

the discussion questions. Specific themes also arose out of the discussion questions that further defined and explained the instructional practices of the international teachers and connected to the multicultural education framework.

The data from the three sources was used to explore the seven research questions of the study, which will be discussed in Chapter 5.

CHAPTER 5

SUMMARY, CONCLUSIONS AND REFLECTIONS, IMPLICATIONS

Introduction

Chapter five is a synthesis of the data results, connects the results to research literature, and provides suggestions for further study. The Summary section provides an overview of the study's purpose and the methodology that was used to complete the study. The Conclusions and Reflections section provides the significant results of the quantitative and qualitative data analysis, connects the results of the study with the literature review, and further blends these connections with personal reflections. The Implications section provides suggestions for further considerations beyond this study.

Summary of the Study

The purpose of this exploratory study was to investigate the differences in cross cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices used by international teachers to meet the needs of a diverse student population. Information and understandings from this study will help further define and extend the knowledge regarding effective teaching practices in multicultural classrooms. The direction of the study was guided by the following research questions:

1. To what extent is there a difference in overall cross cultural effectiveness (Kelley & Meyers, 1995) between teachers in international schools and teachers in Iowa schools?

- 2. To what extent is there a difference in emotional resilience (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 3. To what extent is there a difference in flexibility/openness (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 4. To what extent is there a difference in perceptual acuity (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 5. To what extent is there a difference in personal autonomy (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?
- 6. What instructional practices, within the multicultural education framework, do teachers in international schools use in their classrooms?
- 7. How does the cross-cultural effectiveness of teachers in international schools link to the instructional practices they use in their classrooms?

The differences between the international teachers and the Iowa teachers in regard to cross-cultural effectiveness were hypothesized as follows:

1. The teachers in international schools will be more cross culturally effective overall (Kelley & Meyers, 1995) than the teachers in Iowa schools.

- 2. The teachers in international schools will be more cross culturally effective in regard to emotional resilience (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 3. The teachers in international schools will be more cross culturally effective in regard to flexibility and openness (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 4. The teachers in international schools will be more cross culturally effective in regard to perceptual acuity (Kelley & Meyers, 1995) than the teachers in Iowa schools.
- 5. The teachers in international schools will be more cross cultural effective in regard to personal autonomy (Kelley & Meyers, 1995) than the teachers in Iowa schools.

Results of the study were collected through four data points: demographic information regarding the participants, Cross-Cultural Adaptability Inventory (CCAI) to examine the differences between the two sample groups of teachers, and narrative, openended survey questions and discussion questions to explore the instructional practices of the international teachers. The international teachers completed a survey that consisted of demographic questions, the 50-item CCAI, and narrative, open-ended questions. The answers to the narrative, open-ended survey questions were coded and themes identified. This information was then used to design the follow-up discussion questions that yielded deeper insights into international teachers' instructional practices. The 20 international participants were located in 13 different locations and were known to the researcher. Participation was elicited via email messaging and all data results were gathered via electronic means through an online survey tool and email messaging.

The Iowa teachers completed only the demographic questions and the 50-item CCAI portion of the survey, which reflected the intent of the study design. The 20 Iowa participants were matched to the 20 international teachers on three demographic criteria and were identified with the assistance of district superintendents and school principals. Participation was elicited via email messaging and all data results were gathered via electronic means through an online survey tool.

Data analyses included (a) investigating the demographic characteristics of the participants, (b) descriptive and inferential statistical analyses regarding the CCAI results, and (c) applying analytic induction to find common themes and patterns from the narrative, open-ended question results and the discussion question results relating back to instructional practices within the multicultural education research framework.

Conclusions and Reflections

Research Question 1

To what extent is there a difference in overall cross-cultural effectiveness (Kelley & Meyers, 1995) between teachers in international schools and teachers in Iowa schools?

It was hypothesized that the teachers in international schools would be more cross culturally effective overall (Kelley & Meyers, 1995) than the teachers in Iowa schools. Results indicated that the international group of teachers, when compared to the Iowa teachers, were more cross-culturally effective, but the difference was only marginally significant (Vogt, 1999). This indicates that the international teachers have somewhat more of the "most critical skills needed for interacting effectively with people of other cultures" (Kelley & Meyers, 1995, p.21). In addition, it can be stated that generally the

international teachers, as a group, would exhibit more of "a sense of self, emotional equilibrium, mental flexibility, openness to differences, and communication skills" (Kelley & Meyers, 1995, p. 21).

Research Question 2

To what extent is there a difference in emotional resilience (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?

It was hypothesized that the teachers in international schools would be more cross culturally effective in regard to emotional resilience (Kelley & Meyers, 1995) than the teachers in Iowa schools. This hypothesis was supported with results that indicated a statistically significant difference (Cohen, 1988). This CCAI dimension (subscale) was designed to measure the extent to which individuals, in this case teachers, can rebound from negative and unpleasant feelings and react positively to new experiences (Kelley & Meyers, 1995). Characteristics of this dimension indicate that international teachers have more of an ability to deal with stressful feelings and ambiguity in a confident and constructive manner. Generally, these types of people would like new experiences, can regulate emotions, maintain emotional equilibrium in a new or changing environment, and effectively deal with difficulties that are a normal course of a cross-cultural experience, have a sense of humor, positive self-regard, and tend to take risks. These teachers would most likely also exhibit a positive attitude and would not be overly concerned with making mistakes (Church, 1982; Guthrie, 1975; Kelley & Meyers, 1995).

Research Question 3

To what extent is there a difference in flexibility/openness (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?

It was hypothesized that the teachers in international schools would be more cross culturally effective in regard to flexibility and openness (Kelley & Meyers, 1995) than the teachers in Iowa schools. Even though there was a difference, results indicated that the difference was not statistically significant (Cohen, 1988). This indicates that the international teachers have a bit more flexibility and openness, than the Iowa teachers, but the results are not strong enough to make a general conclusion that can be applied to a larger population of teachers. The flexibility and openness dimension is characterized in this inventory by (a) having an openness toward, liking, having an interest in, and desiring to learn from unfamiliar people and ideas, (b) tolerating, being nonjudgmental, and understanding toward those different from oneself, and (c) being flexible with regard to experiences (Kelley & Meyers, 1995). Teachers who are open and flexible "enjoy interacting with people who think differently from themselves (Kelley & Meyers, 1995, p. 16), are nonjudgmental, are tolerant of others, tend to think creatively and are inquisitive, and can deal constructively with differences by building bridges between oneself and others (Kelley & Meyers, 1995). Basically, teachers scoring higher in this dimension have an overall positive attitude toward the unfamiliar.

Research Question 4

To what extent is there a difference in perceptual acuity (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?

It was hypothesized that the teachers in international schools would be more cross culturally effective in regard to perceptual acuity (Kelley & Meyers, 1995) than the teachers in Iowa schools. Results indicated a difference, but not a statistically significant difference (Cohen, 1988). This indicates that the international teachers are somewhat more perceptually acute than the Iowa teachers, but again, as with the previous research question, the results are not strong enough to make a general conclusion that can be applied to a larger population of teachers. Characteristics of perceptual acuity, as measured in this inventory, include (a) being attentive to and accurately interpreting verbal and nonverbal cues, behavioral cues within the context of communication and interpersonal relations, (b) being a highly accurate communicator, and (c) perceiving and being sensitive to the feelings of others and knowing the effect they have on others (Kelley & Meyers, 1995).

Research Question 5

To what extent is there a difference in personal autonomy (Kelley & Meyers, 1995) in regard to cross-cultural effectiveness between teachers in international schools and teachers in Iowa schools?

It was hypothesized that the teachers in international schools would be more cross cultural effective in regard to personal autonomy (Kelley & Meyers, 1995) than the

teachers in Iowa schools. Even though there was a difference, results indicated that the difference was not statistically significant (Cohen, 1988). This indicates that the international teachers are somewhat more personally autonomous than the Iowa teachers, but again, as with the previous two research questions, the results are not strong enough to make a general conclusion that can be applied to a larger population of teachers. Characteristics of personal autonomy, as measured in this inventory, include (a) not being dependent on the environment to maintain their self-identity, but have a strong internal sense of self and are able to maintain it independent of the environment (not needing external reinforcement from their native culture to maintain self), (b) have a strong sense of who they are, have clear personal values, and have respect for themselves and others, (c) are "internally grounded and can tolerate external fluctuation and change" (Kelley & Meyers, 1995, p. 18), (d) feel at home in most any environment, (e) are self-directed and enjoy making their own decisions, and (f) do not lose their own self-identity when in a new culture. This dimension is associated with the concept of empowerment and the confidence to act from one's own personal values and beliefs when in new or unfamiliar cultures (Kelley & Meyers, 1995).

To summarize, from the statistical outcomes of the CCAI results, it can be suggested that the international teachers would be more likely to do well cross-culturally overall, and can cope better than the Iowa teachers with stress and ambiguity, can better accept and rebound from imperfections and mistakes, are willing to try new things and experiences, and interact better with people in new or unfamiliar situations. However,

the international and Iowa teachers reflected similar skills in being flexible and open, in perceptual acuity, and in personal autonomy.

Research Question 6

What instructional practices, within the multicultural education framework, do teachers in international schools use in their classrooms?

The narrative, open-ended questions were derived from a compilation of effective teacher descriptors in the multicultural education research and the discussion questions were derived from the narrative, open-ended question responses. Thus, the qualitative questions of this study were built from the effective teacher descriptors in the multicultural education framework. The resulting themes that were revealed in the teacher responses identify and describe the international teachers' instructional practices directly linked to the multicultural education research. In addition, the responses from the teachers further define and extend the multicultural education research framework within the perceived reality of international educational practice.

General, overall attributes of multicultural education. Findings that emerged from the participant responses indicated that the international teachers exhibited characteristics that link to the general, overall attributes of multicultural education descriptors in the research. The international teachers (a) validated and affirmed differences in ethnicity and cultures, (b) engaged students to view society through the lens of multiple perspectives, and (c) fostered intergroup understanding and respect for others' cultures (Banks, 1999; Gay, 2000; Howard, 1999; Leistyna, 2002; Merryfield, 2002; Nieto, 2004; Reissman, 1994). The teachers revealed their beliefs that the use of home cultures in

instructional practices values student cultures and expands student thinking about other cultures, helps teachers better understand students, supports positive interactions between teachers and students, and is a way to connect new learning to known cultural concepts and to a sense of community. The teachers exhibited beliefs in the value of each child, that it is important to understand students, that teachers need to know the world from which their students come, and must show interest in cultural values different from their own. It was also expressed that teachers need to believe in families and understand the importance of connecting students' parents and home cultures. As participant 10 stated, the classroom should be a place where "differences are explored, discussed, and celebrated," to enable students to learn and appreciate others, to learn how to negotiate multiple cultures, and to be able to contribute to society.

Multicultural teachers. A positive finding is that respondents elected to capitalize on the multiple and rich opportunities that the international teaching experience provides and in which they believe to be important as a function of their lives wherever they might be living. The international teachers (a) see their profession as a way to give back to society and encourage their students to do the same, (b) show respect to students and develop personal bonds with them, (c) believe they can make a difference through their teaching, and (d) exhibit high expectations for all students (Banks, 1999; Howard, 1999; Ladson-Billings as cited in Webb, 1998; Leistyna, 2002; Nieto, 2004; Reissman, 1994; Zeichner, Melnick, & Gomez, 1996).

The international teachers in this study were highly engaged in school activities, in volunteer work, and in social and cultural events within the local community. These

activities provided an avenue for the teachers to not only participate for their own enjoyment, but for the betterment of others and society. This epitomizes the concept of giving back to society. Volunteer and service activities included outreach projects, fund raising events, environmental awareness efforts, support of refugee schools and families, sponsorship of economically disadvantaged children and populations, and multiple volunteer work in the community and at school. The teachers also encouraged students to engage in these types of activities and often did service projects side by side with their students. Teachers reported that this type of active involvement was a reflection of their own personal beliefs that giving back to others was an important aspect of who they are as a person, that the involvement helped them learn about other cultures, that it helped them connect with students and families, that it helped themselves and their students connect to the real world, and that it contributed to the development of their own personal characteristics and attributes. It is interesting to note that most of the teachers indicated they would be doing these activities in any location they lived and that an international location provided another avenue for helping others. Thus, it might be possible to conclude that the international experience provides multiple and rich opportunities for the teachers to engage in service activities, which they believe to be important as a function of their lives wherever they might be living.

It was seen over and over again that the international teachers showed respect to their students and developed personal bonds with them. This was seen in the multiple reflections throughout the responses that highlighted the importance of connecting with families, in using the student and family cultures as avenues for teaching and engaging students in learning about many different cultures, and in the approach to designing lessons that utilized and were built upon student cultures. Teachers perceived that using home cultures in the classroom not only valued students' cultures, but also expanded student thinking about other cultures, helped the teacher better understand the students supporting positive interactions between teacher and student, encouraged teacher learning of differing cultures promoting cultural awareness on the part of the teacher, and helped connect new student learning to known cultural concepts. By using home cultures in the classroom, teachers believed this created a personalized connection between the student and the classroom learning. Teachers wanted to understand the cultural values of students and their parents, wanted to be sensitive to the cultural values and norms as expressed through social behaviors, and tried to individualize learning by using cultural connections. Teachers appeared to develop personal bonds with students and their cultures through lessons and through highlighting cultures present in the classroom.

Teachers also created tasks that reflected a global perspective, which they believed, supported an overall classroom climate of respect.

Teachers indicated great sensitivity to students' languages and the challenges students face in classrooms in which the learning was not taught in their native language. Larke (1990) indicated that the value teachers place on the student's language is an important component in the expectation level the teacher sets for the student. Because language is a very important part of children's lives and culture, if teachers value the child's language then they are valuing the student's family and culture. These teachers

valued and respected student languages, thus also valued and respected students' homes and cultures.

Teachers also described their belief in the inherent and intrinsic value of each child, that all students have talents, that it is important to understand each student, and to expect the best from each student while motivating and encouraging them to reach their full potential as successful contributors to their families and society. These beliefs also reflect respect for students and the importance of developing personal relationships.

These teachers reflected a belief they can make a difference through their teaching (efficacy) and detailed multiple attributes that described their passions and personal values regarding teaching. It was obviously apparent that these teachers have a great passion for the education profession and highly value the work they do with students. In order to make a difference in the lives of students, teachers indicated they need to listen to students, let students know they care about them, provide students with challenges and choices and need to help students "see the best in each other and become aware of the value of another's thoughts" (participant 10). Teachers believed that the students' homes and family cultures are a very important connection to student learning and that the classroom should be a risk-free, safe, and comfortable learning environment. In addition, to make a difference, teachers revealed that they themselves need to model personal beliefs and that the classroom should be a place where diversity is celebrated, explored and discussed.

Efficacy and high expectations for students are intertwined. Teacher perceptions and beliefs about non-majority students determine the level of expectations set in the

classroom and the kinds of treatment students receive (Gilbert & Gay as cited in Larke, Wiseman, & Bradley, 1990; Gollnick & Chinn, 2002; Sleeter & Grant as cited in Larke, Wiseman, & Bradley, 1990). The international teachers overwhelmingly stated they hold high expectations for all students and that expectations were not differentiated based on ethnicity or culture. However, academic expectations were reported to be differentiated for students when necessitated by ability (special needs students) or language (English Language Learners), which is appropriate. Warren (2002) described that teachers who exhibit low expectations often times lower their teaching standards, produce less effort, and use less rigorous curricula. He further indicated that some teachers who lowered expectations expressed beliefs that they are not responsible for helping students overcome what is perceived as deficits, they viewed students' families and cultures as deficits rather than strengths, they lacked confidence to help students overcome barriers to learning, and they lacked determination to carry out the work needed to teach these students. However, when "teachers hold high expectations and when those expectations are reflected in their behavior toward children, these expectations are related to student cognitive changes" (Persell, 1997, p. 98). The international teachers truly exhibit the attitude, belief, and determination to maintain high expectations for all students, which supports high academic achievement.

Multicultural schools. Linking to the descriptors in the research that specifically define a multicultural school, findings that emerged from the responses provided in this study indicate that the international teachers (a) encourage parents to be a part of school activities and programs, (b) proactively encompass diversity in curriculum materials,

instructional techniques, and assessments used in the classroom, and (c) seek to teach students the culture of the school while maintaining the students' sense of their own cultural identity and pride (Banks, 1999; Gay, 2000; Leistyna, 2002; Nieto, 2004; Zeichner, Melnick, & Gomez, 1996).

As has been seen over and over again the international teachers in this study are highly committed to the connection between school and home cultures and reported that they definitely encourage parents to be a part of school activities and programs. They encourage parents to be guest speakers to share their cultural customs, values, and traditions with students in the classroom, they encourage parents to contribute to classroom lessons and activities, and they encourage parents to participate in cultural experiences within school sponsored activities and in the local culture. The connections that are built from these activities help develop the parent – teacher relationship. In turn, this creates enhanced communication and encourages parents to become more involved in their child's education.

As also highlighted in the next two narrative sections in this study, the international teachers who responded to this study are very responsive to ensuring materials used in the classroom support diverse cultures present in their classrooms and they use appropriate instructional strategies that support high levels of thinking. In regard to assessments, the teachers reported that they modify assessments for students with special needs and English Language Learners (ELL), are sensitive to culture and modified tests accordingly, and ensure tests were culture free. It was noted that vocabulary particularly needed to be paid attention to on assessments and they would

change vocabulary as needed to accommodate students needs, would provide further assistance, allow students to use support mechanisms such as dictionaries when taking a test, use vocabulary words that had been front loaded in learning prior to taking a test, and ensure the assessment actually measures concepts rather than vocabulary. In addition, alternative tests would be used to accommodate student needs, multiple formats for assessments are used, performance and project based assessments are used, and rubrics, or scoring guides, are also used in assessments to help accommodate various and diverse learning styles and cultures.

In this study, and as noted in other sections, that when seeking to teach students the culture of the school while maintaining the student's sense of their own cultural identity and pride, teachers revealed great commitment to connecting learning to the students' homes and cultures. They perceived that using the home cultures in the classroom values students' cultures and expands student thinking about other cultures. It helps teachers better understand the students and supports positive interactions between the teacher and students. It also offers a way to connect new learning to students' known cultural concepts.

Multicultural instructional materials. From the participant responses it was found that the international teachers are culturally responsive to ensuring appropriate materials are used in their classrooms. The international teachers (a) strive to use instructional materials that are inclusive of racial, ethnic, and cultural perspectives, (b) incorporate multicultural information and resources in content areas, and (c) use materials to help

develop more positive racial and ethnic attitudes (Banks, 1999; Gay, 2000; Leistyna, 2002; Zeichner, Melnick, & Gomez, 1996).

The preponderance of international teachers in this study appeared to be aware of possible bias in educational materials, especially in textbooks used in the classroom. Even though they acknowledged that publishing companies are creating textbooks that are more responsive to diversity, the teachers also indicated that textbooks tend to be culturally biased toward the country from which they are written. Being aware of possible textbook and materials bias, teachers counter the apparent bias by highlighting differences and similarities between textbook information and student cultures, by clarifying concepts to ensure student understanding, and by using supplemental materials to ensure a balanced approach when using materials in the classroom.

Multicultural instructional strategies. It would appear from the findings in this study that the international teachers (a) provide for cross-cultural experiential and collaborative learning, (b) create personalized learning for students, (c) set high expectations for students, and (d) recognize the value of using students' cultures as a worthwhile resource from which to draw upon for examples, data, and information in the teaching process (Banks, 1999; Gay, 2000; Leistyna, 2002; Merryfield, 2002; Zeichner, Melnick, & Gomez, 1996). In addition, the teachers reported incorporating the use of cooperative learning, similarities and differences, and other effective teaching strategies in their lessons (Marzano, Pickering, & Pollack, 2001).

The international teachers in this study provided for cross-cultural experiential, collaborative, and challenging learning through multiple strategies. Cross-cultural

experiences were provided by the teachers through their efforts to incorporate students' home cultures into lessons and tasks, as well as connecting the local cultural community to classroom learning. Teachers used specific units based on cultures, connected content area units and activities to cultures, and highlighted cultural holidays, ceremonies, and festivals to encourage students to learn about other cultures and celebrate cultural differences. In addition, teachers invited parents into their classrooms to share various aspects of their cultural heritage and home countries and provided learning tasks that reflected a global perspective and supported an overall classroom climate of respect. Teachers also set high expectations for all students, regardless of ethnicity, but with consideration for ability and language.

When provided the list of most effective teaching strategies from the metaanalysis of Marzano, Pickering, and Pollock (2001), the international teachers identified
cooperative learning and identifying similarities and differences as the two most powerful
effective teaching strategies that supported cross-cultural experiential, authentic and
interactive learning in their diverse classrooms. Research indicates "of all the grouping
strategies, cooperative learning may be the most flexible and powerful" (Marzano,
Pickering, & Pollack, 2001, p. 91). The international teachers viewed cooperative
learning as a strong strategy for group processing, building group dynamics, and
supporting English Language Learners in the learning process. They also indicated that
cooperative learning allows students to be more responsible and take ownership of their
learning, provides social opportunities for students, and lowers anxiety levels for students
in regards to learning.

Research also indicates that identifying similarities and differences has the highest average effect size of the nine categories that identify instructional strategies that affect student achievement (Marzano, Pickering, & Pollock, 2001). Within the context of a diverse classroom, international teachers reported that similarities and differences was an excellent way for students to gain understandings of various cultures and cultural elements. It was also reported that this was an effective strategy to help students understand how their own perspective fits into a broader global perspective.

To create personalized learning for students, teachers reported the use of various instructional strategies, reported responding to student needs regarding language, reported strategies that connected to students' home cultures, and reported modifying assessments. Various specific strategies included one-one-one and small group conferencing and instruction, making use of students' individual learning styles and modes, adapting and modifying tasks, connecting new learning to currently held schema, employing strategies to support the learning of English Language Learners (ELL), also known as English as a Second Language students (ESL), connecting learning to students' home cultures, and modifying assessments. It was very apparent that teachers were highly sensitive to the needs of ELL students and that they modified assessments in response to specific needs of students on individual education plans and those not proficient in English.

Recognizing that students' cultures and experiences are a worthwhile resource from which to use examples, data, and information in the teaching process was reflected multiple times in the data in this study. The importance of connecting to students'

cultures was obvious over and over again and revealed that teachers believed this to be a compelling response required of teaching effectively in a diverse classroom and acted on that belief. The theme of connecting to students' cultures arose multiple times throughout the narrative, open-ended question responses and the discussion question responses. Connecting to students' home cultures was important to teachers for various reasons. The international teachers believed that connecting to the home cultures of their students helped the teachers understand their students better and understand the values and social behaviors of the students and their families. The teachers generally sought to identify and create connections between the curriculum lessons and student cultures, encouraged discussions about cultures in the classroom, highlighted and celebrated cultures represented in the classrooms, encouraged parents to be guest speakers and to volunteer in the classroom, and encouraged students to share cultural views. In addition, teachers created learning tasks in which students could use aspects of their own culture to complete the task, encouraged students to be proud of their cultural heritage, connected real world situations to students' cultures, and allowed students to use learning resources in their native languages when completing assignments. Teachers noted that using concepts, ideas, and values from the students' backgrounds helps the teacher better understand students and supports positive interactions between students and between students and teacher. Teachers can learn from the students, increasing the teacher's schema of various cultures. Teachers also perceived connecting home cultures to the classroom values students' cultures and expands student thinking about other cultures.

Lastly, using home cultures in the classroom was perceived as a way to link new learning to students' known cultural concepts and to a sense of community.

Social justice perspective. Linking to the descriptors in the research regarding multicultural education and a social justice perspective, a positive finding from responses provided in this study that these international teachers reflected a social justice perspective in varying degrees and sought to become involved in proactive activities that helped develop a more humane society within their communities. In addition, these teachers sought to nurture passion for social justice within their students and helped them become involved in community activities that built their personal knowledge and character in regard to helping others (Howard, 1999; Nieto, 2004; Reissman, 1994; Zeichner, Melnick, & Gomez, 1996).

There was an overwhelming response by the teachers highlighting their commitment to volunteer work and community service in their local communities. The teachers believed that by being involved in various activities they learned about other cultures, connected better to students and their families, connected better to the real world, and contributed to the development of their own personal characteristics and attributes. Their involvement included working with the outreach/service projects, fund raising events, volunteering in various organizations, and raised funding for various medical research projects and environmental awareness issues. They worked with and supported various outreach programs for the local underprivileged populations, completed service learning projects with students, supported refugees, and sponsored children in schools with economically disadvantaged students. These teachers were

committed to making the world a better place and identified in their personal values that caring about others was important. Participant 10 very eloquently stated beliefs that encompass a social justice perspective and to which others expressed in differing degrees.

Values that I hold to include: the right of all to a life free from pain, threat or suffering, the responsibility of the more fortunate towards the rest of humankind, the need for all to be aware of environmental issues and act to protect the earth, tolerance and quest for understanding of those who are different from ourselves, perseverance (especially in the face of ignorance and apathy), all living creatures are worthy of respect. I hope I can help students to see the best in each other and become more aware of the value of another's thoughts. I believe that if I can encourage them to feel positive about themselves they will approach learning confidently and master it successfully. If my classroom is a place where differences are explored, discussed, and celebrated, perhaps my students will grow up better able to lead the world forward, in whatever capacity and in whatever country and culture they find themselves.

Research Ouestion 7

How does the cross-cultural effectiveness of teachers in international schools link to the instructional practices they use in their classrooms?

From analyses of the participant responses, it is apparent that these teachers exhibited emotional resilience as measured by the CCAI and that these characteristics are linked to their beliefs and instructional practices. As evidenced in the statistical analysis of the CCAI, the means of the total scores and the scores of each of the four dimensions indicated the international teachers were more cross-cultural adaptable than the Iowa teachers. However, only in the emotional resilience dimension was there a statistically significant difference. Thus, research question 7 of this study will be specific to reviewing the link between the emotional resilience of the international teachers and their instructional practices.

The CCAI dimension of emotional resilience measures the extent to which a person can rebound and react positively to difficult, negative, or unpleasant feelings that tend to be a part of a cross-cultural experience. It also measures the extent to which an individual can maintain emotional equilibrium in a changing or new environment. People who score higher on the emotional resilient dimension tend to have a sense of humor, have self-confidence, exhibit a sense of adventure, have a positive attitude, and tend to take risks and experiment. They tend to like new experiences, things, and situations, can deal with the setbacks and difficulties that are a normal part of a cross-cultural experience, can deal with stressful feelings in a constructive way, and are not overly concerned with making mistakes (Church, 1982; Guthrie, 1975; Kelley & Meyers, 1995).

From the responses of the international teachers, it is possible to loosely link the descriptors of the emotional resilience dimension and attributes reported by the teachers. In discussion question four, the teachers provided great insight describing their perceptions about their personal values and beliefs. Enthusiasm, curiosity, openness, sense of humor, self-confidence, willingness to make mistakes, looking at things with a fresh or new attitude, tolerance, and being creative were reported attributes that directly link to the emotional resilience dimension. From overall patterns and themes from the question responses in the study, it can be stated that these teachers have a passion for the educational work they do in cultural environments that are different from their own and that they have the abilities as described in this dimension to effectively interact with those who are culturally different than themselves. This includes the everyday challenges these

teachers face in teaching the diverse students in their classrooms. Participant seven exemplifies how these teachers deal with the difficulties that are a normal part of a cross-cultural experience and deal with stressful feelings in a constructive way.

The biggest challenge that I have encountered teaching in an international setting is the language differential. In order to reach and teach your students you have to address them in a discourse that they understand. When I first came here I was looking at a sea of blank faces for the first two months of the academic year! It was disconcerting to say the least. I still see the blank faces at the beginning of the school year but I have learned to be patient with myself and my students to overcome this stumbling block to effective instruction.

It can also be noted that there was a sense that these teachers truly enjoyed the new experiences they come across on a regular basis in an international setting. As participant 14 states:

We are products of our experiences...We have had many wonderful cultural experiences because of where we have lived, and because we have been involved with people and their customs from around the world...Learning about other people's cultures has enriched our own. That appreciation and understanding permeates our daily interaction with students, parents, and community members...All of these experiences lend themselves to embracing diversity.

Implications for Future Research

There is an overall passion and intense commitment to effective educational practices, international teaching, and life in general revealed in the responses of the teachers in this study. These characteristics are also identified in the multicultural education research framework used as a foundation in this study. As an exploratory study designed to investigate differences in cross-cultural effectiveness between classroom teachers in international settings and classroom teachers in Iowa, and to examine, within a multicultural education framework, instructional practices international teachers perceive to be effective in meeting the needs of a diverse student population, this

study merely opened the door to looking closer at the implications of instructional practices and teacher attitudes prevalent in international schools. Further studies are recommended that will extend and expand the learning of this study and contribute to the growing research regarding effective teaching practices for diverse classrooms.

It is recommended that an investigation be undertaken to identify the underlying characteristics and personality traits of international teachers that support their overall passion, commitment, and beliefs to the international education concept. Personality differences could be investigated using published instruments between stateside and international teachers. It is possible that this type of study could lead to the creation of a simple inventory that could be used when hiring personnel for highly diverse schools and classrooms.

It is suggested that further research be developed that more deeply explores the topics reflected in the narrative, open-ended questions four, six, and seven from this study. These covered materials used in the classroom, differentiating academic expectations, and professional growth opportunities. In this study, these questions did not lead to deeper probing, but with a restructuring of the questions, informational research could be conducted that would elicit more in-depth answers.

Similar studies could be conducted comparing teachers from monocultural backgrounds with those from multicultural backgrounds, correlating demographic characteristics with CCAI results of differing teacher groups, and/or using larger sample sizes for both the Iowa teachers and the international teachers.

With the advent of the No Child Left Behind Act (U.S. Department of Education, n.d.), it would be an important study to link components investigated in this study with student achievement. For example, multicultural practices could be explored in relationship to specific student achievement. Another example would be to link multicultural education framework with effective teaching practices as described in the meta-analysis of research of effective teaching strategies in the work of Marzano, Pickering, and Pollack (2001) comparing student achievement results with specific strategies.

In the current study, several teachers in the Iowa group were from schools with student populations representing multiple races and cultures, thus giving the teachers more experience with diversity which modified their perceptions and responses. Similar studies involving teachers from more isolated locations reflecting monocultural student populations are recommended. This would include teachers who have had no experience with diversity and are not teaching in an urban, diverse setting. Conversely, a similar study could be completed involving teachers from large, urban settings only.

Each year in late February, the Overseas Teaching Fairs begin on the east coast, the Midwest, and on the west coast. It is at these recruiting fairs that teachers are hired to teach in international schools. It could be a powerful study to complete a pretest using the CCAI with a random sample of teachers with no international experience who were then hired during the recruiting fairs for an international teaching experience. A posttest could then be completed using the CCAI after two years of the teachers' international experience. The differences in their perceptions as measured on the CCAI could be

compared. Such a study would investigate the direct impact of an international experience on the cross-cultural adaptability of teachers and would contribute powerful information to the educational field.

Lastly, it would be suggested that a self-assessment inquiry be conducted investigating how international teachers perceive themselves on a continuum defined in the multicultural education research ranging from an awareness level on one end of the continuum to the social justice perspective on the other end of the continuum. The teachers could place themselves on such a continuum and then identify the factors that contributed to their perceptions of where they are on the continuum.

There are additional multiple studies that could spring from the frameworks that describe multicultural education, effectiveness of teachers with diverse populations, teacher efficacy, instructional practices, and connecting learning to home cultures. The most important aspect to consider is that the classrooms of the future in the United States are projected to continue to become more diverse in regard to ethnicity, race, religion, culture, language, physical ability, and mental ability. Conversely, the teachers of the classrooms are projected to continue to predominantly come from a monocultural background. Research highlights the attributes and practices of effective teachers in diverse settings, as well as the experiences that contribute to creating this type of educator. The most important aspect to consider is that if America is going to be competitive in the new global world, it is imperative that we continue to investigate the realities of teaching practices and add to the research framework. There are thousands of classrooms with thousands of students who need multiculturally effective teachers. The

students deserve to be provided the best and most powerful education possible and it is our responsibility to provide it.

REFERENCES

- Appelbaum, P. (2002). Multicultural and diversity education: A reference handbook. Santa Barbara, CA: ABC-CLIO, Inc.
- Banks, J. A. (1999). An introduction to multicultural education. Needham Heights, MA: Allyn & Bacon.
- Banks, J. A. (2002). An introduction to multicultural education. Boston: Allyn & Bacon.
- Banks, J. A., & McGee-Banks, C. A. (Eds.). (1997). *Multicultural education: Issues and perspectives*. Needham Heights, MA: Allyn & Bacon.
- Banks, J. A., & McGee-Banks, C. A. (Eds.). (2005). *Multicultural education: Issues and perspectives*. Hoboken, NJ: John Wiley & Sons, Inc.
- Bennett, C. I. (2003). Comprehensive multicultural education: Theory and practice. Boston: Pearson Education, Inc.
- Brown v. Board of Education of Topeka, Shawnee County, Kansas, 374 U.S. 483 (1954).
- Cardman, M. (2002). Paige outlines ED's international priorities. *Education Daily*, 35(223), 4.
- Carey, K. (2004). The real value of teachers: If good teachers matter why don't we act like it? *Thinking K 16, 8*(1), 3-32. Retrieved August 31, 2005, from http://www2.edtrust.org/NR/rdonlyres/5704CBA6-CE12-46D0-A852-D2E2B4638885/0/Spring04.pdf
- Cetron, M. & Cetron, K. (2003/2004). A forecast for schools. *Educational Leadership*, 61(4), 22-29.
- Church, A. (1982). Sojourner Adjustment. Psychological Bulletin, 91(3), 540-572.
- Cohen, J. (1988). Statistical power analysis for the behavioral sciences. Hillsdale, NJ: Lawrence Erlbaum Associates, Inc.
- Connelly, F. M., & Clandinin, D. J. (1988). *Teachers as curriculum planners: Narratives of experience.* New York: Teachers College Press.
- Cooper A., Beare, P., & Thorman, J. (1990). Preparing teachers for diversity: A comparison of student teaching experiences in Minnesota and South Texas. *Action in Teacher Education*, XII(3), 1-4.

- Craft, M. (Ed.). (1996). Teacher education in plural societies: An international review. Washington, DC: Falmer Press.
- Craig, S., Hull, K., Haggart, A., Perez-Selles, M. (2000). Promoting cultural competence through teacher assistance teams. *Teaching Exceptional Children*, 32(3), 6-12.
- Cruickshank, D. R. (1996). *Preparing America's teachers*. Bloomington, IN: Phi Delta Kappa Educational Foundation.
- Cummins, J. (2004). Foreword. In S. Nieto (Author). *Affirming diversity: The sociopolitical context of multicultural education* (pp. xv-xvii). Boston: Pearson Education, Inc.
- Cushner, K. (Ed.). (1998). *International perspectives on intercultural education*. Mahwah, NJ: Laurence Erlbaum Associates, Inc.
- Davis, L. E. & Turner, J. S. (1993, November). An investigation of the cultural sensitivity of elementary preservice teachers. Paper presented at the Annual Meeting of the Mid-South Educational Research Association, New Orleans, LA. (ERIC Document Reproduction Service No. ED 372054)
- Delpit, L. (1995). Other people's children: Cultural conflict in the classroom. New York: The New Press.
- Department of Defense Education Activity. (n.d.). *DoDEA facts 2003*. Retrieved September 11, 2005, from http://www.dodea.edu/communications/dodeafacts.htm
- Department of Defense Education Activity. (n.d.). *Annual Report*. Retrieved September 11, 2005, from http://www.dodea.edu
- Dilworth, M. E. (Ed.). (1998). Being responsive to cultural differences: How teachers learn. Thousand Oaks, CA: Corwin Press, Inc.
- Drysdale, T. T. (2000). A brief history of the education of U.S. military dependents: 1821 to 2000. In R. J. Simpson & C. R. Duke (Eds.), *American overseas schools* (pp. 15-19). Bloomington, IN: Phi Delta Kappa Educational Foundation.
- Education America. (2000-2003). Department of defense dependents schools info. Retrieved February 2, 2004, from http://educationamerica.net/browse.phtml?sid=nn&eid=1076&a=eip
- Education Week. (2000). A new minority? Retrieved September 9, 2005, from http://www.edweek.org/ew/articles/2000/09/27/04centdivebox3.html

- Eleanor, D. (1997). Preparing teachers for tomorrow's children: Cross-cultural training for teachers. In J. E. King, E. R. Hollins, & W. C. Hayman (Eds.), *Preparing teachers for cultural diversity* (pp. 73-84). New York, NY: Teachers College Press.
- Finney, S. & Orr, J. (1995). "I've really learned a lot, but...": Cross-cultural understanding and teacher education in a racist society. *Journal of Teacher Education*, 46(5), 327-333.
- Friedman, T. L. (2005). The world is flat: A brief history of the twenty-first century. New York, NY: Farrar, Strauss, and Giroux.
- Futrell, M. H., Gomez, J., & Bedden, D. (2003). Teaching the children of a new America: The challenge of diversity. *Phi Delta Kappan*, 84(5), 381-385.
- Garcia, M. J. (1984). Acceptance by social studies teacher educators of the basic tenets of multicultural education. Dissertation Abstracts International, 45(07), 2059A. (UMI No. 8422230).
- Gay, G. (2000). Culturally responsive teaching. New York, NY: Teachers College Press.
- Glatthorn, A.A., Carr, J.F., & Harris, D.E. (2001). *Planning and organizing for curriculum renewal: A chapter of the curriculum handbook.* Alexandria, VA: Association for Supervision and Curriculum Development.
- Gollnick, D. M. & Chinn, P. C. (2002). *Multicultural education in a pluralistic society*. Upper Saddle River, NJ: Pearson Education, Inc.
- Gomez, M. L. (1996). Prospective teachers' perspectives on teaching "other people's children." In Zeichner, K. M., Melnick, S. L., & Gomez, M. L (Eds.), *Preparing teachers to teach global perspectives* (pp. 109-132). Thousand Oaks, CA: Corwin Press, Inc.
- Goodson, I. F. (1990). Studying curriculum: Towards a social constructivist perspective. *Journal of Curriculum Studies*, 22(4), 299-312.
- Grant, C. A. (1994). Best practices in teacher preparation for urban schools: Lessons from the multicultural teacher education literature. *Action in Teacher Education*, *XVI*(3), 1-18.
- Grant, C. A. (Ed.). (1995). Educating for diversity. Needham Heights, MA: Allyn & Bacon.

- Grant, C. A. & Sleeter, C. E. (2003). Turning on learning: Five approaches for multicultural teaching plans for race, class, gender, and disability. New York, NY: John Wiley & Sons, Inc.
- Gutek, G. L. (1993). American education in a global society: Internationalizing teacher education. New York: Longman Publishing Group.
- Guthrie, G. (1975). Behavioral analysis of culture learning. In R.W. Brislin, S. Bochner, & W. J. Lonner (Eds.), *Cross-cultural perspectives on learning* (pp. 95-115). New York, NY: SAGE Publications, Inc.
- Harmon, D. (2002). They won't teach me: The voices of gifted African American innercity students. *Roeper Review*, 24(2), 68-75.
- Hartoonian, M. & Van Scotter, R. (1996). School to work: A model for learning a living. *Phi Delta Kappan*, 77(8), 555-560.
- Howard, G. R. (1999). We can't teach what we don't know. New York: Teachers College Press.
- Huber, T. & Kline, F. (1993). Attitude toward diversity: Can teacher education programs really make a difference? *Teacher Educator*, 29(1), 15-23.
- Huber, T., Kline, F.M., Bakken, L., & Clark, F. L. (1997). Transforming teacher education: Including culturally responsive pedagogy. In J.E. King, R.R. Hollins, & W.C. Hayman (Eds.), *Preparing teachers for cultural diversity* (pp. 129-145). New York: Teachers College Press.
- Huck, S. & Cormier, W. (1996). *Reading statistics and research*. New York: HarperCollins College Publishers.
- Hurn, C.J. (1993). *The limits and possibilities of schooling*. Needham Heights, MA: Allyn & Bacon.
- International School Services. (2003-2004). *The ISS directory of international schools*. Princeton, NJ: The Backes Group, LLC.
- Iowa Department of Education. (1989). A guide to developing multicultural, nonsexist education across the curriculum. Des Moines, IA.
- Iowa Department of Education. (2002). *Iowa teaching standards and model criteria*. Retrieved February 22, 2004, from http://www.state.ia.us/educate/ecese/tqt/tc/doc/itsmc030122.html

- Iowa Department of Education. (2002). *The annual condition of education report 2002*. Des Moines, IA.
- Iowa Department of Education. (2004). The annual condition of education report 2004. Des Moines, IA.
- Jarchow, E., McKay, J. W., Powell, R., & Quinn, L. F. (1996). Communicating personal constructs about diverse cultures: A cross-case analysis of international student teaching. In D. J. McIntyre & D. M. Byrd (Eds.), *Preparing tomorrow's teachers:* The field experience (pp. 178-196). Thousand Oaks, CA: Corwin Press, Inc.
- Kelly, G. A. (1963). A theory of personality: the psychology of personal constructs. New York: W. W. Norton & Co., Inc.
- Kelley, C. & Meyers, J. (1995). *CCAI: Cross-cultural adaptability inventory manual.* Minneapolis, MN: NCS Pearson, Inc.
- Kendall, F. E. (1996). Diversity in the classroom. New York: Teachers College Press.
- King, J. E., Hollins, E. R., & Hayman, W. C. (1997). Preparing Teachers for Cultural Diversity. New York: Teachers College Press.
- Kissock, C. (1997). Student teaching overseas. In M. M. Merryfield, E. Jarchow, & S. Pickert (Eds.). *Preparing teachers to teach global perspectives: A handbook for teacher educators* (pp. 123-142). Thousand Oaks, CA: Corwin Press, Inc.
- Krathwohl, D. R. (1993). Methods of educational and social science research: An integrated approach. White Plains, NY: Longman.
- Ladson-Billings, G. (1994). The dreamkeepers: Successful teachers of African-American children. San Francisco, CA: Jossey-Bass, Inc.
- Larke, P. J. (1990). Cultural diversity awareness inventory: Assessing the sensitivity of preservice teachers. *Action in Teacher Education*, XII(3), 23-29.
- Larke, P. J., Wiseman, D., & Bradley, C. (1990). The minority mentorship project: Changing attitudes of preservice teachers for diverse classrooms. *Action in Teacher Education*, XII(3), 5-11.
- Latham, A. (1999). The teacher student mismatch. Educational Leadership, 56(7), 84-85.
- Lau v. Nichols, 414 U.S. 563 (1974).

- Lewitt, E. M. & Baker, L. G. (1994). Race and ethnicity changes for children. Critical Health Issues for Children and Youth, 4(3), 134-144.
- Leistyna, P. (2002). Defining & designing multiculturalism: One school system's efforts. Albany, NY: State University of New York Press.
- Ligons, C. M., Rosado, L. A., & Houston, W. R. (1998). Culturally literate teachers: Preparation for 21st century schools. In M. E. Dilworth (Ed.), *Being responsive to cultural differences: How teachers learn* (pp. 129-142). Thousand Oaks, CA: Corwin Press, Inc.
- Mahan, J. & Rains, F. (1990). Inservice teachers expand their cultural knowledge and approaches through practica in American Indian communities. *Journal of American Indian Education*, 29(2), 11-24.
- Mahan, J. M. & Stachowski, L. (1990). New horizons: Student teaching abroad to enrich understanding of diversity. *Action in Teacher Education*, XII(3), 13-21.
- Marshall, P. L. (1996). Multicultural teaching concerns: New dimensions in the area of teacher concerns research? *Journal of Educational Research*, 89(6), 371-379.
- Marzano, R.J., Pickering, D.J., & Pollock, J.E. (2001). *Classroom instruction that works*. Alexandria, VA: Association for Supervision and Curriculum Development.
- McCall, A. L. (1995). Constructing conceptions of multicultural teaching: Preservice teachers' life experiences and teacher education. *Journal of Teacher Education*, 46(5), 340-350.
- McIntyre, D. J. & Byrd, D. M. (Eds.). (1996). Preparing tomorrow's teachers: The field experience. Thousand Oaks, CA: Corwin Press, Inc.
- McKay, J.W. & J. Montgomery. (1996-97). Catalyst for change: International student teaching. National Forum of Teacher Education Journal, 7(1), 20-32.
- Melnick, S. L. & Zeichner, K. M. (1998). Teacher education's responsibilities to address diversity issues: Enhancing institutional capacity. *Theory Into Practice, XXXVII*(2), 88-95.
- Merryfield, M. M. (2000). Why aren't teachers being prepared to teach for diversity, equity, and global interconnectedness? A study of lived experiences in the making of multicultural and global educators. *Teaching and Teacher Education*, 16, 429-443.
- Merryfield, M. M. (2002). The difference a global educator can make. *Educational Leadership*, 60(2), 18-21.

- Michie, A. A. (Ed.). (1967). Diversity and interdependence through international education. USA: Education and World Affairs.
- Miles, M. B. & Huberman, A. M. (1994). *Qualitative Data Analysis*. Thousand Oaks, CA: SAGE Publications, Inc.
- Miller, K. (2000). The office of overseas schools of the United States department of state. In R. J. Simpson & C. R. Duke (Eds.), *American overseas schools* (pp. 9-14). Bloomington, IN: Phi Delta Kappa Educational Foundation.
- National Center for Education Statistics. (n.d.). Various reports and tables. Retrieved January 15 February 2, 2004, from http://nces.ed.gov/.
- National Center for Education Statistics. (2003). Digest of education statistics, 2003: Introduction. Retrieved September 7, 2005, from http://nces.ed.gov/programs/digest/d03
- National Center for Education Statistics. (2005) *Indicator 1: Past and projected* elementary and secondary public school enrollments. Retrieved September 7, 2005 from http://nces.ed.gov/programs/coe/2005/section1/indicator01.asp.
- National Education Association. (2003). Despite long hours, low pay, teachers love their profession. Retrieved February 7, 2004, from http://nea.org/newsreleases/2003/nr030827.html
- Nieto, S. (2000). Placing equity front and center: Some thoughts on transforming teacher education for a new century. *Journal of Teacher Education*, 51(3), 180-187.
- Nieto, S. (2002/2003). Profoundly multicultural questions. *Educational Leadership*, 60(4), 6-10.
- Nieto, S. (2004). Affirming diversity: The sociopolitical context of multicultural education. Boston: Pearson Education, Inc.
- Oakes, J. (1986). Tracking, inequality, and the rhetoric of reform: Why schools don't change. *Journal of Education*, 168(1), 85-102.
- Ornstein, A. C. & Hunkins, F. P. (1993). *Curriculum foundations, principles, and theory.* Boston: Allyn and Bacon.
- Pang, V. O., & Sablan, V. A. (1998). Teacher efficacy. In M. E. Dilworth (Ed.), *Being responsive to cultural differences: How teachers learn* (pp. 39-58). Thousand Oaks, CA: Corwin Press, Inc.

- Persell, C. H. (1997). Social class and educational equality. In J. A. Banks & C. A. M. Banks (Eds.), *Multicultural education: Issues and perspectives* (pp.87-107). Needhams Heights, MA: Allyn & Bacon.
- Pickert, S. M. (1992). Preparing for a global community: Achieving an international perspective in higher education. (ASHE-ERIC Higher Education Report No. 2). Washington, DC: The George Washington University, School of Education and Human Development.
- Quinn, L. F., Jarchow, E., Powell, R., Barr, H., & McKay, J. (1995). International student teaching: New Zealand and the United States perspectives on schooling. *Action in Teacher Education*, XVII(2), 18-27.
- Quinn, L. F., Jarchow, & E., Powell, R. (1995-96). Designing Relevant Teacher Education Experiences for Educational Equity and Excellence in a Global Society. Paper presented at the World Council of Curriculum and Instruction Eighth Triennial and Silver Jubilee Conference, Amritsar, India.
- Reissman, R. (1994). *The evolving multicultural classroom*. Alexandria, VA: Association for Supervision and Curriculum Development.
- Samovar, L. A. & Porter, R. E. (2001). *Communication between cultures*. Belmont, CA: Wadsworth/Thomson Learning.
- San Jose State University. (n.d.) What is a "normal school?" Retrieved September 7, 2005, from http://www2.sjsu.edu/sjsuhistory/what_is_a_normal_school.htm
- Seeberg, V., Swadener, B., Vanden-Wyngaard, M., & Rickel, T. (1998). Multicultural Education in the United States. In K. Cushner (Ed.), *International perspectives on intercultural education* (pp. 259-299). Mahwah, NJ: Laurence Erlbaum Associates, Inc.
- Sergiovanni, T. J. (Ed.). (1982). Supervision of teaching. Alexandria, VA: Association for Supervision and Curriculum Development.
- Simpson, R. J. (2000). Foreword. In R. J. Simpson & C. R. Duke (Eds.), *American overseas schools* (pp. 1-4). Bloomington, IN: Phi Delta Kappa Educational Foundation.
- Slick, G. A. (Ed.). (1995). Emerging trends in teacher preparation. Thousand Oaks, CA: Corwin Press, Inc.

- Stachowski, L. L. & Mahan, J. M. (1995). Learning from international field experiences. In G. A. Slick (Ed.), *Emerging trends in teacher preparation: The future of field experiences* (pp. 99-107). Thousand Oaks, CA: Corwin Press.
- Stachowski, L. L. & Mahan, J. M. (1998). Cross-cultural field placements: Student teachers learning from schools and communities. *Theory Into Practice, XXXVII*(2), 155-162.
- U.S. Census Bureau. (n.d). *Projections of the resident population by race, Hispanic origin, and nativity*. Retrieved February 2, 2004, from http://www.census.gov/population/projections/nation/summary/np-t5-g.pdf
- U.S. Department of Education (n.d.). *NCLB overview: Executive summary*. Retrieved September 10, 2005, from http://www.ed.gov/nclb/overview/intro/execsumm.html
- U.S. Department of State (n.d.). Worldwide fact sheet 2004-2005. Retrieved September 11, 2005, from http://www.state.gove/m/a/os/1253.htm
- Valli, L. (1996). Trusting relations, preservice teachers, and multicultural schools. In D. J. McIntyre & D. M. Byrd (Eds.), *Preparing tomorrow's teachers: The field experience* (pp. 26-40). Thousand Oaks, CA: Corwin Press, Inc.
- Vogt, P.W. (1999). Dictionary of statistics & methodology: A nontechnical guide for the social sciences. Thousand Oaks, CA: SAGE Publications, Inc.
- Vavrus, M. (1994). A critical analysis of multicultural education infusion during student teaching. *Action in Teacher Education XVI*(3), 46-47.
- Warren, S. R. (2002). Stories from the classrooms: How expectations and efficacy of diverse teachers affect the academic performance of children in poor urban schools. *Educational Horizons*, 80(3), 109-116.
- Webb, M. (1998). Culture: A view toward the unexplored frontier. In M. E. Dilworth (Ed.), *Being responsive to cultural differences* (pp.61-77). Thousand Oaks, CA: Corwin Press, Inc.
- Wilson, A. H. (1993). Conversation partners: Helping students gain a global perspective through cross-cultural experiences. *Theory Into Practice, XXXII*(1), 21-26.
- Youngs, C. S. & Youngs, G. A., Jr. (2001). Predictors of mainstream teachers' attitudes toward ESL students. *TESOL Quarterly*, 35(1), 97-120.
- Zeichner, K. (1996). Educating teachers for cultural diversity in the United States. In M. Craft (Ed.). *Teacher education in plural societies*. (pp. 141-158).

- Zeichner, K. M. & Hoeft, K. (1996). Teacher socialization for cultural diversity. In J. P. Sikula (Ed.), *Handbook of research on teacher education* (pp. 525-547). New York: Simon & Schuster Macmillan.
- Zeichner, K. M. & Melnick, S. L. (1996). Community field experiences and teacher preparation for diversity: A case study. In D. J. McIntyre & D. M. Byrd (Eds.), *Preparing tomorrow's teachers: The field experience* (pp. 41-61). Thousand Oaks, CA: Corwin Press, Inc.
- Zeichner, K. M., Melnick, S. L., & Gomez, M. L. (Eds.). (1996). Currents of reform in preservice teacher education. New York: Teachers College Press.
- Zwingle, E. (1999). A world together. National Geographic, 196(2), 6-33.

APPENDIX A LOCATIONS OF PARTICIPATING TEACHERS

School Locations of Participating International Teachers

Location	School	Type
Aviano, Italy	Aviano Elementary School	DoDDS
Baku, Azerbaijan	Baku International School	Independent
Beijing, China	International School of Beijing	Independent
Caracas, Venezuela	Colegio Internacional de Caracas	Independent
Hong Kong, China	Hong Kong International School	Independent
Johannesburg, South Africa	American International School of Johannesburg	Independent
Monterrey, Mexico	Instituto San Roberto	Independent
Mumbai, India	American School of Bombay	Independent
New Delhi, India	American Embassy School	Independent
Okinawa, Japan	Kadena High School	DoDDS
Taipei, Taiwan	Taipei American School	Independent
Wiesbaden, Germany	Gen.H.H.Arnold High School	DoDDS
Yanbu, Saudi Arabia	Yanbu International School	Independent

DoDDS = Department of Defense Dependent Schools

School Locations of Participating Iowa Teachers (AEA 267)

Town/City	School	School District
Charles City	Charles City High School	Charles City CSD
Clarksville	Clarkesville Community School	Clarksville CSD
Denver	Denver Elementary School	Denver CSD
Dysart	Union Middle School	Union CSD
Fairbank	Wapsie Valley High School	Wapsie Valley CSD
Iowa Falls	Riverbend Middle School	Iowa Falls CSD
Iowa Falls	Pineview Elementary School	Iowa Falls CSD
Iowa Falls	Alden High School	Iowa Fails CSD
Janesville	Janesville Elementary School	Janesville CSD
LaPorte City	LaPorte City Elementary School	Union CSD
Shell Rock	Shell Rock Elementary	Waverly-Shell Rock CSD
Tripoli	Tripoli High School	Tripoli CSD
Waterloo	Elk Run Elementary School	Waterloo CSD
Waterloo	West High School	Waterloo CSD
Waterloo	East High School	Waterloo CSD
Waverly	Waverly-Shell Rock High School	Waverly-Shell Rock CSD
Waverly	Waverly-Shell Rock Jr. High School	Waverly-Shell Rock CSD

CSD = Community School District

APPENDIX B DEMOGRAPHIC QUESTIONS

Demographic Questions

What i	s your gender?				
	Female				
	Male				
What	is your race?				
	Caucasian				
_	African-American				
_	Hispanic		*		
_	Asian				
<u> </u>	Native American				
What:	is your age?				
	29 years or less				
	30 – 39 years				
. 🗅	40 – 49 years				
	50 – 59 years				
	60 years or more				
	grade level do you predominantly teach? Lower elementary (K – 2) Upper elementary (3 – 5/6) Middle school (6 – 8) High school (9 – 12) Other at content areas do you predominantly tea Elementary – all or most areas Reading and Language Arts / English Social Studies Math Science				
0	Specialist – PE, Music, Art, Industrial 7 Other	Fechnology, etc	; .		
	many TOTAL years have you taught (in a international locations)? 5 years or less 6-10 years 11-15 years 16-20 years 21-25 years more than 25 years	all locations tog	gether -	in the U	nited States

How many TOTAL years have you taught in international settings (locations outside the United States)?

- □ 5 or less
- \Box 6 10 years
- \Box 11 15 years
- \Box 16 20 years
- \square 21 25 years
- more than 25 years

What is your level of education?

- □ BA
- □ BA + 15
- \Box BA + 30
- □ MA
- □ MA + 15
- \square MA + 30
- □ EdD / PhD

As part of your training to become a teacher, in what location was your student teaching experience?

- □ United States in a MULTICULTURAL school setting. Please identify the predominant races/cultures that were present in your school.
- □ United States in a predominantly SINGLE culture setting. Please identify the predominant race/culture that was present in your school.
- □ International location please identify the country.

When growing up and attending school from elementary years to your graduation, what was the predominant cultural make-up of your school?

- Predominantly a MULTICULTURAL setting; please identify the predominant race/culture; i.e. White, African-American, German, etc.
- □ Predominantly a SINGLE culture setting; please identify the predominant race/culture; i.e. White, African-American, German, etc.

Please provide the following information regarding the school you are currently teaching in.

- □ Name of your school
- □ Location city
- □ Location state or country

Within what system is your school?

- International school
- □ DoDDS school
- □ Iowa school

What is the total enrollment of your school?

- □ Less than 100 students
- □ Between 100-300 students
- □ Between 300-500 students
- □ Between 500-700 students
- \Box Between 700 –1,000 students
- □ More than 1,000 students

What is the cultural composition of your school population? (Give approximate percentages, i.e. 60% American, 20% Korean, 10% Thai, etc.)

What is the racial composition of your school population? Write in approximate percentages next to each race.

- Caucasian
- □ African-American
- Hispanic
- □ Asian
- □ Native American

APPENDIX C CROSS-CULTURAL ADAPTABILITY INVENTORY

Cross-Cultural Adaptability Inventory

Company Information

NCS Pearson, Inc. denied permission to place exact wording of the 50 items of the CCAI in the appendix section of this dissertation. For that reason, only a summary phrase of each item has been provided. Copies of the CCAI may be purchased from Pearson Assessments, by phone 1-800-627-7271, or website, www.pearsonassessments.com. Authors of the CCAI are Colleen Kelley, PhD and Judith Meyers, PsyD.

Descriptive Statistics of the Normative Sample

N = 653

Dimension	M	SD	
Total Score		225.85	19.63
Emotional Resilience	ER	79.58	8.28
Flexibility/Openness	FO	66.92	7.76
Perceptual Acuity	PAC	46.47	4.96
Personal Autonomy	PA	32.88	3.78

(Kelley & Meyers, 1995, p. 78)

Items of the CCAI

Each item is paraphrased below with the dimension also identified, first in order of the items and then by the four dimensions.

Participant response choices were based on Likert scale: Definitely True (DT), True (T), Tends to Be True (TT), Tends to Be Not True (TNT), Not True (NT), and Definitely Not True (DNT).

	CCAI Item	Dimension
1.	coping with stress in new situations	ER
2.	living a fulfilling life in another culture	FO
3.	understanding thoughts and feelings of others	PAC
4.	coping with life, no matter location	ER
5.	relating to people and enjoying it	FO
6.	accomplishing goals, even in unfamiliar environments	PA
7.	making cultural errors and laughing at self	ER
8.	like being with all types of people	FO
9.	perceiving realistically how others see self	PAC
10.	needing approval from those of other cultural backgrounds	ER
11.	liking people even though they don't share interests	FO

12.	valuing others equally	PA
13.	trying new things and liking it	ER
14.	impatience with a slower pace of life	FO
15.	giving people different from self benefit of the doubt	PAC
	hiring people from different backgrounds than self	ER
	following own ideas, if they different than others	PA
	enjoying life while living anywhere	ER
	important to impress those different than self	FO
	perceiving others' feelings, even different than me	PAC
	making friends easily	ER
	sense of loneliness with people different than self	FO
	enjoying new, different foods	ER
	believing all cultures have worthwhile offerings	PAC
	maintaining own values	PA
	liking self, even if failed in different, new living situation	ER
	understanding others different than self	FO
	aware of how others' cultural differences affect their perceptions of me	PAC
	enjoying new experiences	ER
	being alone time, even in unfamiliar environments	FO
	discouragement, even with those different than self	ER
	intolerance of differences in others	FO
33.	thinking how my actions impact others	PAC
34.	positive attitude in unfamiliar situations	ER
	deciding own values even when those around me exhibit different values	PA
36.	coping with difficult feelings in a new culture	ER
37.	judging others when I meet them	FO
38.	understanding others' behaviors in the context of their culture	PAC
39.	coping with unclear situations	ER
40.	learning more about those different than self	FO
41.	beliefs, personal values in relationship to others' standards	PA
42.	trusting own ability to communicate clearly in new situations	ER
43.	enjoying those who think differently than self	FO
44.	keeping an open mind in new situations	PAC
45.	accepting own imperfections, regardless of others' views	ER
46.	giving others different than self benefit of the doubt	FO
47.	expecting that others' respect me, irregardless of their culture	PA
48.	coping with stress in new circumstances or with new people	ER
49.	expecting to like others when meet them	FO
50.	paying attention to body language	PAC

CCAI Questions Organized by Dimension

Emotional Resilience

1.	coping with stress in new situations	ER
4.	coping with life, no matter location	ER
7.	making cultural errors and laughing at self	ER

10. needing	approval from those of other cultural backgrounds	ER
13. trying n	ew things and liking it	ER
16. hiring p	eople from different backgrounds than self	ER
18. enjoying	g life while living anywhere	ER
21. making	friends easily	ER
	g new, different foods	ER
	elf, even if failed in different, new living situation	ER
_	g new experiences	ER
	agement, even with those different than self	ER
	attitude in unfamiliar situations	ER
-	with difficult feelings in a new culture	ER
	with unclear situations	ER
	own ability to communicate clearly in new situations	ER
_	ng own imperfections, regardless of others' views	ER
-	with stress in new circumstances or with new people	ER
Flexibility/	<u>Openness</u>	
2. living a	fulfilling life in another culture	FO
_	to people and enjoying it	FO
	ng with all types of people	FO
	eople even though they don't share interests	FO
~ .	nce with a slower pace of life	FO
-	nt to impress those different than self	FO
-	loneliness with people different than self	FO
	anding others different than self	FO
	one time, even in unfamiliar environments	FO
32. intolera	nce of differences in others	FO
37. judging	others when I meet them	FO
40. learning	more about those different than self	FO
43. enjoyin	g those who think differently than self	FO
	others different than self benefit of the doubt	FO
	ng to like others when meet them	FO
Perceptual	Acuity	
3. underst	anding thoughts and feelings of others	PAC
	ing realistically how others see self	PAC
.=	people different from self benefit of the doubt	PAC
	ing others' feelings, even different than me	PAC
	ng all cultures have worthwhile offerings	PAC
	f how others' cultural differences affect their perceptions of me	PAC
	g how my actions impact others	PAC
	anding others' behaviors in the context of their culture	PAC
	an open mind in new situations	PAC
60		D.4.C

Personal Autonomy

6.	accomplishing goals, even in unfamiliar environments	PA
12.	valuing others equally	PA
17.	following own ideas, if they different than others	PA
25.	maintaining own values	PA
35.	deciding own values even when those around me exhibit different values	PA
41.	beliefs, personal values in relationship to others' standards	PA
47.	expecting that others' respect me, irregardless of their culture	PA

APPENDIX D

NARRATIVE, OPEN-ENDED QUESTIONS

Narrative, Open-Ended Questions

Only the international teachers were directed to answer these questions, which arose out of the multicultural education research framework. The Iowa teachers did not answer these. From these questions, four in-depth discussion questions were derived for the international teachers to complete in an email type discussion.

- 1. What types of activities, lessons, do you use to bridge, or connect, the students' cultures to classroom learning?
- 2. How do you personalize instructional delivery and your teaching to students' cultures, learning styles, and diversity in your classroom?
- 3. How do you modify tests to provide fair assessments for diverse student populations in regard to race, ethnicity, language, and/or ability?
- 4. How do the materials used in your lessons/instruction reflect cultural non-bias? What process is used in your school to ensure the use of culturally non-biased materials?
- 5. How do you provide meaningful tasks for students that are responsive to race, culture, ethnicity, and language, in a culturally respective classroom?
- 6. In your class/school, how are academic expectations differentiated in regard to cultural identity or ability of the students?
- 7. What types of professional growth opportunities do you participate in that are pertinent to teaching a diverse student population?
- 8. What activities are you engaged in within the local community?

APPENDIX E DISCUSSION QUESTIONS

Discussion Questions

These questions arose out of the shorter narrative open-ended questions provided to the international teachers in the survey. Each question was posted as a type of email discussion to which participants responded. Respondents could also read each other participants' answers and respond to those if desired.

Question #1

Within the multicultural educational models in research literature, one of the characteristics of effective teachers is their ability to negotiate other cultures. In addition, the CCAI inventory that you completed gauges a participant's ability to empathize with those who are different from oneself. Involvement with, and understanding of, students' home backgrounds is part of this process, which implies understanding and negotiating community/local cultures. In the survey/narrative question responses, nearly all the participants indicated they are involved with 1) sponsorship of school activities (clubs, sports, drama, PTA, committees, etc), 2) volunteer work in the local community (fundraising, service work, outreach work, etc.) and/or 3) cultural activities (art, music, travel, local events, etc.). Why are these types of involvement important to you as an international teacher? How does it contribute to your effectiveness as a teacher in a diverse classroom?

Question #2

In the survey/narrative question responses, nearly all participants indicated they use a wide variety of instructional strategies from one-on-one to the use of graphic organizers and multiple others. Research highlights the most effective instructional strategies as 1) identifying similarities and differences, 2) summarizing and note taking, 3) reinforcing effort and providing recognition, 4) homework and practice, 5) nonlinguistic representations, 6) cooperative learning, 7) setting objectives and providing feedback, 8) generating and testing hypothesis and 9) questions, cues and advance organizers. In your classroom, which of these do you find to be the most powerful effective teaching strategy that supports cross-cultural experiential, authentic, and interactive learning? Why?

Question #3

In the research within the multicultural education framework, effective teachers are seen as those who seek to teach the students the culture of the school while maintaining the students' sense of their own cultural identity and pride. In the survey/narrative question your responses indicated a variety of ways of using students' home cultures in the instruction of the classroom. Explain your beliefs in how important this practice is in helping students construct knowledge and learn.

Ouestion #4

Research indicates that one of the most powerful factors in student achievement is what the teacher believes about students and what the teacher expects of students. In addition, the teacher's perspective is shaped by their own upbringing, their life experiences, their beliefs and values, and their sense of efficacy (the belief they can change a situation). What personal values do you hold regarding teaching in an international and diverse setting? What do you believe about how you affect student learning?

APPENDIX F

EMAIL INVITATION TO PARTICIPATE - INTERNATIONAL TEACHERS

Debbie Lee

From:

"Randy & Debbie Lee" <rdlee143@mchsi.com>

To:

<rdlee143@mchsi.com>

Sent:

Friday, April 15, 2005 11:37 AM

Invitation to Participate - International

Subject:

Good Day,

I am a doctoral candidate at the University of Northern Iowa working on my research dissertation and invite you to participate in my study. I am (1) examining the cross-cultural effectiveness of international teachers and lowa teachers, and (2) examining the instructional practices of international teachers. I would be asking you to complete a survey, complete a few narrative questions about your instructional practices, and engage in an email discussion about your instructional practices. My design is to have 20 international teachers participate. Please be assured that all answers are kept confidential to the research study and no names that can identify you will be used in the analysis of results in the final dissertation.

<u>Please reply to this email</u> and let me know if you are interested in receiving more information about participation. I would greatly appreciate your assistance and support.

Debbie Lee 823 Sonya Dr. Waterloo, IA 50702 319.233.0018 email: rdlee143@mchsi.com

APPENDIX G

EMAIL INFORMED CONSENT - INTERNATIONAL TEACHERS

Debbie Lee

From:

"Randy & Debbie Lee" <rdlee143@mchsi.com>

To:

<rdlee143@mchsi.com>

Sent: Subject: Friday, April 15, 2005 11:38 AM Informed Consent - International

This is the first step in the study. The university requires that I obtain informed consent from you prior to participating. Please read the information below carefully and then reply with a YES or NO in the subject line of a reply to this email. Just an added note - it's very unusual to conduct research for a dissertation via email messaging. It's kind of interesting and fun to try it this way. Let me know if you have any difficulties at any time. Debbie Lee

UNIVERSITY OF NORTHERN IOWA HUMAN PARTICIPANTS REVIEW - INFORMED CONSENT

Project title: Teacher Perceptions of International Setting and Diverse Student

Populations on Instructional Practices

Name of Investigator: Debbie Lee

After reading the information regarding the study, listed below, it is important that you respond to this email in one of two ways: Please read the information and then respond either YES or NO.

1)
I AM WILLING TO PARTICIPATE (AGREEMENT):

Type YES in the subject field in replying to this email. If you volunteer to participate in this study, please click on REPLY to create an email reply and then type YES in the subject field of the email reply to indicate agreement to the this statement - / am fully aware of the nature and extent of my participation in this project as stated above and the possible risks arising from it. I hereby agree to participate in this project. I acknowledge that I have received an email copy of this consent statement. I am 18 years of age or older. It is very important that you understand this. Please contact me if you have any questions at all about your voluntary participation.

OR

2)
CHOOSE NOT TO PARTICIPATE (NO AGREEMENT):
Type NO in the subject field. If you choose not to participate in this study, please click on REPLY to create an email reply and then type NO in the subject field of the email reply to

indicate that you do not want to participate. It is very important that you understand this. Please contact me if you have any questions at all about your choice not to participate.

STUDY INFORMATION

You are invited to participate in a research project conducted through the University of Northern Iowa. The University requires that you give your agreement to participate in this dissertation research study. The following information is provided to help you make an informed decision whether or not to participate.

This is a combined quantitative and qualitative study designed to explore differences of cross-cultural effectiveness of international classroom teachers and of Iowa classroom teachers and to examine instructional practices of international teachers from within the multicultural education framework.

The process for international teachers involves - 1) completing the Cross-Cultural Adaptability Inventory (CCAI) and answering short, open-ended questions through an online survey tool, and 2) answering discussion questions via email messaging.

All directions for completion of the inventory and the discussion questions will be provided after you have agreed to participate.

Step 1: The CCAI

The participants will complete several demographic questions and the Cross-Cultural Adaptability Inventory (CCAI). The CCAI contains 50 items to be answered on a Likert type scale and will examine cross-cultural effectiveness in both groups of teachers. The beginning of the inventory will include a series of demographic items, such as school location, size of school, years of experience, etc. The inventory and demographic data should not take more than approximately 15 minutes to complete. The time frame for online access to the questions will be a 2 week period designated in later directions.

Narrative, open-ended questions

International participants will complete a series of approximately 8-10 questions that are narrative, open-ended in nature and are specific to instructional practices. These questions will be at the end of the CCAI. It is estimated that these questions will take not more than 30-35 minutes to complete.

Step 2:

Discussion Questions

International participants will also complete a series of discussion questions that will more deeply explore their instructional practices. These questions will come from strands arising from the narrative, open-ended questions. The discussion questions will be completed via email messaging and you will have opportunity to respond to others' answers - thus extending the answers into discussions. Individual posted questions will take approximately 5-10 minutes to complete. The time frame for online access to the questions will be a 2 - 3 week period designated in later directions.

Any information obtained during this study which could identify you will be kept confidential to the degree to which information collected online can be kept confidential. Once the study is completed, all identifying data will be destroyed. Up to that point, all information will remain confidential and kept in a secure location. When reporting data results in the final published dissertation, real names will not be used. The summarized findings with no identifying information may be published in an academic journal or presented at a scholarly conference.

In the data collection stage, all email discussion question responses in step 2 will be open for all participants to read and respond to. Your name may become known to the other participants through the email discussion exchanges, thus your name will not be confidential at this stage of data collection. Answers will be shared and confidentiality in this step will not be possible. Participants will be directed to delete all email messages once the 2-3 week collection period has been completed. Throughout the data analysis and in the published research dissertation, participants' real names will not be used, thus, maintaining anonymity and confidentiality when the data information is compiled, analyzed, and presented in the final published form.

There are minimal risks to you as a participant. There may be some discomfort or mild anxiety when answering questions about personal professional practices, but it is anticipated that this will generally not occur. There is definitely time involved in completing the various steps, which could be a minor inconvenience. However, your participation in the study may offer you a chance for professional reflection on instructional practices you use in your classroom and will offer opportunities for learning through the online discussion regarding instructional practices.

Your participation in this project is completely voluntary. You may choose to not participate at all. If you do participate, you are free to withdraw at any time you choose with no repercussions of any kind.

Your participation in the study is greatly appreciated. If you have any questions at all, you may contact me at any time, presently or in the future. My email address is rdleel43@mchsi.com and my telephone number is 319.233.0018. You may also contact my faculty advisor, Dr. Greg Stefanich. His email is gregory.stefanich@uni.edu and his telephone number is 319.273.2073. You can also contact the office of the Human Participants Coordinator, University of Northern Iowa, at 319.273.2748, for answers to questions about rights of research participants and the participant review process.

APPENDIX H

PARTICIPANT INSTRUCTIONS - INTERNATIONAL TEACHERS

(accompanied the hard copy of the Cross-Cultural Adaptability Inventory

and sent via U.S. postal services)

Participant Instructions

Dissertation Research Study
Teacher Perceptions of International Setting and Diverse Student Populations on Instructional Practices
Debbie Lee
University of Northern Iowa
October 2004

Thank you for agreeing to participate in my research study. I have listed the steps below and the approximate time frames. If at any time you need any questions answered of me, please do not hesitate to email me at rdlee143@mchsi.com or call me 319-233-0018.

The information in this study will be obtained via an online survey tool and email messaging.

Step 1

Enclosed in this packet, please find the Cross-Cultural Adaptability Inventory (CCAI) for you to complete. Due to copyright laws, the publishing company of the CCAI will not allow me to place the exact inventory questions into the online survey tool. However, to enable me to complete my study and to avoid you having to mail the inventory back to me, you will go to the online survey tool at the website listed below, will read the questions from the hard copy in this packet, and will record the answers in the appropriate number in the online survey tool. You will NOT need to mail anything back to me via postal service. (I would encourage you to complete the hard copy of the CCAI if you are interested in seeing your personal results. The directions are included in the inventory. I've done this and it's interesting.)

The Inventory (survey)

At the beginning of the online survey are several demographic questions for you to answer. You will then answer the CCAI inventory questions. Lastly, I've included several narrative, openended questions for you to complete that ask you to reflect on your teaching practices. The answers for the narrative section do not need to be extensive, but should be complete. They are intended to be short answer type questions - a paragraph or less (unless you choose to be more extensive).

You should be prepared to set aside 30-40 minutes for the completion of all three sections of the survey, although it is most likely that it will take less time. *Internet Explorer* is the best browser tool to use to access the online survey tool, but using *Netscape* will also work. If your online system is somewhat slow, be prepared to be patient with the online survey tool.

Survey website... http://survey.aea267.k12.ia.us/survey/3261/30ff/ (Please note that the last segment of the website is "3 zero ff" and not a letter O)

Please complete the survey as soon as possible, but no later than ... November 1

Step 2

After completion of the survey by all participants, several weeks will pass and then I will post discussion questions as email messages to all participants for a short length of time. Participants will use the **REPLY ALL** when answering so all participants will read all answers and will have the opportunity to respond through using the **REPLY ALL** function. This creates a type of *email discussion* and allows me to more deeply explore instructional strategies of international teachers.

Discussion Questions will be posted for a 2 - 3 week period around mid - November (possibly sooner).

I will provide directions again in an email message when I start posting questions and will indicate time frames at that time as well.

If you are interested in seeing the results of this study, please send me an email after your participation is completed and I will make arrangements for you to receive a copy of the results section when I have completed the dissertation.

Once again, I am very appreciative of your participation. If there is anything I can do for you, please do not hesitate to ask.



APPENDIX I

IOWA PRINCIPAL / SUPERINTENDENT PHONE SCRIPT

Principal / Superintendent - Phone Script

- I am Debbie Lee, secondary curriculum coordinator for Waterloo.
- I am also a student at UNI working on my doctoral dissertation.
- Purpose
 - o I need your assistance in identifying teachers who will complete a 15 minute online survey as part of my research study.
 - The Iowa teachers need to fit a specific set of 3 criteria that match a sample of teachers from the international community. The 3 match criteria are gender, teaching level (elementary, middle school, high school), and years of teaching experience.
 - o I will provide a chart for you to complete with the criteria listed. I am asking you to identify teachers who match the criteria and provide their name and email address. From that point, I will contact the teachers and ask for their participation.
 - o Please make sure the teachers agree to being placed on the list.
- I will send you the chart via both snail mail and email as an attachment and information regarding the research study I am doing.
- You can return the completed chart to me either in the provided envelope or via email as an attachment. I am not expecting that you complete the entire chart, just complete it the best you can. With several school districts participating, I will get the cross section I need. Please just complete it the best you can.
- Will you be able to assist me with this study?
- Check that I have the email listed correctly.

Thank you!!

APPENDIX J

EMAIL IOWA TEACHER IDENTIFICATION - PRINCIPALS / SUPERINTENDENTS

(also sent via U.S. postal services)

Debbie Lee

From:

"Randy & Debbie Lee" <rdlee143@mchsi.com>

To:

<rdlee143@mchsi.com>

Sent:

Friday, April 15, 2005 11:42 AM

Subject:

Iowa Teacher Identification - Principals and Superintendents

Good Day,

As per our telephone conversation you have agreed to assist me in obtaining teachers from your district to participate in my UNI doctoral research study. I've attached the chart for completion, which also includes a brief description of my study on the backside. I've also included the description below. Please feel free to share the description with teachers. I am also sending the same chart and description of the study to your school district office via snail mail. It should arrive shortly.

I am asking you to complete the name, school, and email address of teachers in your district that match the criteria listed on the attached chart. Please be sure the teachers have agreed to be placed on the list. I am not expecting that each listing will be completely filled by every school district. Do the best you can. I believe a cross sample from multiple districts will provide the information needed. Return the list to me either via the stamped, addressed envelope provided with the snail mail or via email as an attachment (using WORD): rdlee143@mchsi.com. Please return the information within one week of receiving this chart or no later than November 24th.

Thank you.

Debbie Lee 391.233.0018 rdlee143@mchsi.com 823 Sonya Dr. Waterloo, IA 50702

University of Northern Iowa Doctoral Research Study

Teacher Perceptions of International Setting and Diverse Student Populations on Instructional Practices

My research study is a combined quantitative and qualitative study designed to explore differences of cross-cultural effectiveness of international classroom teachers and of lowa classroom teachers and to examine the instructional practices of international teachers from within the multicultural education framework. The involvement for the lowa teacher sample is completion of several demographic questions along with the Cross-Cultural Adaptability Inventory (CCAI).

The CCAI contains 50 items that are answered on a Likert type scale and will examine cross-cultural effectiveness in both groups of teachers. The beginning of the inventory will include a series of demographic questions, such as school location, size of school, years of teaching experience, etc. The inventory and the demographic data should take no longer than approximately 15 minutes to complete. The entire survey will be completed via an online survey tool. It is recommended that teachers have access to the Internet using Internet Explorer as the browser. Further information for completing the survey will be provided once teachers have agreed to participate. All correspondence will be completed via email.

Confidentiality:

Any information obtained during this study which could identify individual teachers will be kept confidential to the degree to which information collected online can be kept confidential. Once the study is completed, all identifying data will be destroyed. Up to that point, all information will remain confidential and kept in a secure location. When reporting data results in the final published dissertation, real names will not be used. The summarized findings with no identifying information may be published in an academic journal or presented at a scholarly conference.

Debbie Lee University of Northern Iowa Doctoral Research Study

Teacher Perceptions of International Setting and Diverse Student Populations on Instructional Practices

My research study is a combined quantitative and qualitative study designed to explore differences of cross-cultural effectiveness of international classroom teachers and of Iowa classroom teachers and to examine the instructional practices of international teachers from within the multicultural education framework. The involvement for the Iowa teacher sample is completion of several demographic questions along with the Cross-Cultural Adaptability Inventory (CCAI).

The CCAI contains 50 items that are answered on a Likert type scale and will examine cross-cultural effectiveness in both groups of teachers. The beginning of the inventory will include a series of demographic questions, such as school location, size of school, years of teaching experience, etc. The inventory and the demographic data should take no longer than approximately 15 minutes to complete. The entire survey will be completed via an online survey tool. It is recommended that teachers have access to the Internet using Internet Explorer as the browser. Further information for completing the survey will be provided once teachers have agreed to participate. All correspondence will be completed via email.

Confidentiality:

Any information obtained during this study which could identify individual teachers will be kept confidential to the degree to which information collected online can be kept confidential. Once the study is completed, all identifying data will be destroyed. Up to that point, all information will remain confidential and kept in a secure location. When reporting data results in the final published dissertation, real names will not be used. The summarized findings with no identifying information may be published in an academic journal or presented at a scholarly conference.

If there are any questions, please feel free to contact me either by phone, 319.322.0018 or via email, rdlee143@mchsi.com.

Iowa Teacher Sample

Directions: Please complete the name, school, and email address of teachers in your district that match the criteria listed below. Please be sure the teachers have agreed to be placed on the list. I am not expecting that each listing will be filled by every school district. Do the best you can. Return the list to me either via the envelope provided or via email as an attachment (using WORD): rdlee143@mchsi.com. Please return the information within one week of receiving this chart or no later than November 24th. A description of my research study is on the backside, which you can share with teachers if you choose. Thank you.

Grade Level	Gender	Years Experience	Name	School	Email Address
Elementary K-2	Female	Less than 5			
Elementary K-2	Female	6-10			
Elementary K-2	Female	11- 15			
Elementary K-2	Male	Over 25			
Elementary 3-5/6	Male	6-10			
Elementary 3-5/6	Male	11-15			· · · · · · · · · · · · · · · · · · ·
Elementary 3-5/6	Female	21-25			
Elementary 3-5/6	Male	Over 25			
Elementary 3-3/0	Iviaic	Over 23			
Middle School	Male	Less than 5			
Middle School	Female	Over 25	,		
Middle School	Male	Over 25			
High School	Female	11-15			
High School	Male	11-15			
High School	Male	11-15			
High School	Male	11-15			
High School	Male	21-25			
High School	Female	Over 25			
High School	Female	Over 25			
High School	Female	Over 25			
High School	Male_	Over 25			

APPENDIX K

EMAIL INVITATION TO PARTICIPATE - IOWA TEACHERS

Debbie Lee

From:

"Randy & Debbie Lee" <rdlee143@mchsi.com>

To:

<rdlee143@mchsi.com>

Sent: Subject: Friday, April 15, 2005 11:34 AM Invitation to Participate - Iowa

Good Day,

Your school district Superintendent provided your name to me regarding the possibility of your participation in my research study. I am a doctoral candidate at the University of Northern Iowa working on my research dissertation and invite you to participate in my study. I am (1) examining the cross-cultural effectiveness of international teachers and lowa teachers, and (2) examining the instructional practices of international teachers. I would be asking you, as an lowa teacher, to complete a few demographic questions along with the Cross-Cultural Adaptability Inventory (CCAI). This will only take approximately 15 minutes. My design is to have 20 international teachers and 20 lowa teachers participate in the study. Please be assured that all answers are kept confidential to the research study and no names that can identify you will be used in the analysis of results in the final dissertation.

Step #1 - <u>Please reply to this email</u> and let me know if you are interested in receiving more information about participation. I would greatly appreciate your assistance and support. <u>If you reply in the affirmative, please also include your snail mail (postal) address.</u>

Step #2 -1 will send you the Consent for Participation form via email and when you have replied to that, I will send you a packet of information via snail mail and you will be able to complete the survey.

Debbie Lee 823 Sonya Dr. Waterloo, IA 50702 319.233.0018 email: rdlee143@mchsi.com

APPENDIX L

EMAIL INFORMED CONSENT - IOWA TEACHERS

Debbie Lee

From:

"Randy & Debbie Lee" <rdlee143@mchsi.com>

To:

<rdlee143@mchsi.com>

Sent: Subject: Friday, April 15, 2005 11:40 AM Informed Consent - Iowa

This is the first step in the study. The university requires that I obtain informed consent from you prior to participating. Please read the information below carefully and then reply with a YES or NO in the subject line of a reply to this email. Just an added note - it's very unusual to conduct research for a dissertation via email messaging. It's kind of interesting and fun to try it this way. Let me know if you have any difficulties at any time. Debbie Lee

UNIVERSITY OF NORTHERN IOWA HUMAN PARTICIPANTS REVIEW - INFORMED CONSENT - IOWA TEACHERS

Project title: Teacher Perceptions of International Setting and Diverse Student

Populations on Instructional Practices

Name of Investigator: Debbie Lee

After reading the information regarding the study, listed below, it is important that you respond to this email in one of two ways: Please read the information and then respond either YES or NO.

1) I AM WILLING TO PARTICIPATE (AGREEMENT):

Type YES in the subject field in replying to this email. If you volunteer to participate in this study, please click on REPLY to create an email reply and then type YES in the subject field of the email reply to indicate agreement to this statement -lam fully aware of the nature and extent of my participation in this project as stated above and the possible risks arising from it. I hereby agree to participate in this project. I acknowledge that I have received an email copy of this consent statement. I am 18 years of age or older.

It is very important that you understand this. Please contact me if you have any questions at all about your voluntary participation.

OR

2)

I CHOOSE NOT TO PARTICIPATE (NO AGREEMENT):

Type NO in the subject field in replying to this email. If you choose not to participate in this study, please click on REPLY to create an email reply and then type NO in the subject field of the email reply to indicate that you do not want to participate. It is very important that you understand this. Please contact me if you have any questions at all about your choice not to participate.

STUDY INFORMATION

You are invited to participate in a research project conducted through the University of Northern Iowa. The University requires that you give your agreement to participate in this dissertation research study. The following information is provided to help you make an informed decision whether or not to participate.

This is a combined quantitative and qualitative study designed to explore differences of cross-cultural effectiveness of international classroom teachers and of Iowa classroom teachers and to examine instructional practices of international teachers from within the multicultural education framework.

For Iowa teachers, the process involves completing the Cross-Cultural Adaptability Inventory (CCAI) through an online survey tool. The CCAI contains 50 items to be answered on a Likert type scale and will examine cross-cultural effectiveness in both groups of teachers. The beginning of the inventory will include a series of demographic items, such as school location, size of school, years of experience, etc. The inventory and demographic data should not take more than approximately 15 minutes to complete. The time frame for online access to the questions will be a 2-week period designated in later directions. All directions for completion of the survey will be provided after you have agreed to participate.

Any information obtained during this study which could identify you will be kept confidential to the degree to which information collected online can be kept confidential. Once the study is completed, all identifying data will be destroyed. Up to that point, all information will remain confidential and kept in a secure location. When reporting the data results in the final published dissertation, real names will not be used. The summarized findings with no identifying information may be published in an academic journal or presented at a scholarly conference.

There are minimal risks to you as a participant. There may be some discomfort or mild anxiety when answering the inventory questions, but it is anticipated that this will generally not occur. There is definitely time involved for completing the inventory, about 15 minutes, which could be a minor inconvenience. However, your participation in the study may offer you a chance for professional reflection regarding cross-cultural effectiveness.

Your participation in this project is completely voluntary. You may choose to not participate at all. If you do participate, you are free to withdraw at any time you choose with no repercussions of any kind.

Your participation in the study is greatly appreciated. If you have any questions at all, you may contact me at any time, presently or in the future. My email address is

rdleel43@mchsi.com and my telephone number is 319.233.0018. You may also contact my faculty advisor, Dr. Greg Stefanich. His email is gregory.stefanich@uni.edu and his telephone number is 319.273.2073. You can also contact the office of the Human Participants Coordinator, University of Northern Iowa, at 319.273.2748, for answers to questions about rights of research participants and the participant review process.

APPENDIX M

PARTICIPANT INSTRUCTIONS - IOWA TEACHERS

(accompanied the hard copy of the Cross-Cultural Adaptability Inventory and sent via U.S. postal services)

Participant Instructions – Iowa Teachers

Dissertation Research Study
Teacher Perceptions of International Setting and Diverse Student Populations on Instructional Practices
Debbie Lee
University of Northern Iowa
December 2004

Thank you for agreeing to participate in my research study. If at any time you need any questions answered of me, please do not hesitate to email me at rdlee143@mchsi.com or call me 319-233-0018.

The information in this study will be obtained via an online survey tool and email messaging.

Enclosed in this packet with these directions, please find the Cross-Cultural Adaptability Inventory (CCAI). Due to copyright laws, the publishing company of the CCAI will not allow me to place the exact inventory questions into the online survey tool. However, to enable me to complete my study and to avoid you having to mail the inventory back to me, you will go to the online survey tool at the website listed below, will read the questions from the hard copy in this packet, and will record the answers in the appropriate number in the online survey tool. You will NOT need to mail anything back to me via postal service. (I would encourage you to complete the hard copy of the CCAI if you are interested in seeing your personal results. The directions are included in the inventory. I've done this and it's interesting.)

At the beginning of the online survey are several demographic questions for you to answer along with the CCAI questions. You should be prepared to set aside 15-20 minutes for the completion of the entire survey. *Internet Explorer* is the best browser tool to use to access the online survey tool, but using *Netscape* will also work. If your online system is somewhat slow, be prepared to be patient with the online survey tool.

Survey website... http://survey.aea267.k12.ia.us/survey/3261/30ff/ (Please note that the last segment of the website is "3 zero ff" and not a letter O)

Please complete the survey as soon as possible, but no later than...December 20

If you are interested in seeing the results of this study, please send me an email after your participation is completed and I will make arrangements for you to receive a copy of the results section when I have completed the dissertation. Once again, I am very appreciative of your participation. If there is anything I can do for you, please do not hesitate to ask.



APPENDIX N

CROSS CULTURAL ADAPTABILITY INVENTORY DATA

Cross-Cultural Adaptability Inventory Numerical Raw Data

Column	Variable Label	Values and Value Labels	
A	Respondent Number	This column is repeated on each page	
В	Gender	1 = Female	
		2 = Male	
С	Race	1 = Caucasian	
D	Age	1 = 29 years or less	
		2 = 30 - 39	
		3 = 40 49	
		4 = 50 - 59	
		5 = More than 60 years	
E	Grade Level	1 = Elementary	
	Glade Devel	2 = Middle School	
		3 = High School	
F	Content Area	1 = Elementary	
1	Contont Area	2 = Reading/Language Arts/English	
		3 = Social Studies	
·		4 = Math	
		5 = Science	
		6 = Specialist	
		7 = Other	
G	Total Years Taught	1 = 5 years or less	
		2 = 6 to 10	
		3 = 11 to 15	
		4 = 16 to 20	
		5 = 21 to 25	
		6 = more than 25 years	
Н	Total Years Taught International	1 = 5 years or less	
		2 = 6 to 10	
		3 = 11 to 15	
		4 = 16 to 20	
		5 = 21 to 25	
		6 = more than 25 years	
I	Education Level	1 = BA	
		2 = BA + 15	
		3 = BA + 30	
· ·		4 = MA	
		5 = MA + 15	
		6 = MA + 30	
		7 = EdD/PhD	
J	System	1 = Iowa	
		2 = International	
K	School Enrollment	0 = less than 100 students	
		1 = 100 - 300	
		2 = 300 - 500	
		3 = 500 - 700	
		4 = 700 - 1,000	
		5 = more than 1,000 students	
		2 more man 1,000 students	

L	Student Teaching Experience	1 = Multicultural setting
		2 = Single Cultural setting
	·	3 = International setting
		4 = Other
M	School Attended When Growing Up	1 = Multicultural setting
		2 = Single Cultural setting

Specific CCAI Question/Item Numbers

Values:

1 = Definitely Not True 4 = Tends to Be True 2 = Not True 5 = True

3 = Tends to Not Be True 6 = Definitely True

Column	Item Number	Column	Item Number
N	1	AP	29
O	2	AQ	30
P	3	AR	31
Q	4	AS	32
R	5	ΑT	33
S	6	AU	34
T	7	AV	35
U	. 8	$\mathbf{A}\mathbf{W}$	36
V	9	AX	37
W	10	AY	38
\mathbf{X}	11	AZ	39
Y	12	BA	40
Z	13	BB	41
$\mathbf{A}\mathbf{A}$	14	BC	42
AB	15	BD	43
AC	16	\mathbf{BE}	44
AD	17	\mathbf{BF}	45
AE	18	BG	46
AF	19	BH	47
AG	20	BI	48
AH	21	BJ	49
ΑI	22	BK	50
AJ	23		
AK	24	\mathbf{BL}	Total CCAI Score
AL	25	BM	Total Emotional Resilience
AM	26	BN	Total Flexibility/Openness
AN	27	во	Total Perceptual Acuity
AO	28	BP	Total Personal Autonomy

Reversed Scored Items:

10, 14, 19, 22, 23, 27, 32, 34, 37

A	В	C	D	E	F	G	Н	I
1	1	1	4	3	6	6	5	6
2	1	1	5	3	3	. 6	6	6
3	2	1	4	2	5	6	2	6
4	2	1	4	3	7	3	3	5
5	1	1	4	3	6	6	3	6
6	2	1	2	1	1	2	1	1
7	1	1	2	. 1	1	2	1	4
8	2	1	3	3	5	5	2	4
9	2	1	4	3	4	6	2	4
10	1	1	4	1	1	5	4	1
11	1	1	1	1.	1	1	1	1
12	2	1	4	2	3	1	1	1
13	2	1	3	3	3	. 3	2	6
14	1	1	4	2	2	6	4	4
15	1	1	4	3	5	3	2	7
16	2	1	2	1	1	3	2	. 6
17	2	1	4	1	1	6	5	6
18	1	1	2	1	1	3	3	3
19	2	1	5	1	5	6	3	6
20	2	1	3	3	4	3	3	4
21	2	1	4	3	2	6	1	3
22	2	1	4	1	1	6	1	3
23	2	1	4	2	2	6	11	3
24	· 1	1	3	3	2	6	1	3
25	1	1	3	1	1	5	1	3
26	1	1	4	3	7	6	1	6
27	2	1	2	3	4	3	1	2
28	1	1	1	1	1	1	1	1
29	1	1	4	2	3	6	. 1	3
30	. 2	1	3	3	3	5	1	2
31	2	1	3	1	1	3	1	2
32	1	1	3	3	3	3	1	6
33	1	1	4	3	2	6	1	3
34	2	11	3	3	3	3	1	4
35	2	1	4	1	1	6	1	3
36	1	1	3	1	1	3	1	1
37	2	. 1	1	2	3	1	11	1
38	2	1	3	3	7	5	1	5
39	2	1	2	1	1	2	1	1
40	1	1	4	1	1	6	11	6

A	J	K	L	M	N	0	P	Q
1	2	4	1	2	5	6	5	5
2	2	3	1	1	4	6	5	5 .
3	2	4	2	2	4	5	5	- 5
4	2	1	4	2	6	6	6	5
5	2	1	2	2	5	6	4	5
6	2	2	2	2	4	6	6	4
- 7	2	4	2	2	6	6	5	4
. 8	2	5	2	2	- 5	6	6	6
9	2	5	3	2	5	6	6	6
10	2	5	3	2	6	5	5	6
11	2	5	1	2	5	5	6	5
12	2	2	3	2	4	5	4	5
13	2	5	1	2	4	5	4	5
14	2	5	2	2	5	5	6	6
15	2	2	3	2	5	5	6	5
16	2	1	1	1	6	. 6	6	6
17	2	3	1	1	5	6	4	5
18	2	2	2	2	4	5	-5	5
19	. 2	5	2	2	6	6	5	5
20	2	3	1	2	4	6	6	5
21	1	2	2	2	5	5	5	- 5
22	1	1	2	2	4	5	5	5
23	1	1	2	2	4	3	5	4
24	1	2	2	2	5	5	6	5
25	1	1	2	2	5	3	5	4
26	1	4	2	11	5	4	6	4
27	1	3	2	2	5	4	4	5
28	1	1	2	2	6	5	6	6
29	1 .	5	2	2	5	4	5	5
30	. 1	2	1	2	6	5	6	6
31	1	2	2	1	5	5	6	5
32	. 1	- 5	2	1	4	5	5	6
33	1	4	2	2	5	4	. 5	5
34	1	5	1	2	5	5	5	4
35	1	5	2	2	4	4	6	4
36	1	2	2	2	4	4	5	4
37	1	2	2	2	5	5	5	4
38	1	5	2	1	6	5	6	5
39	1	1	2	2	- 5	5	5	4
40	1	1	2	2	5	4	6	5

Raw Numerical Data - CCAI

A	R	S	T	U	v	W	X	Y
1	5	5	5	5	4	5	5	5
2	6	5	6	6	4	6	5	6
3	4	5	5	4	4	3	4	4
4	6	6	5	6	5	4	3	6
5	5	5	5	6	5	4	5	6
6	5	4	3	5	4	2	3	4
7	4	5	6	5	5	4	4	6
8	6	6	5	5	5	4	5	6
9	6	6	6	5	5	3	5	6
10	5	6	4	5	4	5	5	6
11	5	5	5	5	4	5	4	6
12	5	5	6	6	4	4	5	6
13	4	4	5	4	4	. 5	3	5
14	5	6	5	4	4	5	5	6
15	5	5	5	5	. 5	5	5	6
16	5	5	4	5	5	5	5	6
.17	5	5	4	5	5	4	5	6
18	4	5	5	6	4	4	3	6
19	5	5	5	4	5	4	5	6
20	6	4	6	4	4	4	4	6
21	5	5	5	5	5	5	5	5
22	5	5	5	6	4	3	5	5
23	5	4	6	5	6	4	4	6
24	.6	6	5	5	5	5	6	6
25	6	4	3	5	3	4	5	6
26	5	- 4	5	6	4	5	5	6
27	4	4	5	6	5	2 .	5	5
28	5	- 5	5	5	5	2	3	6
29	5	5	5	5	5	4	4	5
30	6	6	6	6	5	5	5	6
31	4	4	6	4	5	4	4	6
32	5	4	5	6	4	3	4	6
33	6	5	5	6	5	4	4	6
34	5	4	5	5	4	4	4	5
35	6	4	5	6	4	4	4	- 6
36	5	5	4	4	4	3	4	6
37	6	5	5	4	5	2	2	6
38	6	6	5	6	5	5	5	6
39	5	6	4	5	5	5	3	5
40	6	5	6	6	5	4	6	6

A	Z	AA	AB	AC	AD	AE	AF	AG
1	5	2	5	5	4	5	2	4
2	6	3	5	6	4	5	2	4
3	4	3	4	4	4	5	3	4
4	6	3	5	4	5	5	1	4
5	6	3	4	5	5	5	2	4
6	5	6	4	5	6	5	2	5
7	6	3	3	5	6	3	2	3
8	6	4	5	5	4	6	2	4
9	6	3	6	6	6	6	1	6
10	5	2	4	5	3	5	2	4
11	6	4	4	5	3	5	2	5
12	4	2	5	5	3	4	2	4
13	5	4	5	5	4	5	3	5
14	5	3	. 6	6	4	5	2	5
15	6	2	5	5	3	4	3	4
16	6	5	5	6	5	6	3	4
17	4	3	5	6	5	4	2	4
18	4	2	3	4	4	. 3	3.	. 3
19	5	1	5	6	2	3	2	3
20	6	3	6	5	4	6	2	4
21	5	2	5	5	2	2	2	4
22	- 5	2	4	4	3	4	3	4
23	4	3	4	6	- 5	3	2	5
24	5	3	6	6	5	5	2	5
25	5	1	5	6	4	2	3	4
26	5	6	4	5	4	4	2	4
27	6	3	6	6	4	4	2	4
28	5	4	5	6	6	5	1	6
29	5	4	5	4	3	4	2	4
30	5	3	6	6	5	5	2	5
31	3	3	5	.5	4	4	3	5
32	5	3	5	4	4	5	1	4
33	5	3	4	4	3	4	2	4
34	5	4	5	. 5	4	3	3	4
35	3	3	5	5	4	4	3	4
36	5	3	5	4	4	3	. 2	4
37	4	6	4	5	4	1	1	4
38	5	4	6	6	. 4	5	3	5
39	5	4	4	5	5	4	3	4
40	6	3	5	5	4	5	2	4

A	AH	AI	AJ	AK	AL	AM	AN	AO
1	5	2	2	5	5	5	2	5
2	5	1	1	6	5	5	1	4
3	5	2	2	6	4	5	3	4
4	4	2	1	6	6	6	2	5
5	4	3	1	6	6	6	3	2
6	6	5	2	6	4	3	4	5
7	6	3	1	6	6	6	2	5
8	5	3	2	6	6	6	2	5
9	5	1	1	6	6	5	1	4
10	3	3	1	6	5	5	2	6
11	5	2	1	6	5	5	2	4
12	4	3	3	6	4	5	3	4
13	4	2	2	5	4	5	3	5
14	4	2	2	6	6	6	2	4
15	5	2	2	6	6	4	2	4
16	5	2	5	5	5	6	2	5
17	5	3	2	5	4	3	3	4
18	5	3	2	6	5	6	2	3
19	4	1	1	6	6	6	2	5
20	4	2	1	5	5	6	2	3
21	4	2	2	5	5	5	2	2
22	5	3	3	6	5	4	3	3
23	5	3	2	5	5	5	3	4
24	4	3	6	6	6	5	2	5
25	4	2	1	6	6	4	3	3
26	4	2	6	6	6	5	2	4
27	5	3	2	6	5	3	5	2
28	6	1	2	6	6	5	2	4
29	5	3	3	5	4	4	3	4
30	4	3	3	6	6	6	2	5
31	4	2	4	6	6	6	1	4
32	5	1	1	6	5	6	2	4
33	4	. 3	3	6	6	5	4	4
34	4	3	3	5	5	4	3	4
35	4	-3	5	6	6	5	3	5
36	4	2	3	6	5	5	2	4
37	5	3	4	6	5	5	1	2
38	4	3	4	6	6	6	2	4
39	5	4	5	5	4	4	4	4
40	5	2	1	6	5	6	2	4

Raw Numerical Data - CCAI

A	AP	AQ	AR	AS	AT	AU	AV	AW
1	6	3	5	1	5	2	5	5
2	6	6	4	1	6	2	4	4
3	5	3	4	2	5	2	4	4
4	6	4	5	2	6	1	5	5
5	6	5	3	3	6	2	6	5
6	6	5	4	4	3	4	4	3
7	6	- 5	3	4	5	2	5	4
8	6	5	3	2	4	2	4	5
9	6	3	6	1	6	1	6	5
10	5	5	4	2	6	2	4	5
11	5	5	4	2	5	2	6	5
12	4	5	4	2	5	2	4	5
13	5	5	4	4	4	2	5	4
14	6	5	. 5	1	5	2	6	6
15	6	6	5	1	5	2	5	5
16	6	5	5	6	5	2	5	6
17	5	4	3	2	4	2	5	5
18	5	5	3.	2	5	2	5	5
19	6	5	4	2	5	2	5	5
20	6	6	4	1	5	2	5	4
21	5	4	4	2	5	2	5	4
22	5	4	3	2	5.	2	5	4
23	4	- 5	4	1	5	4	5	4
24	5	6	4	1	6	2	6	5
25	5	5	4	1	4	3	4	4
26	5	5	4	2	5	3	4	4
27	5	5	5	1	6	3	5	5
28	5	5	4	1	6	1	6	5
29	5	4	4	3	4	3	3	4
30	5	5	5	5	6	2	5	5
31	3	4	4	.1	4	2	5	4
32	6	5	4	3	5	1	4	4
33	5	3	4	2	5	4	4	4
34	4	3	4	3	5	2	4	4
35	4	6	3	2	5	4	5	4
36	5	4	4	2	5	2	5	4
37	5	4	4	2	5	3	5	4
38	5	4	4	3	6	2	6	5
39	5	4	5	2	4	2	4	4
40	6	6	5	1	6	1	1	5

A	AX	AY	AZ	BA	BB	BC	BD	BE
1	3	5	4	6	5	5	5	5
2	3	4	4	6	6	4	5	4
3	3	4	5	4	5	4	4	4
4	1 ,	5	5	6	6	4	6	6
5	3	4	4	6	6	5	5	4
6	4	5	3	4	5	4	5	4
7	- 3	4	4	5	5	5	4	4
8	3	4	5	5	6	5	5	5
9	3	4	5	4	6	6	4	5
10	3	5	6	5	4	5	5	6
11	2	4	4	5	5	5	5	5
12	3	5	5	5	6	5	4	5
13	3	5	4	5	5	4	4	4
14	1	5	5	5	6	5	5	5
15	2	5	4	5	6	5	• 6	5
16	3	. 5	5	5	5	5	5	5
17	2	2	4	5	6	5	5	5
18	2	5	4	5	5	5	3	4
19	2	4	4	5	5	5	4	5
20	3	5	4	6	5	5	5	5
21	2	2	4	5	5	5	4	4
22	3	4	4	6	5	4	5	5
23	2	4	4	5	5	5	5	5
24	2	5	5	5	5	5	5	5
25	2	2	3	5	5	5	5	4
26	3	5	4	6	6	5	5	5
27	5	2	3	6	5	6	6	4
28	1	. 5	3	5	6	5	5	5
29	3	4	4	5	4	5	5	4
30	2	5	5	5	5	5	5	5
31	2	5	5	1	1	4	5	5
32	3	4	5	5	6	5	4	5
33	2	5	3	5	5	4	4	5
34	3	3	4	6	5	4	4	4
35	3	4	4	5	6	5	5	5
36	3	4	3	5	5 .	4	4	4
37	3	2	5	4	6	6	4	4
38	2	4	5	5	6	4	5	5
39	4	4	4	4	4	4	4	4
40	1	4	5	6	5	5	6	5

A	BF	BG	ВН	BI	BJ	BK	BL	BM
1	5	5	4	5	5	5	222	84
2	4	4	4	4	4	5	221	81
3	4	4	4	4	4	4	200	74
4	5	5	5	6	6	5	234	83
5	4	4	4	5	4	5	225	80
6	4	4	5	3	4	4	214	70
7	4	5	3	5	4	5	220	80
8	5	5	5	6	5	5	236	87
9	6	6	4	6	4	6	237	90
10	4	5	3	6	4	5	222	82
11	3	4	5	5	5	5	220	80
12	4	5	4	5	5	6	218	78
13	4	4	5	4	4	5	211	76
14	5	6	5	6	4	6	235	89
15	5	5	5	5	5	5	228	83
16	5	5	5	6	5	6	249	95
17	6	6	3	5	5	4	214	77
18	4	4	4	5	5	4	205	75
19	6	4	5	5	5	5	217	82
20	6	6	- 5	6	5	4	226	84
21	5	5	4	5	5	5	205	77
22	5	5	4	5	5	4	212	74
23	5	5	4	4	4	3	212	77
24	6	6	5	5	5	6	244	88
25	3	5	4	4	4	4	197	69
26	5	5	5	5	6	5	230	83
27	6	5	6	3	5	2	219	79
28	5	5	5	5	5	5	227	81
29	4	4	5	4	4	5	209	77
30	6	5	5	5	5	5	246	90
31	5	6	4	.4	. 5	5	207	77
32	5	5	5	5	4	5	216	79
33	4	4	5 .	4	4	5	214	76
34	4	4	3	4	3	4	203	72
35	4	5	4	5	5	5	223	76
36	4	5	4	4	4	4	201	69
37	4	4	6	5	5	5	209	76
38	5	5	6	5	. 6	6	243	86
39	4	4	4	4	4	4	212	78
40	5	6	5	6	6	6	231	86

A	BN	ВО	BP
1	57	48	33
2	59	47	34
3	52	44	30
4	59	53	39
5	63	44	38
6	66	46	32
7	59	45	36
8	63	49	37
9	53	54	40
10	58	51	31
11	57	48	35
12	60	48	32
13	57	46	32
14	55	52	39
15	59	50	36
16	67	51	36
17	61	42	34
18	54	42	34
19	53	48	34
20	61	47	34
21	55	42	31
22	62	44	32
23	55	46	34
24	62	55	39
25	55	40	33
26	64	48	35
27	65	41	34
28	53	53	40
29	58	45	29
30	64	54	38
31	50	50	30
32	56	47	34
33	56	48	34
34	58	43	30
35	63	49	35
36	53	45	34
37	54	42	37
38	64	53	40
39	59	43	32
40	63	51	31

APPENDIX O

NARRATIVE, OPEN-ENDED QUESTION RESPONSES

Narrative, Open-Ended Question 1 Responses

Text answers

What types of activities, lessons, do you use to bridge, or connect, the students' cultures to classroom learning?

∢Back

Responses

(20 total)

- 1 Almost all students have had experience living in the US. Some have not. Students are able to choose certain subject matter for their art work that might reflect their ethnicity.
- In a diverse classroom, it is important to get students to share THEIR own cultural views and experiences. That will assist others in a global perspective of similar events. One lesson I used was a "Naming" Unit. I had each student look in a baby book the background of his/her own name and then I shared with the class how different cultures have ceremonies to "give the child their name". Examples used were African (used the naming ceremony in the Roots video). Native American's way to choose the child's name; Hispanic families that join the mother's and father's name to make a new last name; Korean, where the most important member of the extended family get to give the child a name; and the Jewish ceremony that occurs on or before the 8th day. Then I let students tell of other ceremonies that they were involved in such as Christening in churches. One final part of this project was to use Japanese Haiku poetry to explain their name.
- 3 I emphasize the International/global nature of science....noting discoveries, research, etc. done by individuals from many different cultures and eras.
- 4 I try to make my assignments open-ended so that students can incorporate aspects of their own cultures. In class discussions I try to frame questions in the contexts of individual students' culture and experiences.
- I try to connect the students' cultures to the readings we do in class, to the writing assignments, and class discussions. Since I am a language teacher, I also try to recognize the language differences and the different ways that the language reflects the culture.
- Our school has really been focusing on Student Team Learning activities. This involves cooperative learning strategies, Bloom's Taxonmy, and other brain research specific lessons. We had a guest speaker from Canada come to the school and speak about concept attainment. However, most teachers agree that following the complete process is time consuming. We also use the four blocks curriculum to reach the ESL students.
- 7 The Social Studies cicrriculum is very loosely defined here. I use this as an opportunity to develop my own curriculum to learn with the students about their culture and mine. I generally let them explain Mexican customs and holidays to me and then I do a several month long unit on Canada with them. This is a Roman Catholic school so we pray together every morning and ask God to help us be the best me that we can be!
- 8 Not too much in the science classroom... I try to emphasize more the universality of science and scientists.. people from all cultures have contributed
- 9 Whenever I experience a situation which will add cultural meaning to our work, I integrate it with the lesson.
- I encourage students to talk about and bring in examples of things they do or use at home. We try to include stories from many cultures and to celebrate holidays by inviting parents into the classroom. Having so many different cultures and natioanalities represented, there are often times when discussion and elaboration are needed in order for us all to understand why something is done, or why a certain value is honoured. We connect our social studies theme of community to students' own experience as much as we can, by enlisting parental help to explain and give examples of culture, customs, traditions etc. In their own community.
- 11 Many times, I use social studies lessons to provide students the opportunity to teach me about holidays and local events.
- 12 Joural exercise where each student has to choose a new article from any media source on his home country and one other country. Student is expected to make a comment/analysis of the article. Exercise leads to group discussions on various cultures
- As an economics teacher I usually refer to local conditions that they will see on their way to school or in their interactions with markets.
- We create scrapbooks of who we are timelines of their lives, maps of their countries, family trees, etc. These are shared in "class tour" sessions as well as individual pieces are shared in large and small groups. Children share cultural differences by comparing how they celebrate holidays, ceremonies, and festivals. When reading books, we share "Text to Self" observations, which often times reflects their culture, but also shows how we are all alike.
- 15 highlight cultural achievements
- 16 World geography lessons where teacher and students get to talk about the country culture that they come from. Keep the world map open and refer to as much as possible. Celebrate all holidays.
- 17 I draw on each child's cultural background and compare them to others of different backgrounds in an effort to expleain tht no matter how different we might be we are all basically the same.
- 18 I connect our host country as often as possible. I have guest speakers throughout the year talking about various cultural holidays that are celebrated with in my class. My class participates in the schools Festival of Nations celebration. I teach using the PYP/IB program that promotes Internationalism.
- math algoritms from other cultures, history of scientists from other cultures or countries, sharing of countries and homelands on United Nations Day, week-long activities on culture and arts of India, words from other languages in science and math.
- 20 I try to relate to language, places, religions, and geography. Also historical facts about that country, and personal anecdotes.

(20 total)

http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerID=338935&theReport=15883&the... 4/29/2005

Narrative, Open-Ended Question 2 Responses

Text answers

How do you personalize instructional delivery and your teaching to students' cultures, learning styles, and diversity in your classrom?

∢Back

Responses

(20 total)

- I teach using visual, verbal and demonstrational approaches. When I have ESL students in my classroom I give them extra time, have other students help them in their first language if possible, always make sure they understand the assignments by monitoring their progress on art works.
- 2 In Social Studies, it is easy to bring in all the contributions of many cultures by using the family histories of my students. I believe that all children are different learners, and work hard to get the best from everyone, so that means that I adapt assignments to meet the strengths and weakness of each student.
- 3 I work closely with the ESL (English as a Second Language)teachers to assess the level of understanding of each student. Some lessons are shared between the two classes (science and ESL) for the students that need additional time/help.
- 4 I try to be conscious of my students' backgrounds, nationalities, and cultures so that I can individualize my teaching.
- I try to learn as much as possible about the students' cultures. I try to be aware of any "teachable moment" related to the students' cultures where I can personalize the instructional delivery. I often work one-on-one with my students which gives me the opportunity to adapt either the methodology or the materials to students' learning styles.
- I preface each lesson in class with the objective and reason why we are learning a particular concept. I also find that if I can relate it to their previous knowledge, then I find that children are more likely to be motivated in learning the newly presented material. Students like to hear stories of past adventures that I have had in the US. My experiences and thoughts about their culture really seem to interest them. There also seems to be a general level of concern when I am having difficulty with a certain adjustment.
- 7. Yhe culture is extremely homogenous so this is not a big concern. However, I think it is very important to expose students to other cultures and socioeconomic levels (the children at this school tend to be very wealthy). We have a lot of class discussion during our Morning Meeting daily after our prayer. Each child is viewed as an individual in my classroom and I try to set aside as much time as possible for one-on-one conferencing. I also try to provide the children with time on a daily basis to express individual opinions and ideas with the class. Cooperative learning is also encouraged on a regular basis.
- 8 I think I tend to emphasize more about the differences in learning styles than those due to cultural differences. Most of my kids who struggle in science struggle with the understanding the cognitive ideas themselves rather than any cultural issues about them.
- 9 I try to differeniate instruction on many levels which in learning styles and cultural background as well as ability levels.
- In the first place, we have classes of twenty or less, which makes individual attention easier. I try to make myself aware early in the year of students' preferred learning styles and to build opportunities to use them into my units. The overall curriculum drives the focus of instruction, but how I deliver it depends on the students. Language differences in particular demand a lot of practical discovery activities, visibly posted vocabulary, graphic aids and organisers, repetition of key ideas, restatement in simple language and questioning. As far as culture is concerned, I make myself aware (as much as possible- after 23 years abroad I know that I can still make bad mistakes) of the broad brush-strokes of a child's culture. For example, many cultures are not comfortable with disagreeing with a point of view expressed by someone in authority. It is part of my job, not to change their outlook completely, but to introduce them gently to the idea that this is another way of doing things. I try to be sensitive to parents' views as well, although they may also be reluctant to express them!
- I provide a variety of opportunities to use written and spoken language throughout the school day. I look for opportunities to make curriculum connections to students' lives especially in social studies, science, and guided reading.
- 12 I try to discuss each students home country and make comment on the culture. Students are encourage to discuss their home culture and how it mixes/clashes with their present country's culture.
- 13 I usually will schedule more one on one conferences in an effort to clarify any issues that are present.
- 14 Variety and flexibility are key elements to meeting individual needs. Students are allowed to use dictionaries that translate English into their language to gain clarity and understanding of new vocabulary. Instruction and activities are given orally, visually, and kinesthetically. If I am aware that student's culture will play a part in our activities, I will make a special effort to have them share their perspectives and reasons for them during discussion times.
- 15 students research information relevant to home culture and present to class
- 16 Understanding the values of the student and their parents as well as the ways in which they learn best then letting them perform to their potential instead of thrusting particular learning styles.
- 1 really don't. I stress to my students that we are all more alike than different. Student's cultures generally don't impact upon my teaching. Learning styles, on the other hand, I will address on a one to one basis.
- 1 use differentiation and when appropriate I will highlight different cultures that are represented in the classroom. But, not one single culture is highlighted in my classroom. Because we are a PYP school we need to think global in our instruction.
- accommodation of ESL students in cooperative group learning, reduction and altering assignments for students who are receiving remedial or second-language help, using the teaching assistant for small group work with those students, and giving lots of opportunities for ESL students to present orally their thinking and visual ideas in math and science, tutoring every Monday afternoon for 1 hour in math and science for students who need help.
- For cultures, see above. Learning styles: provide diverse activities and assessment strategies. Diversity: lots of random pairing, and for ESL kids, giving them questions I know they can answer aloud.

http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerID=338936&theReport=15883&the... 4/29/2005

Narrative, Open-ended Question 3 Responses

Text answers

How do you modify tests to provide fair assessments for diverse student populations in regard to race, ethnicity, language, and/or ability?

∢Back

Responses

(20 total)

- The few tests I give are available for ESL or LI students to take in a resource room.
- Every child can learn, so it is important to set the learning module to fit that child. The smartest kid in the class CAN learn the whole lesson, the middle child can learn about 1/2 to 3/4, and the bottom quartrile needs to be asked to learn about 1/4 of the total lesson (they can get the major points, without all the supporting details.) So using some of the concepts of Bloom's Taxlomy, I give each child the chance to do the questions that are in their range of ability.
- 3 I allow beginning ESL students to use dictionaries on assessments and to finish tests, etc. in their ESL class for further assistance with vocabulary.
- I try to avoid vocabulary that might be specific to a particular region or country and that not all students would understand. I generally give my students a copy of the test beforehand. That way they can ask any questions about it before actually taking it. If necessary, I modify the test based on their questions and ability to understand it.
- I modify tests by providing background information or cultural explanations when necessary. In reading tests I adapt both the level of the texts and the types of questions to fairly assess a student's ability. I teach specific learning strategies in my classes and help students become identify their own learning styles. I provide students with a broad range of assessments. I also consider alternative assessments for students who need them.
- I don't modify too much of the cirriculum to meet their abilities. Sometimes, I allow students to use a Spanish/English dictionary to understand vocabulary not associated with the lesson. The school tests the 3rd, 5th, and 7th grades using the Iowa Test of Basic Skills. I find that the students generally perform below grade level in each catergory other than math. The math computation section seems to be easier because of the removal of written english. We formed a small committee at one time to suggest that the school find a more suitable measurement of basic skills, however that has since disbanded.
- This has not been an issue for me here. I have not had children in IEP's and, as stated previously, the group is culturally homogeneous. Sometimes a student requires a little bit of extra time or positive reinforcement to complete a task, but that is about the extent of it!:)
- 8 Ability? Heck yes Language? depends (ESOL, for example) Race/ethnicity? No
- all of assessment as differentiated in terms of ability and are sensitive to language and culture
- For language, we often have tests administered in the English as a Second language classroom, and we work with ESL teachers to develop fair and reasonable assessments. In class I will often modify to allow a child to demonstrate or verbally give answers. I may read directions to the child or read each question individually aloud.
- 11 I make a lot of language modifications rewriting exams published by textbook companies, reading instructions, discussing answers before giving written questions to set students up for success. As their language skills improve, I make fewer modifications to their work.
- Students are allowed as much time as required to finish assessment exercises. Students who are challenged are permitted to take the exercise home to complete. Language, ability etc is taken into account making the assessment which is based on the MYP assessment criteria.
- 13 Being in grades 11 and 12, it is not as big an issue as it might be in the lower grades. However, more time or asking students to use examples form their own culture is usually the primary method of assistance.
- 14 Concepts do not change, but tests are given to students in vocabulary that they know and understand. My tests are teacher made and I am aware of the "common" terms that all of the students have been exposed to.
- 15 tests contain different types of tasks and some illustrated questions
- 16 Make sure test questions are culturally free and offer a variety of assessment techniques instead of simply one style.
- 17 I don't.
- 18 I don't give "tests". In second grade I look at more performance bases assessments or project based using a rubric. If a test has to be given I proof read for cultural blas' and change a test or assignment.
- 19 modify language of exams, provide support with leaders in cooperative group investigations in science and group problem solving strategies in math, and do not assign grades to any descriptor on the report card which relies on first language competence.
- I don't modify tests, but I do make a point of asking ESL or new kids what questions they have, or things they don't understand. I try to work with weak kids BEFORE the test.

(20 total)

∢Back

http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerlD=338937&theReport=15883&the... 4/29/2005

Narrative, Open-ended Question 4 Responses

Text answers

How do the materials used in your lessons / instruction reflect cultural non-bias? What process is used in your school to ensure the use of culturally non-biased materials?

< Back

Responses

(20 total)

- Since I teach art, I don't think we have an issue with biased materials. We look at artwork from many cultures and discuss it in relation to art elements and principles.
- Over the years, our textbooks have become more globally written, to reflect stories, contributions, and pictures of all ethnic/racial groups. Our book committees are tasked to make sure that the text supports that global view when we have the opportunity to select our new books.
- 3 This is one of the critera that we use when selecting new curriculum. Realizing that most textbooks are written in the USA, we are sensitive to and aware of examples, words, geographic allusions, etc. that may cause confusion. We discuss these situations in class, often having fun with differences and similarities.
- 4 Our school follows the IB Diploma Program and Middle Years Program. The IB has put a lot of effort into eliminating cultural bias from their materials.
- Our school curriculum follows the MYP and IB programs. These programs aim to promote intercultural understanding and tolerance of cultural differences. The materials we use in our lessons support these programs; this ensures that they are culturally non-blased.
- The school does not have a formal process in selecting materials that are used in class. Most of the books and materials are suggested by the teachers. In Mexico, most teachers do not have a degree in Education. Most teachers here are degreed, but in different areas. I feel that the school is lacking sufficent resources to meet the demands of the curriculum.
- 1 find that most of our resources in this school are American biased. I, however, use a literature based approach to teaching an integrated curriculum. This makes it much easier to expose children to a wide variety of cultures and lifestyles in an informative and entertaining way. I brought many,many books with me from Canada as there is a limited supply of English literature here; especially cross-cultural stuff. As far as I know there is no set initiative at this school to ensure the use of culturally non-biased materials. It is not a huge priority as there is very little risk of making anyone feel "other" here. Same cultural background and same socioeconomic background except for a nominal amount of scholarship students.
- Hard to answer that since one of our big school goals is to help emphasize the importance of Chinese culture.... also, in science, we seem to be more interested in the concepts rather than the culture
- 9 Most of the books tend to have an american bias so I try to add problems to counteract that situation.
- In choosing US materials, lack of bias is a key criterion when we select. It is impossible to find perfect materials, especially as an international setting (Third culture) is so different from a US-based classroom in cultural and national make-up. I tend to try to adapt and create my own materials, using either the host country, or the students' own backgrounds as a basis.
- Our school is accredited by SACS and the Mexican Board of Education. Materials and teaching resources are purchased to meet SACS standards, so many times there are lessons included within the supplies that aren't as relevant for our students U.S. currency, history
- 12 The school does not employ a process to ensure materials are culturally non-blased. It is up to the individual teach to adjust his/her teaching style and methods to take this into consideration
- 13 Material, being in economics, generally are free from bias. The bias usually comes in the interpretation by students and teachers.
- 14 We have committees that select textbooks and required reading texts. There are many values that go across cultures: honesty, dedication, respect, etc. If bias issues arise in the classroom, we refocus on the fact that all people share these values. As part of their scrapbooks, I ask parents to write a letter to their children explaining what they think are important values their children should adapt for future generations. It is amazing how the letters say the same kinds of values whether the children are American, Thai, Korean, Israeli, or German.
- 15 Due to the British background of the school, there is a bias towards British-based materials. With a growing population of non-British teachers, resources are being ordered that are more culturally balanced.
- Instructional material is the least culturally sensitive area of the classroom and we remind ourselves who wrote the textbook and who it is for and then adapt it to our particular circumstances. The school tries to vet out textbooks that won't work but decisions are made at a district level.
- 17 Texbooks that are chosen and used by Dodds will always reflect cultural diversity. No one culture or race or qhatever is depicted as superior nor inferior to others. Dodds uses a racially and ethnically diverse team of educators to form a team whose responsibility it is to choose the best text books available with an emphals on cultural diversity.
- 18 There is not a process. It is up to the teacher to use materials that are non-bias.
- Instructional materials are evaluated and selected by vertical curriculum teams and are in keeping with the mission of our school to provide education for international students who will meet the needs of a global society.
- 20 They don't. Even in South Africa, we use American textbooks, with American units of measure. I try and translate things into meters and kilos. I also try and tie in Africa and other relevant diverse examples.

(20 total)

 $\label{lem:http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerID=338938\& the Report=15883\& the... \qquad 4/29/2005 + 1/2 + 1$

Narrative, Open-ended Question 5 Responses

Text answers

How do you provide meaningful tasks for students that are responsive to race, culture, ethnicity, and language, in a culturally respective classroom?

≺Back

Responses

(20 total)

- It is rare that comments are made in regards to racial differences in my classroom. We are currently very concerned with sexual harrassment, since this is a major issue with the military.
- It is again, important to make each student see that they are a part of the "whole". It would be mean spirited to focus on the contributions of one culture/racial group to the exclusion of others. As a teacher, this even-handed presentation of all groups must drive your planning of lessons, activities, and testing.
- I am not sure I understand the question.....but we often talk in my science class and in our advisory class about how fortunate we are to be in a school with such great diversity (students from 55 different countries). Because we have a normal turnover of about 25% each year (and often in the middle of the school year)the students are used to new faces coming and going and do an excellent job of welcoming new students.
- 4 I try to ask them to use aspects of their own backgrounds in their work. For example, in the Digital Graphics class that I teach I encourage the students to incorporate aspects of their cultures and nationalities in some of their assignments.
- Again, the MYP and IB programs are very useful in providing meaningful tasks for students and in helping teachers design them in their classes. Since I teach languages, I try to ensure that all of the reading, writing, speaking, and listening tasks are meaningful for students. I also try to a broad range of texts and activities that are responsive to race, culture, ethnicity, and language.
- 1 am lucky enough that the students in my classroom receive 2 hours of formal Spanish instruction. I believe that the tasks we both assign address the issue of cultural sensitivity. We also discuss weekly the current events that are relevant to their lives.
- Please see above response regarding literature. I also use Canada as an example of a very culturally diverse country. Two official languages and numerous ethnic backgrounds coexisting in one country. This seems to captivate the students because the idea of people from many different backgrounds living in one place is guite novel to them.
- from many different backgrounds living in one place is quite novel to them.

 8 by trying to emphasize that the study of science is a human endeavor that can at times transcent race, culture, etc.
- 9 One vehicle I use the history of math from different parts of the world.
- By being constantly on the alert and never trusting a publisher! By changing, adapting, looking for ideas in all kinds of places, by ensuring that the children themselves value and respect differences. While I can never know every culture represented in my classroom well enough to avoid mistakes and offence, I can give my students the right and confidence to correct me if I am wrong. Tasks which are meaningful in one context can often be adapted to work in another, good teaching strategies are amazingly flexible.
- Teachers are given a lot of freedom to determine curriculum, so we make some modifications to our resources and supplement materials when we think that it is needed.
- Primarily through role play, and through journal discussions
- 13 Usually they include use of research sources in their native tongue (nation) as well as use of examples from their native culture or country.
- 14 The tasks are centered around concepts that universal. I teach ESL, so the language that I teach is English. Although the students may translate words and ideas from their own language they speak only English in the classroom. We all work on correct pronunciation, so the issue of an accent due to race is a nonissue. We concentrate on reading skills and strategies, on presentation skills. The focus is not on individual differences, but the things that we all have in common.
- 15 Science-based investigations
- Tasks should apply to their real world experience and back to their home country culture. Students feel free to express their differences.
 A lot of cooperative learning takes place in culture free setting.
- 17 Meaningful tasks provided to my students are not bassed upon any of the chilren's differences, hose students who need coddling or reassurance in reference to their heritage or background are given such. I allow such students to express to me and to others why they are proud of their backgrounds.
- 18 Through projects, sharing and cultural events at the school.
- 19 I have 21 students from 14 countries in my class. I provide the best instruction and meaningful education for all of them. We have 55 languages spoken in the elementary school, and we respect all cultures, and religions, in the daily routines of our classroom and our elementary school.
- 20 I try and develop examples using various cultural connections, le population density of India vs Botswana.n

(20 total)

∢Back

http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerID=338939&theReport=15883&the... 4/29/2005

Narrative, Open-ended Question 6 Responses

Text answers

In your class / school, how are academic expectations differentiated in regard to cultural identity or ability of the students?

∢Back

Responses

(20 total)

- I don't think academic expectations are differentiated other than with IEP students who are allowed extra time, preferential sitting, etc. and with those who are ESL. There is a new reading program for HS students whose reading levels are especially low.
- In our school, we reward students every quarter for any Improvement, i.e., moving .5 grade higher deserves recognition as Important as the students that hit the 4.0 scale.
- 3 Beginning ESL students are allowed grades other than the standard A,B,C grade that describe their progress in my 7th grade science class.
- 4 I try to keep the same standard for all students regardless of background.
- In some special cases, the curriculum is modified to meet the needs of individual students. However, the academic expectations of the MYP and IB programs are very clear and apply to all students.
- The parents and the school both have set high expectations for the students. I am working to change the focus of high grades and instead focus on the necessary skills needed in today's ever changing world.
- 7 There is not much differentiation line expectations. The school is just beginning to develop its first adapted and individual educational plans at present to meet then learning and evaluation needs of some of its students.
- 8 ability? same as in any school (esp with special needs kids, ESOL kids, etc) cultural identity? none at all
- 9 Three different tracks are offered for varying ability levels in math and I'm too sure of the other disciplines other than we have a humanities program that focuses one year on Asian studies.
- That is a hard question to answer. Children at AES receive ESL instruction to give them good skills in literacy, the report card indicates where modifications are made for them in class while they are learning. Children with learning differences are catered for in the Learning centre, but this help is somewhat limited at present. In class a child with special needs will have modifications made by the teacher to tasks and assessment will reflect that.
- Parents expect their students to receive high marks often times there are discrepancies between perceived ability level by local teachers and internationally hired teachers.
- 12 There is no policy in this regard
- 13 External curriculum of I.B. is pretty straight forward.
- All students are mainstreamed. We do have special classes in addition for those students who are learning English or who have learning difficulties. Until a student is at least in Intermediate ESL, the mainstream teachers have the option of grading them as Not progressing, minimal progress, progressing, or outstanding progress. If a student can achieve a grade of C or better, the teacher will opt for the grade.
- 15 We are an IB/MYP/PYP school. Students are assessed according to prescribed IB criteria
- 16 We have modified programs for students who are new to the english language or may have learning disabilities otherwise it is the standard Elementary grades of Meeting Expections, Exceeding, or Doesn't!
- 17 NY academic expectations are the same for all of my students regardless of their cultural background. Material is adjusted for those students on an IEP to best meet their needs.
- Academic expectations are the same regarding the culture except for ESL students who are learning English.
- 19 There are no academic expections which are different with ESL kids, only the assessments assigned and evaluated. Students in the Learning Center gets special class on content-support for 45 minutes alternate days.
- For cultural identity, they aren't. (Although someone today pointed out that most of the low achieving students are from the Mid East. For low ability, there is a well staffed Resource program. (2 teachers for 20 kids). Also, programs might be modified in terms of reduced content.

(20 total)

< Back

http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerID=338940&theReport=15883&the... 4/29/2005

Narrative, Open-ended Question 7 Responses

Text answers

What types of professional growth opportunities do you participate in that are pertinent to teaching a diverse student population?

∢Back

Responses

(20 total)

- I have taught so long in the overseas setting, I don't even think about the fact that I have so many students of varying ethnic
- backgrounds. They are all kids to me. Computer training/technology is the big push in the art program now.

 Our system here offers a multitude of opportunities to learn how to support a diverse student population. We have had courses that deal with ethnicity, learning style, ways to adapt lessons for the global classroom, and how to include all backgrounds in lessons. Many workshops are given to address learning style, ways to adapt lessons, and methods of teaching diverse students.

 We have a "teaming" approach to our middle school program. The 7th grade team meets twice a week to discuss students, strategies,
- 3 We have a "tearning" approach to our middle school program. The 7th grade team meets twice a week to discuss students, strategies activities, and programs. If I felt I needed extra help or instuction in how to deal with our diverse population then we have excellent professional growth support.
- 4 Learning Spanish.
- I read educational journals and other books that help me understand and teach a diverse population. I discuss the issues with my colleagues. I attend workshops and courses that are designed for internatinal educators.
- I feel that the school has provided very little cutural sensitivity training to myself and the other foreign teachers. The school does offer its teachers a Master's degree from Framingham State College in MA. The degree will be a M.Ed in Teaching in American Schools Overseas. This degree is designed to address the needs of students in any culture.
- 7 None really. Please see answers above for clarification regarding this.
- 8 Conferences help a lot (EARCOS, NESA, etc)
- 9 I regularly attend the regions teacher's conference.
- 10 I am working on a Diploma in Special Needs in Education from a UK university by distance learning. Our ESL department hosts sensitization sessions. ESL conference to be held here in January. Any relevant NESA conference sessions.
- 11 Teachers at the school have the opportunity to earn a masters degree in education from a school in the U.S. The program offers courses that help educators stay up-to-date in the field of education.
- There are no growth opportunities available in the region.
 Will be taking an ESOL in the mainstream PD in the latter
- Will be taking an ESOL in the mainstream PD inthe latter part of the year. ALready attended a seminar on Korean students In our school.
- 14 Workshops on gender issues, lectures at conferences, and ESL conferences.
- 15 ESL/Mother Tongue workshops
- 16 Aero:saw standards for Math and Science PD in English Acquisition Learners Multiple Intelligences
- Few to none. Dodds generally does not provide such to the teachers.
- Speaking to parents I find is important and getting to know your students. Ask your students questions about their cultures. They are often quite excited that you have asked and celebrate the differences in your classroom. TALK TO YOUR STUDENTS!!!! There is not one workshop or conference that provides better professional growth that your own students.
- 19 Saturday workshop with ESL director on strategies in teaching ESL students.
- 20 None. (We do have some PD, but it is focused on teaching techniques). 2 years ago we all attended an AISA conference, but haven't heard a word about it since."

(20 total)

∢Back

http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerID=338941&theReport=15883&the... 4/29/2005

Narrative, Open-ended Question 8 Responses

Text answers

What activities are you engaged in within the local community?

Back

Responses

(20 total)

- I have taught English to local nationals. I have taken a pottery course from a Japanese potter. I enjoy eating out with local nationals who work on base. I go shopping and visit most sights just like I would in the US.
- I have volunteered at the USO during the deployment of troops to Iraq. I have volunteered at the Military Post Office to put up mail while 90% of our soldiers were in Iraq. I have assisted the community with fund raisers to support our students with scholarship funds. I have been the sponsor of the Junior class, Senior class, verted on the Red Ribbon (Say NO to Drugs) campaigns, I have served on the School Advisory Committee, I have helped to plan ethnic activities for Black History, Hispanic History, Asian/Islander History and other groups months, I sponsor new teachers and their families that arrive in our community by getting their mail box, hotel, and information about the new community, I serve as the chair of my department, I serve on the School Improvement Plan (SIP) committee, I am on the Student Activity Fund committee; I am a work-alcoholic for school!
- 3 I am one of two teachers that has put together our 7th grade service learning program. We work with 50 first grade students from a local grade school (in New Delhi, India) on language development, games, and arts and crafts.
- 4 I have only been in Venezuela for about 2.5 months. So far I have travelled outside Caracas three times, visited a number of galleries and exhibits within the city, and begun learning Spanish.
- 5 I try to explore my surroundings. I like to discover new places. I enoy cultural activities such as visiting museums, attending concerts. I seek out opportunities to meet people within the community. Where possible, I try to become involved with community service activities.
- 6 I attend church on a semi regular basis. I also participate in a program that provides food for families of limited resources. I occasionally play organized sports with local ex-pats. There is a large foreign community here that has many activities in which I can participate in.
- Not too many! Sadly, most of my activities are related to the school community. I regularly donate various goods to various individuals. I have a Mexican dog (Chihuahua) who I got here and named Mexico after the country where she is from. Much of my spare time is spent walking and caring for her. There is a dog park and we often go there to meet other dogs and it is a good opportunity to meet other Mexican people as well! I have been involved with several Walk/runs that raise funds for Cancer research and children's hospitals her in Mexico. I also try to attend the theatre when I can to watch various dance and orchestral events.
- 8 Church related stuff Scouts Astronomy related outreach Track and Field coaching Kids sports
- 9 90
- 10 Reach Out (community service weekly with local underpriviledged people) Volunteer with Pulse Polio campaign.
- 11 I attend many of my students' after-school activities (sports, theater, dance, music), social events with local teachers, and church.
- 12 1 am involved in a reachout program which brings local internally displaced students into the school. I am also involved in a sponsored walk program that raises funds for environmental awareness
- 13 None at this time.
- 14 We have a Reach Out program that helps the needy by providing English instruction, health instruction, furnishes a library for children, plays games, etc. We work with Middle School students from a local school in Delhi to promote friendship and common understanding between cultures. We work on projects together, sing together, go to each others' schools to visit, write penpal letters, etc. We know many Indian families through the school and personal associations and get together socially for holidays, weddings, and general funtimes.
- 15 Community Outreach through IB CAS programme. I am also the Community and Service Coordinator for the Middle School.
- Very few as limited to compound due to security concerns. But through the school is sports and drama performances
- 17 ski club Roadrunners (volksmarching group, I organize "Beer runs" and "Furniture Runs", member of the PTA
- 18 Church, local pool club, community service project to help under privileged children, and we live in the local community.
- Staff committee on Tibetan Children's Village alliance, Reach-Out--a service learning project with neighboring Jhuggi (slum) kids, service learning with a school for construction worker children, sponsor of 2 children at Dipalaya (a slum school), sponsor of Tibetan nun at Doma Ling monastery, AIDS Project-Concern International fund raising walk, Terry Fox Run for cancer research,
- 20 Not many. I am Advisor of NHS, so we organize fund raisers, and do monthly trips to a local orphanage. Other than that, very few. On a social note: lots of concerts and plays in the city.

(20 total)

∢Back

http://survey.aea267.k12.ia.us/Member/Public/report.jsp?AnswerID=338942&theReport=15883&the... 4/29/2005

APPENDIX P DISCUSSION QUESTION RESPONSES

Discussion Question Responses by the International Teachers

Question responses are in the order of how they were received in the online discussion forum.

Question #1

Within the multicultural educational models in research literature, one of the characteristics of effective teachers is their ability to negotiate other cultures. In addition, the CCAI inventory that you completed gauges a participant's ability to empathize with those who are different from oneself. Involvement with, and understanding of, students' home backgrounds is part of this process, which implies understanding and negotiating community/local cultures. In the survey/narrative question responses, nearly all the participants indicated they are involved with 1) sponsorship of school activities (clubs, sports, drama, PTA, committees, etc), 2) volunteer work in the local community (fundraising, service work, outreach work, etc.) and/or 3) cultural activities (art, music, travel, local events, etc.).

Why are these types of involvement important to you as an international teacher? How does it contribute to your effectiveness as a teacher in a diverse classroom?

Participant 6

I would like to take just a few minutes to respond to the questions proposed. Your participation in any new culture takes courage, open-mindedness, and a sense of humor. To join a new culture is an intimidating experience. You want to make sure that you are following the societal norms and expectations, so that you don't stand out as a "typical American." The best way to do that is by slowly easing into a new culture by working alongside people who are familiar with the culture in which you are working in. As a teacher supporting a school function, you can have the best of both worlds. As teachers we feel comfortable in the surroundings of our own classroom and schools. Therefore, there is little risk of going out of ones "comfort zone" when we can support school functions in a school setting. When you go outside of that "comfort zone," you are leaving yourself open to be corrected on the proper behavior of being in a new culture. I think I can speak for all teachers when I say that teachers don't like to be corrected. So by having the best of both worlds, we can learn new things about our new culture without going through any embarrassing moments. Participation in these events can be rewarding in many ways. As students and families see you interacting, you are opening yourself up to be invited to new opportunities and experiences within that culture. These experiences can be priceless.

Finally, I feel that these unique experiences help us become more effective by teaching us experiences that cannot be learned through reading books. The skills that you learn while visiting and teaching in another culture are numerable. As classroom teachers, we constantly need to be flexible to meet the needs of the students, parents, colleagues and administrators.

Participant 3

Here at the American Embassy School in India we tend to teach students that are used to having servants, drivers, etc. that give them an unrealistic view of life and how most of the world lives. Therefore we attempt to provide our students with experiences that get them in touch with the less privileged. We also have several cultural events at the school, put on by local students/adults, that give our students the understanding and appreciation that great things can be accomplished by students from a variety of cultures as well as the rich or poor.

It is truly a blessing to teach at a school that genuinely celebrates diversity (we have students from 55 different countries) instead of using differences as an excuse to create dissent.

Participant 8

The traits listed (school activities, volunteer work, and cultural activities) were all things [wife] and I participated in before we went overseas...we coached, did PTA things, scouts, local food shelves, etc. already. These things did not change in terms of participating once we went to Delhi or to Hong Kong but did change in diversity, depth of experience, and variety of locales. To help with hungry people in the streets of Minneapolis, the jhuggi in Delhi, in the sewers of Ulan Bator, were both different in experience yet the same manifestation of who we were and are.

I am not sure of any specific impact in my physics classroom per se from these experiences, but hopefully they make me a better person overall, regardless of whether my class is quite diverse (overseas) or quite un-diverse(White Bear Lake did not have lots of diversity at all).

Participant 1

What a very different situation I am in, teaching in a Dept. of Defense Dependents' School here in Okinawa, Japan. Our students are of diverse racial backgrounds, but most are of moderate means. We have African Americans, Hispanics, Asians and Caucasians all studying quite harmoniously in our military schools. Many live on base rather than off base in the local communities. But about 20% I guess have Okinawan mothers and extended families near the military installations. Some are half-Korean, many are half Phpilipino and maybe 20-30% are White exclusively. We are so used to being in a multi-racial, multiethnic environment. I don't think we think much about it.

I have traveled extensively throughout Asia, India, Mongolia and Europe. I share these experiences with my students. Most do not travel other than to and from the military installation and the US. Traveling in this part of the world is so very expensive for families. Recently, however, 40 American kids (20 basketball players from each of our two high schools) were able to fly to Hong Kong for a basketball tournament. This was quite an adventure for our students who stayed with the Hong Kong International school students' families.

Participant 17

As a teacher with the Department of Defense Dependants Schools working in Aviano, Italy I find that the most notable difference between us and our stateside counterparts is that we have a 100% employment rate! No unemployment throughout the military community. This factor clearly eliminates many of the problems experienced by our state side counterparts.

As a second grade teacher I tend to be colorblind as do most of my students. My class make up consists of 16 Caucasian, 3 African American, 2 African American/Caucasian mix, 2 Oriental/Caucasian mix and 1 Turkish/Caucasian Mix. The best part of such a mix is the ability of the entire class to benefit from the experiences and background of others. For example it is quite common in my class to encourage parents to come in and share their cultural and geographic backgrounds. For example, the parent of my "Turkish" student came in and did a power point presentation of sights in Turkey. The kids loved it. Another parent did likewise with Panama.

Sponsorship of school activities is virtually identical to that in the stateside schools. Volunteer work by teachers is only fair. I believe there was more in years past when the average age of a DoDDS teacher was much lower than now.

Varied culture experiences are readily available to the entire military community. Traditionally, about 1/2 of the families in the

military community take little to no advantage of all that is offered by being stationed in Italy. 40% take advantage of all that is offered while the remaining 10% get out just a bit.

I find that our DoDDS students are much more aware of World Geography than their stateside counterparts. They also seem to exhibit a much great understanding, tolerance and acceptance of those who are different from those who are different than themselves.

Participant 4

I wouldn't say that these are important to me as an international teacher; rather, they are important to me regardless of where I am or what I am doing. However, being overseas makes them even more important because they enhance the overseas experience. As an overseas teacher I am also a student of culture and all that is happening in the environment of the host country.

The more I know about the environment in which I live, the more I am able to bring that knowledge into the classroom to share with my students.

Participant 18

I find that extracurricular activities greatly enhance my communication with parents. My parents and students mostly come from the Indian subcontinent and sometimes don't feel comfortable coming right out and asking me about school. But when I coach cricket they are eager to begin conversations about cricket, which inevitably end up with questions about school. There is more of a bond created with the parents regarding less stressful after school activities. Traveling to tournaments is a great eye opener for the students. As well as the teacher as you become more privy to the general life of your students as you spend hours with them on the bus and/or plane. Thus you become more in tuned with the students' culture and better able to teach in the diverse classroom.

Participant 14

I believe that the saying "you had to be there" applies in this situation. We are products of our experiences. My husband, [name] and I have lived in Argentina, Cairo and now in New Delhi. We have had many wonderful cultural experiences because of where we have lived, and because we have been involved with people and their customs from around the world.

So many times in the U.S. we run into people who have never traveled to the next state or even town! They have all formed

opinions though of places around the world! I remember the night before we left for Argentina, we talked with a woman who was OUTRAGED at our pending adventure - "Why would you want to go there?" were her exact words. When I asked if she had ever been there...NO was the answer. Learning about other people's lives has enriched our own. That appreciation and understanding permeates our daily interaction with students, parents, and community members who together are representative of 50 or more nationalities. People who have never traveled or gone out of their way to learn about other cultures and peoples will never be able to relate to our experiences. All of these experiences lend themselves to embracing diversity. "You have to be there."

Participant 20

I would agree with [participant 3] about part of our role as teachers being to use these opportunities to try and help teach the students that there is more to life than leading a pampered existence. I would agree with [participant 4] about the reason that I am living here is to try and gain a better understanding into how this culture works.

As you might image, the need here in Johannesburg, South Africa, is huge, but nothing compared to the vast gap between rich and poor. These factors result in an environment, which practically begs for action and participation.

Our new mission statement (adopted only 2 weeks ago), stresses that all students will become motivated global contributors. Clearly, involvement in some kids of outreach program is crucial for this. Many of our students are involved, and I think that by my being involved and aware of the kinds of things the students are doing helps me build a better in class rapport with the students.

In addition, I have started to use my position as NHS Advisor to discus some of the potential pitfalls of effective international development work. Clearly, some of these students will eventually go into these types of positions, so I feel like I am continuing to carry out my ongoing Peace Corps mission by helping the kids to focus on these issues.

Participant 5

We were overseas, spent time back in the States, and now are overseas again. As several of you mentioned, the activities and my involvement in them both overseas and in the States were not significantly different. I was even teaching a very diverse group of students in my ESL classes in the States. The difference, for me,

lies in the personal challenge of "understanding and negotiating local cultures"--from the language, the customs, the beliefs to the everyday sounds and sights on the streets. I find this process to be incredibly stimulating and enriching. This is what I missed in the States--the daily opportunity that living overseas provides to grow as a person, to learn about other cultures, to realize that my way is not the only way. The paradox is that it has all helped me become stronger in my own beliefs and values and to know myself better.

How does this all contribute to my effectiveness in the classroom? Like [participant 8] I'm not sure of the direct impact, but I hope that I can in some way to help students grow, develop, and learn about themselves and others every day.

Participant 9

I find myself in a slightly different situation than has been discussed in the Taipei American School in general and my math nirvana classes in particular are very homogenous. In fact, now that I think about it AEAS was only school I have taught in that was diverse. My Peace Corps school, rural northern Michigan and elite private school in Minneapolis have all be quite homogenous.

I look at the multi-cultural aspect of my teaching much more in terms of my learning curve being sensitive to the particular culture that I am teaching within. In Taipei, a rather interesting (from a math nerd's perspective) situation was that I initially concluded that my students tended to reverse numerator and denominator. Upon further investigation, I found out that in the Chinese culture you tend to go from the most general to the more specific. An example in point is that the addresses are given with country first, city next and then street address last, so it made perfect sense that my students would tell me the denominator first, which is of course the whole, and then the specific which is the numerator second and I realized that the problem was mine not theirs but since they were headed to the US of A for university, they needed to change.

In the Peace Corps, it was a situation of not challenging authority and my desire to make the mathematics real. My student's families got paid per kilo of copra so I thought that would a great real life situation to model. Little did I know that the chiefs had been skimming a fair amount because the villagers didn't know what they should receive, that is until I taught the kids. I hadn't even paused to think that the people were being cheated and that I should have approached the high chiefs first (maybe a little Midwestern naivete).

And so back to the question, I feel it is important, I do get involved in activities outside the classroom and into the community, so that I can gain insight into the workings and nuances of the culture I am living and not commit as many politically incorrect moves. I'm not all that confident that I am learning at the rate I should be, but "too soon old, and too late smart" and then of course there is "time and tide wait for no man".

Participant 10

I agree with so many of you who have mentioned the depth and richness that living and working in different cultures adds to our own lives. So many positive and delightful experiences take place every day.

I'd like to add, however, that frequently it is my experience of frustration, misunderstanding, and sheer "Why on earth would they decide to do it THAT way?!" that helps me to work with my students towards a deeper respect of their host culture. When we teach our students back home about another country and its culture, we tend to focus on colorful aspects; tradition, arts, food, achievements, the surface presented to the world. Once living in that country, we often experience, along with the adventure and excitement, occasions when we have to accept that things are just done differently here. Coming to terms with that difference, adjusting to it, accepting, digging into history to find out why, making the change oneself or not, requires a great deal of thought and examination of one's own beliefs and behavior.

Encountering the host culture in that way does not happen for all our students. Some may have families who seldom venture beyond their compound, and feel quit uncomfortable with many aspects of the host country. The school can give those students a window into diverse worlds, and a teacher who has spent time out in the community can help to explain unfamiliar customs and habits in a way that makes them seem less strange. Expressing your own puzzlement over cultural differences in a way that is respectful and positive is one of the hardest challenges to meet, but can also be the most rewarding as people respond with questions of their own.

As a British teacher married to an American and having worked in International schools in Europe and India for many years, I am constantly learning that tolerance, openness and knowledge go a long way to creating trust and understanding. If I can convey some of that to my students, then we are all better off!

Participant 2

Being a teacher, whether in American or in an international setting, should have the same criteria for work. It is important for all teachers to be involved in extra school activities so that you present to your students and the school population in general your commitment to all of education's parts. In the international setting, I believe that it might be a bit more important to be at activities that support our students, so that they can have that community feeling which they get at 'home' while living in a foreign culture. It is, also, important that the students be encouraged to participate in the new environment in which they are residing in order to experience the best of both worlds, their American background and their new one. As a social studies teacher for many years overseas, I suggest before a long weekend or a vacation, a place of historical importance in the area to encourage them to get out and experience the culture. I see my role as a teacher as a guider of ideals and concepts. If I can assist my students in accepting new ideas, adjusting to new cultures, etc., then I see them growing in diversity, global knowledge and becoming world citizens.

Participant 15

What an interesting discussion! I agree that international teachers seem to be more involved and committed to extracurricular activities and community service. However, I think this may reflect the type of people who pursue an international teaching career. They recognize the responsibility of being a part of the community. If you look at the history of these teachers, I'm sure their CV will include a long list of outside activities, which started years before they moved overseas. Right [participant 3] and [wife]?

On a personal note, I have always been particularly committed to community service. I took my own children to help out in an orphanage in Hong Kong, the polio vaccination program and other opportunities and have watched them continue to be involved on their own volition. I see the same process with my students. Once they start to work with others in the community, they continue to do so as adults. Community service has a ripple effect. Thanks to [participant 4] and [participant 5], REACHOUT (a community outreach program) continues in Azerbaijan.

In addition, there seems to be more visible opportunities for community service overseas and more available resources. But whatever the project or activity it must be viewed by the participants as worthwhile. The middle years IB programme attempts to incorporate community and service into the curriculum as an area of interaction but with limited success. The students do not realize the benefit unless they are actually doing something. It can be working to plant trees or laughing with a group of children from a local school as they attempt to solve a puzzle together. How does the involvement in all of this contribute to my effectiveness as a teacher in a diverse classroom? It helps to connect me to my students and it helps to connect us all to the global family.

Participant 7

I have been teaching in Monterrey, Mexico for two and half years. The school's population is primarily Mexican and Roman Catholic.

One of the most rewarding aspects of this experience for me has been learning about the culture and religion via my students. I love to sit and have informal chats with my students about their families and lives outside school. This is when I learn the most about them. This helps me to be a more effective teacher and better able to navigate my way through Mexican society without offending anyone (I hope anyway.)

Another way that I have learned and grown as a person and teacher is by traveling throughout this beautiful country as much as I can. Mexico is very different from region to region and when I am able to experience some of this diversity I share my experiences with my students. They almost always expand my knowledge.

Participant 11

Some of the things that I do outside of the classroom are things that I would be doing if I lived in the U.S. The relationship between my involvement in these things and my effectiveness is simple - These things make me happy, and I'm a better teacher when I'm feeling good about my life. For example, my involvement in an English-speaking church provides a little slice of home. I am able to be involved in a worship experience without needing a translator. At church-sponsored events, I have the opportunity to interact with a large community of ex-pats. (It also provides a much-needed break from my fabulous, but school-related social circle.)

Some of the things that I do outside of the classroom are unique to my living experience in Mexico. School-sponsored events, interactions with families, attending cultural events, etc. all provide me an opportunity to interact more with Mexican culture and help me learn more about the students that I am working with. I learn something new every single day - about Mexican culture, about the

students, about myself - and the new knowledge helps me to more effectively relate to the students.

Ouestion #2

In the survey/narrative question responses, nearly all participants indicated they use a wide variety of instructional strategies from one-on-one to the use of graphic organizers and multiple others. Research highlights the most effective instructional strategies as 1) identifying similarities and differences, 2) summarizing and note taking, 3) reinforcing effort and providing recognition, 4) homework and practice, 5) nonlinguistic representations, 6) cooperative learning, 7) setting objectives and providing feedback, 8) generating and testing hypothesis and 9) questions, cues and advance organizers. In your classroom, which of these do you find to be the most powerful effective teaching strategy that supports cross-cultural experiential, authentic, and interactive learning? Why?

Participant 2

While all the strategies listed below are great, one that I found worked extremely well was to allow the students to become the teacher. I teach high school and one day an angry young man stated, "I wish I were the teacher. I could do what you do. You just tell us to answer all those questions, write that report, and take this test, while sitting on your butt as WE do it." I realized that there was definitely a gap between all my planning to provide the BEST possible lesson to get information to my students and how they saw that system working. So, I popped off and told him that TODAY was his lucky day. He could present the next chapter of material in (almost) any way he wanted. He could even choose 2 friends to assist him in the teaching. Well, he took the challenge and my classes were off to a new way to learn. (Not really so new. Sharing or teaching others in small groups or through presentations has always been going on in the classroom.) What was new was telling them to make lesson plans, choose the videos, decide the class work and the homework, and to give the test. This seemed to make them really the 'teacher' rather than a participating student. That opened up a whole new world of learning. Others in the class begin to want their turn to "teach". So, I taught a lesson on how to make plans, write objectives, look at standards, how to choose to present the material and how to assess what the students learned. Then, I let them go through the book (United States History) and choose the chapter that they would like to do. I would teach a lesson, then a student teaching-team would do one. I had to beg to get to do my every-other chapter. I saw such excitement. I found them in the library studying 'their chapter' weeks ahead of time. I could not believe the diversity that came with this "experiment". They set the standards for learning and attainment a lot higher than I did. The "teachers" were graded on their plan, their presentation of the material, and the way they

graded the work. We made mini movies, acted, interviewed, role played, and yes, we answered the questions in the book, wrote reports, and took tests, but somehow it seemed different when they were the ones telling each other it MUST be done for the grade.

Participant 5

It is difficult to choose only one strategy as most of the strategies complement one another. With my field, languages (French, Spanish, ESOL), it is also difficult to differentiate between which strategy best supports cross cultural experiential, authentic, and interactive learning since language learning should be all of these. However..."Questions, cues, and advance organizers" wins my vote.

I think of this strategy as the scaffolding process that is so important for language learning. It is especially useful for helping students handle authentic materials--by this I mean materials that are intended for native speakers of the language. Students are able to handle native-level written and oral texts if the tasks are appropriate to the students' level. Students are well supported in the language learning process and can build their skills and confidence. The questions and cues--especially visual ones--also support student interaction as the students have a base to work from and a framework with which to organize their communication. As a result, students are well supported in the language learning process and can effectively build their skills and confidence as they increase their cross-cultural understanding.

Participant 14

I also find that cooperative learning is an extremely powerful tool. In my beginning ESL class I have 7 students representing 5 different languages: Japanese, Dutch, German, Korean, and Arabic. At the beginning of the year 2 Korean speakers were "non speakers." Both of them are speaking, reading and writing now. One of the main reasons for this is cooperative learning. My 3 girls all speak different languages and in order for them to communicate with each other, they need to speak in English - a great motivator! My Korean girl is fairly advanced for a beginner and has shared and worked with the other two girls helping them to understand instructions, study new words, practicing pronunciation, writing stories, etc. Cooperative learning is the open door that allows these girls to build that common vocabulary, scaffold meaning, and answer questions that are important to their learning. I use all those other techniques to - whatever works is my motto...Thanks for "listening".

Participant 1

After reading all of your responses to question one it is pretty apparent to me that teaching overseas, but on a US military base, is very different from most of your experiences.

We are diverse in cultures, but homogenous in that all are from military families. Many interact off base in churches, sporting events and with their extended families as I said before. Our differences seem inconsequential until it comes to ESL.

We have many ESL students mostly Korean, Filipino, and some Japanese. In art I, of course, use so many methods to "teach the lesson." Demonstration of the target skills, examples of past work and handouts detailing "how to do it" are all techniques that work for all students, even the ESL kids. Often there will be another student in class who can speak the language of an ESL student and thus, translate fine points of the activity.

We all have a great time figuring out how to make it work. I have a short story though, that's kind of funny. I had a Korean girl who always said, "Too hard; can't do." I found a previous Korean student who had taken the same class and was familiar with the art project. She came in and explained the color mixing, but in the end said, "She doesn't understand Korean either!!"

Participant 8

Here is what I believe I do in my physics classrooms for any kid for most any reason..

1) identifying similarities and differences, 2) summarizing and note taking, 4) homework and practice, 7) setting objectives and providing feedback, 8) generating and testing hypothesis, and 9) questions, cues, and advance organizers.

In science labs we are working in groups but I eschew the more formal "roles" sometimes emphasized by some cooperative learning pieces I have read, and of course there is a wide array of methods to present content. One biggie in science ed is for students to work on overcoming their own conceptual misconceptions about a topic (motion, electricity, etc) through a sequence of activities that are supposed to help them construct their own valid ideas about the phenomena being studied. Again, this is not with any cultural connections implied at all. The toolbox is meant to help all students arrive at a better understanding of the physical world we live in.

Participant 10

This list of instructional strategies is like a rich and varied winter holiday feasts; impossible for me to select just one dish! As [participant 5] says, they all complement one another.

In my third grade classroom, with its mix of 20 children of 11 different passports and varied cultural and linguistic backgrounds, cooperative learning and non-linguistic representations play an import part. I would probably also single out reinforcing effort and providing recognition for the role this plays in making children feel valued and comfortable within the classroom, thus opening their minds to the learning experiences provided there, as well as motivating them to give their best. For our students the school is almost certainly more of a community center than it would be in their native country. Friends, activities, community and cultural events are all within these walls, it is essential that they feel safe and supported here emotionally before they can progress academically. The work our school has been doing over the last few years around the topic of assessment has put the spotlight on setting objectives and providing feedback as a valuable learning tool, so I would throw that one into the mix too. Working with my kids to help them understand what they will know and do when we study a topic helps us all to focus. That said, there are enriching experiences that only come from taking time to follow the questions students raise (8) and sometimes there can be some conflict between the two, especially in terms of time.

Participant 18

As others have stated, it seems that using many of the strategies together works the best. I think I can best answer the questions by responding to each strategy within the question.

In the survey/narrative question responses, nearly all participants indicated they use a wide variety of instructional strategies from one-on-one to the use of graphic organizers and multiple others. Research highlights the most effective instructional strategies as 1) identifying similarities and differences. Venn Diagramming, discussion 2) summarizing and note taking. This is a challenge especially for the ESL students but summarizing and putting into their own words is powerful tool for assessing comprehension 3) reinforcing effort and providing recognition, Certainly 4) homework and practice, This doesn't always work as parents "do" the homework for the kids5) nonlinguistic representations. We often mime words or create mind sketches 6) cooperative learning. Great method if you can pair ESL with fluent speakers 7) setting objectives and providing feedback. Certainly 8) generating and

testing hypothesis, and formative and summative assessment 9) questions, cues, and advance organizers. Great. In your classroom, which of these do you find to be the most powerful effective teaching strategy that supports cross-cultural experiential, authentic, and interactive learning? Why? Depending on the subject I think that summarizing works best for reading: 5 for vocabulary: 6 for ESL and it all depends on what outcome you are seeking.

Participant 4

It's hard to say which is the most effective. Maybe there isn't one method that satisfies that description. But I've found that cooperative learning can be very effective. For several reasons, but among them are the fact that oftentimes students can be effective teachers of their peers. Having students work cooperatively gives them the opportunity to take more responsibility and ownership for their own knowledge. They can't very well work together unless they've mastered some aspect of the task themselves. Also, the collective knowledge of a group can often surpass that of any individual in the group. Once students realize the power of cooperation, they often become very positive about tapping into it.

Participant 17

Upon reading all the responses to question 1 and the responses from [participant 1], [participant 14], and [participant 5] to question 2, it is apparent that many of us interact regularly with kids who are ESL. I know that when I taught in Yokosuka, Japan about 35% or so of the student body was of Filipino, Korean or Japanese heritage. There was a fantastic ESL program in place to support those from diverse backgrounds. Many of the regular classroom teachers implemented many of the strategies generated by the ESL teachers.

However, I currently teach at Aviano Elementary School in northern Italy and find myself with no students who even remotely come close to qualifying for ESL services.

I am not too sure why this is so. If fact, we only have one ESL teacher for a student body of about 800. And her position has been in jeopardy for some time due to low numbers.

I believe that the most powerful teaching strategies for ESL students--or all students for that matter--should include homework and practice. Any time a parent gets actively involved in a child's education it creates a win-win situation. Reinforcing effort and providing recognition for such is also important for our ESL

students. In fact, this strategy is most important for all our students. Kids, as well as us adults, enjoy hearing the good things about ourselves and our efforts. The more good we all hear the more good we all produce. It's simple human nature.

Participant 20

Like so many of you, I have been teaching in international schools with diverse populations for long enough that I don't really focus on the cross-cultural aspects of my class anymore, I suppose I just take the diversity for granted.

In my high school math classes, the strategies I tend to find the most successful are:

Cooperative grouping. I feel that students can often learn much better when kids are teaching other kids, than they can from a teacher. I think it lowers their anxiety levels, and students don't seem to be as wary of asking for further clarification from a peer. In terms of the actual grouping, my decisions are generally based on ability levels, rather than on cultural differences, with the possible exception of trying to place a possible translator in any group containing beginning ESL kids.

In my higher level classes, I like to focus on similarities and differences, which serve as excellent writing prompts. I feel that being able to list certain characteristics help a student see how a particular problem fits into the bigger picture, and how it will be useful to them later on.

Finally, in my geometry classes I use a lot of hypothesis testing. I feel that students can learn much better when they are developing and proving their own conjectures, rather than simply mimicking exercise out of the book.

Participant 7

I teach second grade.

In my classroom I have found the most effective teaching strategy that supports cross-cultural experiential, authentic, and interactive learning are: cooperative learning and identifying similarities and differences.

Participant 11

Cooperative Learning provides great social learning opportunities for students. Students will eventually find themselves working with groups and teams, so it is important for them to learn *how* to work with a group. (Yes, it is ok to politely disagree with your team. No, you don't always get to choose who you work with. Yes, working together will get you much farther than working

against each other.) We're helping students build effective communication skills that will be used throughout their lifetimes.

Identifying Similarities and Differences can be an effective way for students to put their life in perspective with the rest of the world. The differences between things/places/people are easy to see, but the similarities help us to gain a better understanding of whatever it is we are comparing. I teach second grade at a private school in Mexico--the students that attend the school are from upper-class Mexican families. My students have more opportunities to view the world than I ever could have imagined at their age--yet their world views sometimes make me shake my head in confusion. (Last year, one student told me that he was learning French. "Wow-soon you'll be able to speak three languages!" He just shook his head, "No, four! I already know Japanese." He started into a long string of noises combined with martial arts movements. When I asked him where he learned that, he told me "It's on TV. People from Japan talk funny." Fear not, by the end of our conversation, the student told me that foreign languages sound funny to our ears-just like when he started learning English and French, or even during the first weeks of school when the students were adjusting to hearing English spoken with my Midwestern accent.

Ouestion #3

In the research within the multicultural education framework, effective teachers are seen as those who seek to teach the students the culture of the school while maintaining the students' sense of their own cultural identity and pride. In the survey/narrative question your responses indicated a variety of ways of using students' home cultures in the instruction of the classroom. Explain your beliefs in how important this practice is in helping students construct knowledge and learn.

Participant 1

In my art classes I've just begun a project where the kids use the internet to find an artist whose work interests them. Many search for easy-to-paint, abstract-looking work and others are totally taken with Japanese Animae artists. I've noticed that new students to the American culture, i.e. Japanese, Korean kids, will choose artists from their own culture and I certainly encourage them to do this. Eventually, they will have a folder of pictures in their student accounts of artwork, a commentary about the artist's work and an example of their own work painted in the style of the artist they selected. Then with our InFocus machine they will be able to

present their project in a very large ten foot square size. Sure beats showing work on a 27" TV screen.

Participant 7

The students at my school in Mexico are pretty homogeneous. They tend to be Mexican, Roman Catholic, and Spanish is their mother tongue. The culture of the school is pretty much parallel with their culture. There is very little concern here of students feeling "other" or overly different. I am the only non-Mexican in my classroom. I probably learn more, culturally speaking, from my students than they learn from me!

I try to teach them as much as I can about other cultures with a focus on Canadian culture since it is what I know the most about. My students are always very interested to hear that children in Canada learn French and English in school in a similar fashion to the English and Spanish taught at many schools here in Monterrey, Mexico.

I have had one or two students each year who are not Mexican Roman Catholic in my classroom. I have to be careful to be respectful of their religious beliefs when the time roles around for religious celebrations at the school. Fortunately, the parents are very active in the school and many of them come to the school on a weekly basis to teach religious values. There is a separate class for children that are not of a Catholic denomination.

Participant 6

This is probably one of the biggest challenges as a teacher who is teaching "overseas" for the first time. This last year and half, I have been teaching a more privileged population than what I am used to. So I myself don't have the same experiences as some of these students to show relationships to the real world. When we complete classroom novel studies, we generally read books with more conflict that what these students may ever encounter. Therefore, it is sometimes hard to show relationships with characters in the book and real people in their lives. The less privileged population always seems to know of someone who has dealt with similar conflicts as those characters in a story.

I try to use generalizations as I compare my culture with theirs. However, I have learned that this is not a best practice. Students hear things differently from how they are said and then it usually blows up into a simple miscommunication.

Participant 4

I believe that it is an important practice for two reasons. First, by helping students maintain their own culture within the context of a course they are able to more easily connect with the curriculum. Once they make the connection, they "construct knowledge and learn." Second, by connecting the cultures of the students with the curriculum, other students learn that theirs is no the only culture, that there are other ways of looking at the world, that there are various viewpoints (not necessarily equally valid, but all worthy of analysis).

Participant 12

I think that focusing on each child's home culture serves two purposes. First it allows the students to stay in touch with their base values and by extension share their beliefs and practices with others in the class. Secondly, by sharing these values the students gain an appreciation of other cultures. This act of sharing adds value to their lives and hopefully to the lives of many others. In all of my classes I use a weekly journal exercise, which encourages students to locate articles from their home country and write a comment on the article's contents. Students are encouraged to question the presenter's position. This certainly fosters some lively debates. Mind you, rules of decorum are set and strictly adhered to.

Participant 10

It is extremely important. And also incredibly daunting when you look at the variety of cultures represented in our classrooms and realize how little we personally know about them, however long we may have taught. Some of my most effective teachable moments have taken place following the discussion that arises out of a child sharing an experience or custom with the group. Students construct knowledge on the foundation of what they already know or believe to be true. Giving them the opportunity to share culturally specific ideas with the group gives the teacher insights into where the child is 'coming from' and helps her to formulate worthwhile learning activities that make sense to the child.

Teaching the culture of the school is interesting. International schools can be mini communities with significantly different values from those of the host culture, yet within them are teachers and staff from a variety of cultural backgrounds. Would we all agree on what the culture of the school actually is, even if it has been articulated in a mission statement?

Participant 13

Being our first semester here in Beijing I am finding that this impacts overseas teachers in many ways. The school culture here is radically different from the Arabian Gulf. Attempting to make cultural connections with students who come from incredibly strong micro communities makes it difficult to foster a much larger sense of community inside and outside the classroom. It can be a frustrating experience. However, allowing students to analyze their reasons for being here in China and what it means to them and their home society is crucial. The Korean students often have a very different perception of why they are here as opposed to the Canadians. Recognizing that first there may be larger cultural forces at work within each micro community allows for students to begin better understand that there just may be some similarities. This new knowledge coupled with their own understanding of their home culture (a dash of China thrown in) and they can begin to construct understanding of the greater community at ISB. Additionally, I find that the size of the school plays a role in the building of a greater sense of school culture or community. It is somewhat simplistic, yet appears to ring true, that a small school will encourage a closer interaction between different communities and allow for students from different cultures to more quickly make learning more meaningful.

Participant 14

Well, since this is the "Christmas" season, it can be a great time to celebrate traditions from different cultures. We just had a 6th grade gathering where students from around the world told about holiday celebrations that happen in their cultures at this time of the year. Students shared songs they sing, and we sang Christmas Carols in many different languages. These kinds of experiences are important for students to feel comfortable and accepted as well as appreciated for their diversity.

Within the ESL classroom it is critical that one part of the students' culture, namely language, is used as a special learning tool. When learning new vocabulary, many students not only write the words and meanings in English but also in their native language. When instructions are given to beginning ESL students, many times a "translator" within the class is a valuable tool. I feel that allowing students to get help from using their own language definitely promotes learning while highlighting the importance of their culture.

Participant 5

Students need to feel accepted and secure in the classroom to learn. They need to feel that they belong. This can be difficult for

students everywhere, but in an international school there's the added variable of being surrounded by people from other cultures and nations. It is critical for teachers to help students maintain their own cultural identity as they learn about other cultures. The classroom is an ideal place for this to happen. Recognizing students' home cultures and countries goes a long way in helping students feel valued in the classroom.

Participant 20

I think that having an awareness of students' cultural backgrounds can be very helpful in forming my interactions with them.

For example, my time in the mid East taught me that most Arab cultures are decidedly more vocal than written. I notice that my Arab students are much less successful having to perform a task in writing, than they are in explaining the same concept orally. On the other hand, naturally quiet cultures, such as my Japanese students are just the opposite. This year, my students with the least English are all Japanese, and with them I notice that they are able to express very complex ideas in writing, but struggle (or are reluctant) to verbalize them.

I feel that being aware of these differences can be beneficial to the teacher, in that students can be placed in a position where they can choose in which manner they feel most comfortable demonstrating their understanding.

Participant 17

Being a DoDDS teacher I find that virtually all of my students are of American heritage. Their parents are American. Their grandparents are American. Great Grandparents are American. And so on down the ole family tree. Even those students who are from "mixed heritage" parents have a strong sense of being American. This is a feeling of pride in one's American heritage is a direct result of the problems we face in Iraq and elsewhere. Being on a US military base only enhances this sense of price.

The greatest challenge is in teaching the kids tolerance for those whose views are different from their own whether they be Italian or whatever.

Ouestion #4

Research indicates that one of the most powerful factors in student achievement is what the teacher believes about students and what the teacher expects of students. In addition, the teacher's perspective is shaped by their own upbringing, their life experiences, their beliefs and values, and their sense of efficacy (the belief they can change a situation). What personal values do you hold regarding teaching in an international and diverse setting? What do you believe about how you affect student learning?

Participant 8

So, what do I believe? Honesty (science facts can tell a lot, as quoted in "CSI" all the time)...integrity.... openness...sense of humor.... belief in the inherent value of each kid...tolerance.... self-confidence.... the important of perseverance, persistence. constant effort, willingness to make mistakes, look at things anew, be open to the wonders of the universe (for goodness sake look up in the sky and ponder what is up there, I tell my astronomy kids)...lightening up when necessary..being the hard ass when necessary...being a human being? Like I had said previously, these are the very traits I feel I tried to value in lily-White suburban Minnesota as well as in Delhi and Hong Kong. Being the product of intelligent lower-middle class parents who were unable in the late 1940's to attend college has had an impact....that old dream of getting the next generation a bit better educated and further along still is a powerful one. That is the epitome of efficacy.

Participant 7

Whether I am in an international or local setting the most important variable for me that helps me to be more effectively teach my students is to understand them. This encompasses many things and can take a lot of time. I need to know the world my students come from in order to more effectively motivate and captivate them. This means finding out about their families, friends, religion, and activities outside of school.

The biggest challenge that I have encountered teaching in an international setting is the language differential. In order to reach and teach your students you have to address them in a discourse that they understand. When I first came here I was looking at a sea of blank faces for the first two months of the academic year! It was disconcerting to say the least. I still see the blank faces at the beginning of the school year but I have learned to be patient with myself and my students to overcome this stumbling block to effective instruction. The children need time to get accustomed to my Canadian accent and I need time to listen to them try to express themselves in a language that is very secondary for them. I have discovered that learning a smattering of Spanish has helped me exponentially. I would like to learn more but, of course, time always seems to be too limited.

I know that the more comfortable that I am with my students the more comfortable they are with me as their teacher. When a safe, comfortable environment is established in the classroom then my students and I are free to learn and grow together. Jumping the language barrier and letting the students know that you truly care about them are just two ways to achieve this comfort level.

Participant 12

I believe that as a teacher you can have a powerful influence on your students. I believe that if you show pride in your own cultural values and openly and sincerely show interest in the cultural values of others you make a positive impact.

What I like about teaching in an international school is that I get to work with my wife. It is such a joy for me to see her in her classroom, walking the halls or sitting across from me in a meeting. I believe that the students pick up on the strength of our relationship and feel that they can ask us questions that otherwise would go unasked.

I also believe that the diversity of my life experiences adds a positive aspect to my teaching. My 26 years of experience outside of teaching gives me a unique point of view, especially given my last job in Delhi.

Participant 14

All adults have a responsibility to be models for the next generations. Teachers above all have to fulfill this obligation because we are such a large part of children's lives. Everything we do - from planning lessons, to reading out loud, to counseling students - mirrors our beliefs and experiences and models the things we value. Some of the values that come through in my teaching and working with my students everyday would be: getting an education, working hard, being on time, being prepared, caring about others, respecting others, thinking before you speak, being creative, giving - the list is too long to continue here. These values are modeled each day by teachers and our students learn from the way we teach them.

Participant 5

[Participant 8's] list pretty much covered it all. (Thank you, [Participant 8]). My one addition would be the power of language. Regarding how I affect student learning--for me it comes down to modeling the beliefs and behavior that I value. The only way that these values can ring true for students is if I try to live them out in my own life and in the classroom. Made a mistake? Admit it. Think students should be honest? Be honest. Want students to set

high standards for themselves? Set high standards for yourself. Believe language is important? Take some time to learn the host country language... Easier sand than done, but worth striving for.

Participant 1

Yea, everything [Participant 8] said! I value doing the best each person can do, reading the directions, going outside of the box when the directions don't limit it. I value helping your neighbor with suggestions, but doing your own work. I value mutual respect and a peaceful environment in which to work. I value the leadership of intelligent people. I value honesty and integrity. The list could do on and on. I think that I express these values and more to my students. I have high expectations for everyone regardless of any skin color or cultural background.

Participant 13

This is crucial to teaching or coaching. The ability to coax out the best in any student is the mark of a good education. My own experiences in my youth reinforce my belief that we all have talents: they just might not be the ones that the system wants to measure. I believe that the communication of the value that we all have some intrinsic worth gives students the safety to experiment (or in some cases even try what they are supposed to do). Students who have been "unsuccessful" in the classroom need to have their outside accomplishments celebrated in an attempt to transfer some of that positive momentum into the formal curriculum. This is where coaches and teachers can have their biggest impact: look at the whole person and reinforce positive behavior. Call them on inappropriateness but do it with an arm around the shoulder and chances are they will be more responsive. Encourage and encourage as much as possible to give them the chance to realize that they are capable of success!

Even though in some cultures this is not a common role for the teacher, if the student has been in a North American oriented school long enough, they realize that this caring stuff is one of our quirks. After a while they sort of get used to it and that can be problematic for reentry into their home culture. Yet, aren't they better off for the approach? This can be particularly true when their family's expectations are incredibly high. It's about creating a safe environment for students to realize their full potential. It's our job to motivate them and take them from where we get them and move them as far as we can in realizing their full potential. Not just in academics, but in the interpersonal, social and physical realism as well. At some point school ends and they have to contribute to their families in addition to society at large. Well

rounded people seem to have better adjusted lives and enjoy a modicum of success that make them valuable members of society.

Participant 10

Values that I hold to include: the right of all to a life free from pain, threat or suffering, the responsibility of the more fortunate towards the rest of humankind, the need for all to be aware of environmental issues and act to protect the earth, tolerance and quest for understanding of those who are different from ourselves, perseverance (especially in the face of ignorance or apathy), all living creatures are worthy of respect...

I hope I can help students to see the best in each other and become more aware of the value of another's thoughts. I believe that if I can encourage them to feel positive about themselves they will approach learning confidently and master it successfully. If my classroom is a place where differences are explored, discussed and celebrated, perhaps my students will grow up better able to lead the world forward, in whatever capacity and in whatever country and culture they find themselves.

Participant 20

The values that I try to impart to my students are honesty, integrity, dependability, dedication, enthusiasm, curiosity, and a sense of the value of hard work.

I do this by giving the students a fair amount of latitude in terms of classroom behavior, but having high enough expectations that they will need to work hard in order to succeed. I believe that by providing students with challenges and choices, we as teachers can not only provide them with the knowledge and problem solving skills they will need in the future, but also help them to develop the mindset that will allow them to be motivated lifelong learners.

Participant 17

As a DoDDS teacher, working on a military base in Aviano, Italy, I find that many of the parents are here not by choice, but rather because they have been told this is where they are to go. There is always a percentage of these parents who never venture beyond the base itself one than what is necessary. It is this segment of the population, which I can impact the most. I do this by:

1. Telling parents at the get go how lucky they are to have been assigned to Italy. I highlight local points of interest and other places I have been which I think they might enjoy! In short, I offer myself as an independent travel agent of sorts. I find that a little encouragement

- goes a long way. I have even had parents approach me asking for information about this sight or another.
- 2. Study trips--it is amazing how few families have not been to local points of interest. When my class goes on study trips I encourage all parents to come and join in the fun. It is not uncommon for me to have 50-60% of my parents attend any given study trip. For many this experience has been their first venture out into the economy. In addition many of the parents return to such sights on their own.

In short I have the opportunity to open up a whole new world to incoming families. Experiencing, accepting and embracing foreign cultures is what living and working overseas is all about. It is with the great sense of pride that I encourage all my parents to do just that.

Participant 4

The values that I think are the most important and that I try to communicate to my students are: there is joy in learning-learning about your subject, your fellow students, and yourself; it is only by doing your best work that you can expect to really reach your potential; be open to other ideas, cultures, and people; you should remain a student for your entire life.